Form 23.03

2023

SUPREME COURT OF NOVA SCOTIA

IN THE MATTER OF: Application by IMV Inc. and Immunovaccine Technologies Inc. (the "Applicants"), for relief under the *Companies' Creditors Arrangement Act*

NOTICE OF MOTION

To: Attached Service List at Schedule "A"

Motion

The Applicants, IMV Inc. and Immunovaccine Technologies Inc., move for an order:

- 1. abridging time pursuant to Civil Procedure Rule 2.03;
- 2. extending the Stay of Period until and including July 17, 2023;
- increasing the Administration Charge to \$750,000 and reducing the Directors' Charge to \$275,000 and declaring that such charges have priority over all other charges and security interests, including, without limitation, over the claims of the federal and provincial governments subject to a deemed trust;
- 4. authorizing the Applicants, with the consent of the Monitor, to pay certain prefiling amounts owed to suppliers which they deem critical to their business, up to a total aggregate amount of \$350,000;
- 5. approving a sale and investment solicitation process to solicit offers for a broad range of executable transactions in respect of the business and/or assets of the Applicants;
- 6. approving a key employee retention plan (the "**KERP**") and related KERP Charge in the amount of \$575,000;
- 7. approving the procedure establishing how claims against the Applicants and the Applicants' present and former, *de jure* and *de facto*, directors and officers a claims process order, will be determined and adjudicated; and
- 8. sealing certain portions of the KERP, which contain identifiable and confidential information about the participants of same.

Time and place

The motion is to be heard by a judge at 2:00 p.m. on Friday, May 5, 2023 in the Halifax Law Courts at 1815 Upper Water Street in Halifax, Nova Scotia. The moving party has set the motion for hearing in a half day or less in chambers. The moving party says that the motion will not require more time.

References

The moving party relies on the following legislation, Rules, or points of law:

- 1. Companies' Creditors Arrangement Act, ss. 11, 12, 19, 36; and
- 2. Civil Procedure Rules 2.03 and 85.04.

Evidence

The evidence in support of the motion is as follows:

- 1. Affidavit of Service of Laura Smith sworn on May 3, 2023 and filed with this notice;
- 2. Affidavit of Andrew Hall sworn on April 28, 2023 a copy of which was filed with the Court; and
- 3. Affidavit of Andrew Hall sworn on May 3, 2023 and filed with this notice.

Possible order against you

You may file an affidavit and a brief, attend the hearing of the motion, and state your position on whether the proposed order should be made. If you do not attend, the judge may grant an order without further notice to you.

Signature

Signed May 4, 2023

Mc darthy Titrault LP

Alain N. Tardif François Alexandre Toupin McCarthy Tétrault LLP 1000 De La Gauchetière Street West Suite MZ400 Montréal, QC H3B 0A2 Phone : (514) 397-4100 Fax : (514) 875-6246

Sara L. Scott Stewart McKelvey 600-1741 Lower Water Street Halifax, NS B3J 0J2 Phone: (902) 420-3363 Fax: (902) 420-1417

Counsel for the Applicants, IMV Inc. and Immunovaccine Technologies Inc. Schedule "A" Service List

SUPREME COURT OF NOVA SCOTIA

IN THE MATTER OF: Application by IMV Inc., Immunovaccine Technologies Inc. and IMV USA Inc. (the "Applicants"), for relief under the *Companies' Creditors Arrangement Act*

SERVICE LIST

Party	Method of Service
McCarthy Tétrault LLP 1000 De La Gauchetière Street West Suite MZ400 Montréal, QC H3B 0A2 <i>Counsel to the Applicants.</i>	M ^{tre} Alain N. Tardif Tel : 514.397.4274 E-mail : atardif@mccarthy.ca M ^{tre} François Alexandre Toupin Tel: 514.397.4210 E-mail: fatoupin@mccarthy.ca
Stewart McKelvey Lawyers Queen's Marque 600-1741 Lower Water Street Halifax, NS B3J 0J2 <i>NS Counsel to the Applicants</i>	M^{tre} Sara L. Scott Tel : 514.397.4274 E-mail : sscott@stewartmckelvey.com
Aird & Berlis LLP Brookfield Place 181 Bay Street, Suite 1800 Toronto, ON M5J 2T9 <i>Counsel to the Horizon Technology Finance</i> <i>Corporation, as agent</i>	M ^{tre} Miranda Spence Tel : 416.865.3414 E-mail : mspence@airdberlis.com M ^{tre} Graham Topa Tel: 416.865.4739 E-mail : gtopa@airdberlis.com M ^{tre} Kyle Plunkett Tel: 416.865.3406 E-mail : kplunkett@airdberlis.com
Ropes & Gray LLP Prudential Tower 800 Boylston Street Boston, MA 02199-3600 US Counsel to the secured creditor Horizon Technology Finance Corporation, as agent	M ^{tre} Cristine Schwarzman Tel: 212.596.9635 E-mail : Cristine.Schwarzman@ropesgray.com M ^{tre} Patricia I. Chen Tel : 617.951.7553 E-mail: patricia.chen@ropesgray.com

FTI Consulting Canada Inc. TD South Tower 79 Wellington Street West Suite 2010, P.O. Box 104 Toronto, ON M5K 1G8 <i>Monitor</i>	Jodi Porepa Tel : 437.332.5743 E-mail : Jodi.Porepa@fticonsulting.com Jeffrey Rosenberg Tel: 416.649.8073 E-mail: Jeffrey.Rosenberg@fticonsulting.com Adsaran Vithiyananthan E-mail : Adsaran.Vithiyananthan@fticonsulting.com
Stikeman Elliot LLP 5300 Commerce Court West 199 Bay Street Toronto ON M5L 1B9	M ^{tre} Maria Konyukhova Tel : 416.869.5230 E-mail : mkonyukhova@stikeman.com <i>Counsel to the Appointed Monitor</i>
Atlantic Canada Opportunities Agency Legal Services Department of Justice / Government of Canada PO Box 6051 644 Main Street Moncton, NB E1C 9J8	M^{tre} Karine LeBlanc Tel: 506.851.2153 Email: <u>karine.leblanc@acoa-apeca.gc.ca</u>
Canada Revenue Agency	General address for service notificationPGC-AGC.fiscal-tax@justice.gc.ca notificationPGC-AGC.civil@justice.gc.ca

Email distribution list:

<u>Atardif@mccarthy.ca; fatoupin@mccarthy.ca; sscott@stewartmckelvey.com;</u> <u>Adsaran.Vithiyananthan@fticonsulting.com; Jodi.Porepa@fticonsulting.com;</u> <u>Jeffrey.Rosenberg@fticonsulting.com; mkonyukhova@stikeman.com;</u> <u>mspence@airdberlis.com; gtopa@airdberlis.com; kplunkett@airdberlis.com;</u> <u>Cristine.Schwarzman@ropesgray.com; patricia.chen@ropesgray.com; karine.leblanc@acoaapeca.gc.ca; notificationPGC-AGC.civil@justice.gc.ca; notificationPGC-AGC.fiscaltax@justice.gc.ca</u> Schedule "B" Draft Amended and Restated Initial Order

SUPREME COURT OF NOVA SCOTIA

IN THE MATTER OF: Application by IMV Inc. and Immunovaccine Technologies Inc. and IMV USA Inc. (the "Applicants"), for relief under the *Companies' Creditors Arrangement Act*

Amended and Restated Initial Order

Before the Honourable

in chambers:

The Applicants propose to make a compromise or arrangement under the *Companies' Creditors Arrangement Act*, R.S.C. 1985, c. C-36 as amended (the "**CCAA**") and they applied for an initial order and, now or in the future, other relief under the CCAA as may be sought on notice of motion.

The following parties received notice of this application: see attached at Schedule "A".

The following parties, represented by the following counsel, made submissions:

Party

<u>Counsel</u>

Applicants	McCarthy Tétrault LLP Alain N. Tardif François Alexandre Toupin
	Stewart McKelvey Lawyers Sara L. Scott
Monitor, FTI Consulting Canada Inc.	Stikeman Elliott LLP Maria Konyukhova
Horizon Technology Finance Corporation, as agent	Aird & Berlis LLP Miranda Spence Kyle Plunkett

On motion of the Applicants, the following is ordered and declared:

Service

1. The service of the Notice of Motion, the associated pleadings, and the supporting documents, as set out in the affidavit of service is hereby abridged and validated so that this Motion is properly returnable today and further service thereof is hereby dispensed with.

Application

2. The Applicants are affiliated debtor companies within the meaning of the CCAA and are companies to which the CCAA applies.

2023

Effective Time

3. This Order and all of its provisions are effective as of 12:01 a.m. Halifax time, province of Nova Scotia, on May 1, 2023 (the "**Effective Time**").

Plan of Arrangement

4. The Applicants, in consultation with the Monitor, shall have the authority to file and may, subject to further order of this Court, file with this Court a plan of compromise or arrangement (the "**Plan**").

Administrative Consolidation

- 5. The CCAA proceedings of the Applicants shall be consolidated under one single Court file, in file Hfx No. 523334.
- 6. All proceedings, filings, and other matters in the CCAA proceedings shall be filed jointly and together by the Applicants under file Hfx No. 523334.
- 7. The consolidation of these CCAA proceedings in respect of the Applicants shall be for administrative purposes only and shall not effect a consolidation of the assets and property or of the debts and obligations of each of the Applicants including, without limitation, for the purposes of any Plan or Plans that may be hereafter proposed.

Possession of Property and Operations

- 8. The Applicants shall remain in possession and control of their current and future assets, undertakings, and properties of every nature and kind whatsoever, and wherever situated including all proceeds thereof (the "**Property**"). Subject to further order of this Court, the Applicants shall continue to carry on business in a manner consistent with the preservation of their business (the "**Business**") and Property. The Applicants shall be authorized and empowered to continue to retain and employ consultants, agents, experts, accountants, counsel, and such other persons (collectively "**Assistants**") and the employees currently retained or employed by it, with liberty to retain such further Assistants as it deems reasonably necessary or desirable in the ordinary course of business or for the carrying out of the terms of this Order.
- 9. The Applicants shall be entitled to continue to utilize the central cash management system currently in place as described in the Affidavit of Andrew Hall dated April 28, 2023, or replace it with another substantially similar central cash management system (the "**Cash Management System**") and that any present or future bank providing the Cash Management System shall not be under any obligation whatsoever to inquire into the propriety, validity or legality of any transfer, payment, collection or other action taken under the Cash Management System, or as to the use or application by the Applicant of funds transferred, paid, collected or otherwise dealt with in the Cash Management System, shall be entitled to provide the Cash Management System without any liability in respect thereof to any Person (as defined hereinafter) other than the Applicants, pursuant to the terms of the documentation applicable to the Cash Management System, and shall be, in its capacity as provider of the Cash Management System, an unaffected creditor under the Plan with regard to any claims or expenses it may suffer or incur in connection with the provision of the Cash Management System.

- 10. The Applicants may pay the following expenses whether incurred prior to or after this Order
 - all outstanding and future wages, salaries, employee benefits, vacation pay, severance and expenses payable to employees on or after the date of this Order, in each case incurred in the ordinary course of business and consistent with existing compensation policies and arrangements;
 - (b) all existing and future employee health, dental, life insurance, short and long term disability and related benefits (collectively, the "Group Benefits") payable on or after the date of this Order to employees on or after the date of this Order, in each case incurred in the ordinary course of business and consistent with existing policies and arrangements or such amended policies and arrangements as are necessary or desirable to deliver the existing Group Benefits;
 - (c) with prior written approval of the Monitor, the fees and disbursements for any Assistants retained or employed by the Applicant in respect of these proceedings, at their reasonable standard rates and charges; and
 - (d) with the consent of the Monitor, amounts owing for goods or services actually provided to the Applicants prior to the Effective Time by third parties, if, in the opinion of IMV, such third party is critical to the Business and ongoing operations of IMV and such third party would sustain material prejudice if such payment is not made, up to a maximum aggregate amount of \$350,000.
- 11. Except as otherwise provided to the contrary herein, the Applicants may pay all reasonable expenses incurred by the Applicants in carrying on the Business in the ordinary course after this Order, and in carrying out the provisions of this Order, which expenses shall include, without limitation:
 - (a) all expenses and capital expenditures reasonably necessary for the preservation of the Property or the Business including, without limitation, payments on account of insurance including directors and officers insurance, maintenance, and security services; and
 - (b) payment for goods or services actually supplied to the Applicants following the date of this Order.
- 12. The Applicants shall remit or pay, in accordance with legal requirements or on terms as may be agreed to between the Applicants and the applicable authority:
 - (a) any statutory deemed trust amounts in favour of the Crown in right of Canada or of any Province thereof or any other taxation authority which are required to be deducted from employees' wages, including, without limitation, amounts in respect of: i) employment insurance, ii) Canada Pension Plan, iii) Quebec Pension Plan, and iv) income taxes;
 - (b) all goods and services or other applicable sales taxes (collectively, "Sales Taxes") required to be remitted by the Applicants in connection with the sale of goods and services by the Applicants, but only where such Sales Taxes are accrued or collected after the date of this Order; and

- (c) any amount payable to the Crown in right of Canada or of any Province or any regulatory or administrative body or any other authority, in all cases in respect of municipal realty, municipal business, or other taxes, assessments or levies of any nature or kind which are: (i) entitled at law to be paid in priority to claims of secured creditors; (ii) attributable to or in respect of the ongoing Business carried on by the Applicants; and (iii) payable in respect of the period commencing on or after the date of this Order.
- 13. Until such time as the Applicants disclaims or resiliate a real property lease in accordance with the CCAA, the Applicants shall pay all amounts constituting rent or payable as rent under real property leases, including, for greater certainty, common area maintenance charges, utilities and realty taxes, and any other amounts payable to the landlord under the lease, or as otherwise may be negotiated between the Applicants and the landlord from time to time ("**Rent**"), for the period commencing from and including the date of this Order, in accordance with its existing lease agreements. On the date of the first of such payments, any arrears relating to the period commencing from and including the date of this Order shall also be paid.
- 14. Except as specifically permitted herein or by further order of this Court, the Applicants are hereby directed, until further order of this Court: (i) to make no payments of principal, interest thereon or otherwise on account of amounts owing by the Applicants to any of their creditors as of this date without prior written consent of the Monitor; (ii) to grant no security interests, trusts, liens, charges, or encumbrances upon or in respect of any of its Property; and (iii) to not grant credit or incur liabilities except in the ordinary course of the Business or with the prior written approval of the Monitor.

Restructuring

- 15. The Applicants shall, subject to such requirements as are imposed by the Monitor and under any agreements for debtor in possession financing which may be granted, have the right to:
 - (a) permanently or temporarily cease, downsize or shut down any of their operations or locations as they deem appropriate and make provision for the consequences thereof in the Plan;
 - (b) pursue all avenues to finance or refinance, market, convey, transfer, assign or in any other manner dispose of the Business or Property, in whole or part, subject to further order of the Court and sections 11.3 and 36 CCAA, and under reserve of subparagraph (c);
 - (c) convey, transfer assign, lease, or in any other manner dispose of the Property, outside the ordinary course of business, in whole or in part, provided that the price in each case does not exceed \$ 50 000 or \$ 150 000 in the aggregate;
 - (d) terminate the employment of such of their employees or temporarily or permanently lay off such of their employees as they deem appropriate and, to the extent any amounts in lieu of notice, termination or severance pay or other amounts in respect thereof are not paid in the ordinary course, make provision, on such terms as may be agreed upon between the Applicants, as applicable, and such employee, or failing such agreement, make provision to deal with, any consequences thereof in the Plan, as the Applicants may determine;

(e) disclaim or resiliate agreements, subject to the provisions of section 32 CCAA which are as follows:

32 (1) Subject to subsections (2) and (3), a debtor company may — on notice given in the prescribed form and manner to the other parties to the agreement and the monitor — disclaim or resiliate any agreement to which the company is a party on the day on which proceedings commence under this Act. The company may not give notice unless the monitor approves the proposed disclaimer or resiliation.

(2) Within 15 days after the day on which the company gives notice under subsection (1), a party to the agreement may, on notice to the other parties to the agreement and the monitor, apply to a court for an order that the agreement is not to be disclaimed or resiliated.

(3) If the monitor does not approve the proposed disclaimer or resiliation, the company may, on notice to the other parties to the agreement and the monitor, apply to a court for an order that the agreement be disclaimed or resiliated.

(4) In deciding whether to make the order, the court is to consider, among other things,

(a) whether the monitor approved the proposed disclaimer or resiliation;

(b) whether the disclaimer or resiliation would enhance the prospects of a viable compromise or arrangement being made in respect of the company; and

(c) whether the disclaimer or resiliation would likely cause significant financial hardship to a party to the agreement.

(5) An agreement is disclaimed or resiliated

(a) if no application is made under subsection (2), on the day that is 30 days after the day on which the company gives notice under subsection (1);

(b) if the court dismisses the application made under subsection (2), on the day that is 30 days after the day on which the company gives notice under subsection (1) or on any later day fixed by the court; or

(c) if the court orders that the agreement is disclaimed or resiliated under subsection (3), on the day that is 30 days after the day on which the company gives notice or on any later day fixed by the court.

(6) If the company has granted a right to use intellectual property to a party to an agreement, the disclaimer or resiliation does not affect the party's right to use the intellectual property — including the party's right to enforce an exclusive use — during the term of the agreement, including any period for which the party extends the agreement as of right, as long as the party continues to perform its obligations under the agreement in relation to the use of the intellectual property.

(7) If an agreement is disclaimed or resiliated, a party to the agreement who suffers a loss in relation to the disclaimer or resiliation is considered to have a provable claim.

(8) A company shall, on request by a party to the agreement, provide in writing the reasons for the proposed disclaimer or resiliation within five days after the day on which the party requests them.

- (9) This section does not apply in respect of
 - (a) an eligible financial contract;
 - (b) a collective agreement;
 - (c) a financing agreement if the company is the borrower; or
 - (d) a lease of real property or of an immovable if the company is the lessor.

and

(f) subject to section 11.3 CCAA, assign any rights and obligations of the Applicants.

No Proceedings Against the Applicant or the Property

- Until and including the day of July 17, 2023, or such later date as this Court may order 16. (the "Stay Period"), no claim, grievance, application, action, suit, right or remedy, or proceeding or enforcement process in any court, tribunal, or arbitration association (each, a "Proceeding") including but not limited to seizures, right to distrain, executions, writs of seizure or execution, any and all actions, applications, arbitration proceedings and other lawsuits existing at the time of this Order in which any of the Applicants is a defendant, party or respondent (either individually or with other Persons (as defined hereinafter)) shall be commenced, continued, or enforced against or in respect of any of the Applicants or the Monitor, or affecting the Business or the Property, except with the written consent of the Applicants and the Monitor, or with leave of this Court, and any and all Proceedings currently under way against or in respect of the Applicants or affecting the Business or the Property are hereby stayed and suspended pending further order of this Court. The entities related to or affiliated with the Monitor shall also be entitled to the protection, benefits and privileges afforded to the Monitor pursuant to this paragraph.
- 17. The rights of His Majesty in right of Canada and His Majesty in right of a Province are suspended in accordance with the terms and conditions of section 11.09 CCAA.

No Exercise of Rights or Remedies

18. During the Stay Period all rights and remedies, including, but not limited to modifications of existing rights and events deemed to occur pursuant to any agreement to which any of the Applicants is a party as a result of the insolvency of the Applicants and/or these CCAA proceedings, any events of default or non-performance by any of the Applicants as a result of the insolvency of the Applicants and/or these CCAA proceedings or any admissions or evidence in these CCAA proceedings, of any individual, firm, corporation, governmental body or agency, or any other entities (all of the foregoing, collectively being "Persons" and each being a "Person") against or in respect of the Applicants or the Monitor, or affecting the Business or the Property, are hereby stayed and suspended except with the written consent of the Applicants and the Monitor, or leave of this Court, provided that nothing in this Order shall (i) empower the Applicant to carry on any

business which the Applicants are not lawfully entitled to carry on; (ii) affect such investigations, actions, suits or proceedings by a regulatory body as are permitted by section 11.1 of the CCAA; (iii) exempt the Applicants from compliance with statutory or regulatory provisions relating to health, safety, or the environment; (iv) prevent the filing of any registration to preserve or perfect a security interest; or (v) prevent the registration of a claim for lien and the related filing of an action to preserve the right of a lien holder, provided that the Applicants shall not be required to file a defence during the stay period.

No Interference with Rights

19. During the Stay Period, no Person shall discontinue, fail to renew per the same terms and conditions, fail to honour, alter, interfere with, repudiate, terminate, or cease to perform any right, renewal right, contract, agreement, licence, or permit in favour of or held by the Applicants, including but not limited to renewal rights in respect of existing insurance policies on the same terms, except with the written consent of the Applicants and the Monitor, or leave of this Court.

Exemption from Certain Reporting Obligations

20. Any and all continuous disclosure, reporting and filing obligations of, and audit committee requirements applicable to IMV Inc. as a result of its status as a reporting issuer in each of the provinces and territories of Canada subject to Canadian securities laws, rules, regulations and policy statements is hereby suspended until further order of this Court.

Shareholders' Meeting

21. The time limit to call and hold the Applicants' annual shareholders' meeting is extended until after the conclusion of the CCAA Proceedings, subject to further order of this Court.

Continuation of Services

During the Stay Period, all Persons having oral or written agreements with the 22. Applicants or statutory or regulatory mandates for the supply of goods or services, including without limitation all computer software, communication and other data services, centralized banking services, payroll services, insurance, transportation services, utility, or other services to the Business or the Applicants, are hereby restrained until further order of this Court from discontinuing, failing to renew per the same terms and conditions, altering, interfering with, or terminating the supply of such goods or services as may be required by the Applicants, and the Applicants shall be entitled to the continued use of their current premises, telephone numbers, facsimile numbers, internet addresses, and domain names, provided in each case that the normal prices or charges for all such goods or services received after the date of this Order are paid by the Applicants, without having to provide any security deposit or any other security, in accordance with normal payment practices of the Applicants or such other practices as may be agreed upon by the supplier or service provider and each of the Applicants and the Monitor, or as may be ordered by this Court.

Non-Derogation of Rights

23. Notwithstanding anything else contained herein, no Person shall be prohibited from requiring immediate payment for goods, services, use of leased or licensed property, or other valuable consideration provided on or after the date of this Order, nor shall any

Person be under any obligation on or after the date of this Order to advance or readvance any monies or otherwise extend any credit to the Applicants.

Proceedings Against Directors and Officers

24. During the Stay Period, and except as permitted by subsection 11.03(2) of the CCAA, no Proceeding may be commenced or continued against any of the former, current, or future, *de jure* or *de facto*, directors or officers of the Applicants, nor against any person deemed to be a director or an officer of any of the Applicants under subsection 11.03(3) of the CCAA (the "**Directors and Officers**"), with respect to any claim against the directors or officers that arose prior to the Effective Time and that relates to any obligations of the Applicants whereby the directors or officers are alleged under any law to be liable in their capacity as directors or officers for the payment or performance of such obligations, until a compromise or arrangement in respect of the Applicants, if one is filed, is sanctioned by this Court or is refused by the creditors of the Applicants or this Court, these proceedings are dismissed by final order of this Court, or with leave of this Court.

Appointment of Monitor

- 25. FTI Consulting Canada Inc. is hereby appointed pursuant to the CCAA as the Monitor, an officer of this Court, to monitor the Business and financial affairs of the Applicants, the Property, and the Applicants' conduct of the Business with the powers and obligations set out in the CCAA or set forth herein and the Applicants and their shareholders, officers, directors, employees and Assistants shall advise the Monitor of all material steps taken by the Applicants pursuant to this Order, cooperate fully with the Monitor in the exercise of its powers and discharge of its obligations, and provide the Monitor with the assistance that is necessary to enable the Monitor to adequately carry out the Monitor's functions.
- 26. The Monitor, in addition to its prescribed rights and obligations under the CCAA, is hereby directed and empowered to:
 - (a) monitor the Applicants' receipts and disbursements;
 - (b) report to the Court on the state of the business and financial affairs of the Applicants or developments in these proceedings or any related proceedings within the time limits set forth in the CCAA and at such time as considered appropriate by the Monitor or as the Court may order and may file consolidated reports for the Applicants;
 - (c) advise the Applicants in their development of the Plan and any amendments to the Plan, and, to the extent deemed appropriate by the Monitor, assist in their negotiations with creditors, customers, vendors, and other interested Persons;
 - (d) assist the Applicants, to the extent deemed appropriate by the Monitor, with the holding and administering of creditors' or shareholders' meetings for voting on the Plan;
 - (e) retain and employ such agents, advisers and other assistants as are reasonably necessary for the purpose of carrying out the terms of this Order, including, without limitation, one or more entities related to or affiliated with the Monitor;

- (f) have full and complete access to the Property, including the premises, books, records, data, including data in electronic form, and other financial documents and to the Business of the Applicants, to the extent that is necessary to adequately assess the Applicant's Business and financial affairs or to perform its duties arising under this Order;
- (g) be at liberty to engage independent legal counsel or such other persons as the Monitor deems necessary or advisable respecting the exercise of its powers and performance of its obligations under this Order, including any affiliate of, or person related to the Monitor;
- (h) develop a claims process to ascertain the quantum of the claims of all creditors; and
- (i) be at liberty to perform such other duties as are required by this Order or by this Court from time to time.
- 27. The Monitor shall, as soon as practicable, (i) publish once a week for two (2) consecutive weeks or as otherwise directed by the Court, in La Presse+, the Globe & Mail National Edition and the Wall Street Journal and (ii) within five (5) business days after the date of this Order (A) post on the Monitor's website (the "**Website**") a notice containing the information prescribed under the CCAA, (B) make this Order publicly available in the manner prescribed under the CCAA, (C) send, in the prescribed manner, a notice to all known creditors having a claim against the Debtors of more than \$1,000, advising them that this Order is publicly available, and (D) prepare a list showing the names and addresses of such creditors and the estimated amounts of their respective claims, and make it publicly available in the prescribed manner, all in accordance with Section 23(1)(a) CCAA and the regulations made thereunder;
- 28. The Monitor shall not take possession of the Property and shall take no part whatsoever in the management or supervision of the management of the Business and shall not, by fulfilling its obligations hereunder, be deemed to have taken or maintained possession or control of the Business or Property, or any part thereof.
- 29. Nothing herein contained shall limit the protections afforded the Monitor at law including those protections set out in the CCAA.
- 30. The Monitor may provide, with copy to counsel to the Applicants, creditors and other relevant stakeholders of the Applicants with information in response to reasonable requests for information made in writing by such creditor or relevant stakeholder addressed to the Monitor. The Monitor shall not have any responsibility or liability with respect to the information disseminated by it pursuant to this paragraph. In the case of information that the Monitor has been advised by the Applicants is confidential, the Monitor shall not provide such information to creditors or other relevant stakeholders unless otherwise directed by this Court or on such terms as the Monitor and the Applicants may agree.
- 31. The Monitor, counsel to the Monitor, and all counsel to the Applicants shall be paid their reasonable fees and disbursements, in each case not to exceed their standard rates and charges, by the Applicants as part of the costs of these proceedings. The Applicants are hereby authorized and directed to pay the accounts of the Monitor, counsel for the Monitor (Stikeman Elliott LLP) and counsel for the Applicants (McCarthy Tétrault LLP, Stewart McKelvey Lawyers and Troutman Pepper Hamilton Sanders LLP) on a weekly

basis and, in addition, the Applicants are authorized to provide each with a reasonable retainer in advance on account of such fees and disbursements, if so requested.

- 32. The Monitor and its legal counsel shall pass their accounts from time to time before a judge of this court or a referee appointed by a judge.
- 33. The Monitor, counsel to the Monitor and the Applicants' counsel shall be entitled to the benefit of and are hereby granted a charge, security and hypothec (the "Administration Charge") on the Property, which charge shall not exceed an aggregate amount of \$750,000, as security for their professional fees and disbursements incurred at the standard rates and charges of the Monitor and such counsel, both before and after the making of this Order in respect of these proceedings. The Administration Charge shall have the priority set out in paragraphs 39 and 41 herein.

Directors' Charge

- 34. The Applicant shall indemnify their Directors and Officers against obligations and liabilities that they may incur as directors or officers of the Applicants after the commencement of the within proceedings, except to the extent that, with respect to any officer or director, the obligation or liability was incurred as a result of the Directors or Officers' gross negligence or wilful misconduct.
- 35. The Directors and Officers of the Applicants shall be entitled to the benefit of and are hereby granted a charge, security and hypothec (the "**Directors' Charge**") on the Property, which charge shall not exceed an aggregate amount of \$275,000, as security for the indemnity provided in paragraph 34 of this Order. The Directors' Charge shall have the priority set out in paragraphs 39 and 41 herein.
- 36. Notwithstanding any language in any applicable insurance policy to the contrary, (i) no insurer shall be entitled to be subrogated to or claim the benefit of the Directors' Charge, and (ii) the Directors and Officers shall only be entitled to the benefit of the Directors' Charge to the extent that they do not have coverage under any directors' and officers' insurance policy, or to the extent that such coverage is insufficient to pay amounts indemnified in accordance with paragraph 34 of this Order.

Key Employee Retention Plan

- 37. The Key Employee Retention Plan (the "**KERP**") attached as **Schedule B** to this Order (with its schedules filed under seal) is hereby approved and the Applicants are authorized to make peyments contemplated thereunder in accordance with the terms and conditions of the KERP with such minor alterations, changes, amendments, deletions or additions made thereto, with the consent of the Monitor, save and except with respect to the quantum of the KERP.
- 38. The employees referred to in the KERP (the "**Key Employees**") shall be entitled and are hereby granted a charge on the Property (the "**KERP Charge**") in the aggregate amount of \$575,000, to secure any payments to the Key Employees under the KERP. The KERP Charge shall have the priority set out in paragraphs 39 and 41 herein.

Validity and Priority of the Charges Created by this Order

39. The priorities of the Administration Charge, the Directors' Charge and the KERP Charge, as among them, with respect to the Property to which they apply shall be as follows:

First – Administration Charge (to the maximum amount of \$750,000);

Second – Directors' Charge (to the maximum amount of \$275,000); and

Third – KERP Charge (to the maximum amount of \$575,000).

- 40. The filing, registration or perfection of the Administration Charge, the Directors' Charge or the KERP Charge (collectively, the "**Charges**") shall not be required, and the Charges shall be valid and enforceable for all purposes, including as against any right, title or interest filed, registered, recorded or perfected subsequent to the Charges coming into existence, notwithstanding any such failure to file, register, record or perfect.
- 41. Each of the Administration Charge, the Directors' Charge and the KERP Charge (all as constituted and defined herein) shall constitute a charge, security and hypothec on the Property and such Charges shall rank in priority to all other hypothecs, mortgages, security interests, priorities, trusts (including deemed or constructive trusts), liens, charges and encumbrances, claims of secured creditors, statutory or otherwise, of whatever nature or kind (collectively, the "Encumbrances") in favour of any Person.
- 42. Except as otherwise expressly provided for herein, or as may be approved by this Court, the Applicants shall not grant any Encumbrances over any Property that rank in priority to, or *pari passu* with, any of the Administration Charge, the Directors' Charge or the KERP Charge, unless the Applicants obtains the prior written consent of the Monitor, the Secured Lenders and the beneficiaries of the Administration Charge, the Directors' Charge or the Directors' Charge or further Order of this Court.
- 43. The Charges shall not be rendered invalid or unenforceable and the rights and remedies of the chargees entitled to the benefit of the Charges (collectively, the "**Chargees**") thereunder shall not otherwise be limited or impaired in any way by (i) the pendency of these proceedings and the declarations of insolvency made herein; (ii) any application(s) for bankruptcy order(s) issued pursuant to BIA, or any bankruptcy order made pursuant to such applications; (iii) the filing of any assignments for the general benefit of creditors made pursuant to the BIA; (iv) the provisions of any federal or provincial statutes; or (v) any negative covenants, prohibitions or other similar provisions with respect to borrowings, incurring debt or the creation of Encumbrances, contained in any existing loan documents, lease, sublease, offer to lease or other agreement (collectively, an "**Agreement**") which binds the Applicants, and notwithstanding any provision to the contrary in any Agreement:
 - (a) neither the creation of the Charges nor the execution, delivery, perfection, registration or performance same shall create or be deemed to constitute a breach by the Applicants of any Agreement to which it is a party;
 - (b) none of the Chargees shall have any liability to any Person whatsoever as a result of any breach of any Agreement caused by or resulting from the creation of the Charges; and

(c) the payments made by the Applicants pursuant to this Order and the granting of the Charges, do not and will not constitute preferences, fraudulent conveyances, transfers at undervalue, oppressive conduct, or other challengeable or voidable transactions under any applicable law.

Sale and Investment Solicitation Process

- 44. The sale and investment solicitation process ("SISP") procedures attached as **Schedule C** to this Order (the "SISP Procedures") are hereby approved, and the Monitor is hereby authorized to commence the SISP in accordance with the terms and conditions of the SISP Procedures. The Monitor and the Applicants are hereby authorized and directed to perfrom their respective obligations and to do all things reasonably necessary to perform their obligations under the SISP Procedures.
- 45. The Monitor and its affiliates, partners, directors, employees, agents and controlling persons shall have no liability with respect to any and all losses, claims, damages or liabilities, of any nature or kind, to any person in connection with or as a result of the conducting of the SISP, including any steps taken by the Monitor prior to the approval of the SISP and described in the First Report of the Monitor which are hereby approved *nunc pro tunc*, except to the extent such losses, claims, damages or liabilities result from the gross negligence or wilful misconduct of the Monitor in performing its obligations under the SISP Procedures, as determined by this Court.
- 46. Pursuant to clause 7(3)(c) of the Canada Personal Information Protection and Electronic Documents Act, the Applicants, the Monitor and their respective advisors are hereby authorized and permitted, subject to the execution of a non-disclosure agreement, to disclose and transfer to prospective SISP participants (each, a "SISP Participant") and their advisors personal information of identifiable individuals but only to the extent desirable or required to negotiate or attempt to complete a transaction pursuant to the SISP (a "Transaction"). Each SISP Participant to whom such personal information is disclosed shall maintain and protect the privacy of such information and limit the use of such information to its evaluation for the purpose of effecting a Transaction, and if it does not complete a Transaction, shall return all such information to the Applicants or the Monitor, or in the alternative destroy all such information and provide confirmation of its destruction if requested by the Applicants or the Monitor. Any Successful Party shall maintain and protect the privacy of such information and, upon closing of the Transaction(s) contemplated in the Successful Bid(s), shall be entitled to use the personal information provided to it that is related to the Business and/or Property acquired pursuant to the SISP in a manner that is in all material respects identical to the prior use of such information by the Applicants, and shall return all other personal information to the Applicants or the Monitor, or ensure that all other personal information is destroyed and provide confirmation of its destruction if requested by the Applicants or the Monitor.

Service and Notice

- 47. Subject to further Order of this Court, all applications in these CCAA proceedings are to be brought on not less than five (5) days' notice to all Persons on the service list. Each application shall specify a date (the "Initial Hearing Date") and time (the "Initial Hearing Time") for the hearing.
- 48. Any Person wishing to object to the relief sought on an application in these CCAA proceedings must serve a detailed written contestation stating the objection to the

application and the grounds for such objection (a "**Contestation**") in writing to the moving party, the Applicants and the Monitor, with a copy to all Persons on the service list, no later than 5 p.m. Halifax time on the date that is three (3) days prior to the Initial Hearing Date (the "**Objection Deadline**").

- 49. If no Contestation is served by the Objection Deadline, the Judge having carriage of the application (the "**Presiding Judge**") may determine: (i) whether a hearing is necessary; (ii) whether such hearing will be in person, virtually, by telephone or by written submissions only; and (iii) the parties from whom submissions are required (collectively, the "**Hearing Details**"). In the absence of any such determination, a hearing will be held in the ordinary course.
- 50. If no Contestation is served by the Objection Deadline, the Applicants shall communicate with the Presiding Judge regarding whether a determination has been made by the Presiding Judge concerning the Hearing Details. The Applicants shall thereafter advise the service list of the Hearing Details and the Applicants shall report upon its dissemination of the Hearing Details to the Court in a timely manner.
- 51. If a Contestation is served by the Objection Deadline, the interested parties shall appear before the Presiding Judge on the Initial Hearing Date at the Initial Hearing Time, or such earlier or later time as may be directed by the Court, to, as the Court may direct: (i) proceed with the hearing on the Initial Hearing Date and at the Initial Hearing Time; or (ii) establish a schedule for the delivery of materials and the hearing of the contested application and such other matters, including interim relief, as the Court may direct.
- 52. Except as otherwise specified herein, the Applicants and the Monitor are at liberty to serve any notice, proof of claim form, proxy, circular or other document in connection with these CCAA proceedings by forwarding copies by prepaid ordinary mail, courier, personal delivery or electronic transmission to Persons or other appropriate parties at their respective given addresses as last shown on the records of the Applicants and that any such service shall be deemed to be received on the date of delivery if by personal delivery or electronic transmission, on the following business day if delivered by courier, or three business days after mailing if by ordinary mail.
- 53. The Applicants and the Monitor may give notice of this Order, any other materials and orders in these CCAA proceedings, and any notices, and provide correspondence, by forwarding PDF copies originals or true copies by prepaid ordinary mail, courier, personal delivery, or electronic transmission, including by email, to the Applicants' creditors or other interested parties at their respective addresses as last shown on the records of the Applicants and any such notice by courier, personal delivery, or electronic transmission shall be deemed to be received on the next business day following the date of forwarding thereof, or if sent by ordinary mail, on the third business day after mailing.
- 54. The Applicants and the Monitor, and any party who has filed a demand of notice may serve any court materials in these proceedings by e-mailing a PDF or other electronic copy of such materials to counsel's e-mail addresses as recorded on the service list from time to time, and the Monitor may post a copy of any or all such materials on its website at http://cfcanada.fticonsulting.com/imv.
- 55. Unless otherwise provided herein, under the CCAA, or ordered by this Court, no document, order or other material need be served on any Person in respect of these CCAA proceedings, unless such Person has filed a demand of notice, or appears on the

service list prepared by counsel for the Monitor, save and except when an order is sought against a Person not previously involved in these CCAA proceedings.

General

- 56. The Applicants or the Monitor may, from time to time, apply to this Court for directions concerning the exercise of their respective powers, duties and rights hereunder or in respect of the proper execution of this Order.
- 57. Nothing in this Order shall prevent the Monitor from acting as an interim receiver, a receiver, a receiver and manager, construction lien trustee, or a trustee in bankruptcy of the Applicants, the Business or the Property.
- 58. This Order and all other orders in these proceedings shall have full force and effect in all provinces and territories in Canada.
- 59. IMV Inc. is authorized to apply as it may consider necessary or desirable, with or without notice, to any other court or administrative body, whether in Canada, the United States of America, or elsewhere, for orders which aid and complement this Order and any subsequent orders of this Court and, without limitation to the foregoing, any orders under Chapter 15 of the United States Bankruptcy Code, including an order for recognition of these CCAA proceedings as "Foreign Main Proceedings" in the United States of America pursuant to Chapter 15 of the United States Bankruptcy Code, and for which IMV Inc. shall be the foreign representative of the Applicants (the "Foreign Representative"). All courts and administrative bodies of all such jurisdictions are hereby respectively requested to make such orders and to provide such assistance to the Applicants and the Foreign Representative as may be deemed necessary or appropriate for that purpose.
- 60. The aid and recognition of any Court, tribunal, regulatory or administrative body in Canada, the United States of America or elsewhere, to give effect to this Order and to assist the Applicants, the Monitor and their respective agents in carrying out the terms of this Order. All Courts, tribunals, regulatory and administrative bodies are hereby respectfully requested to make such orders and to provide such assistance to the Applicants, and the Monitor as may be necessary or desirable to give effect to this Order, to grant representative status to the Monitor or the authorized representative of the Applicants in any foreign proceeding, to assist the Applicants, and the Monitor, and to act in aid of and to be complementary to this Court, in carrying out the terms of this Order.
- 61. For the purposes of any applications authorized by paragraphs 59 and 60, Applicants' *"centre of main interest"* is located in the province of Nova Scotia, Canada.
- 62. Each of the Applicants and the Monitor may apply to any court, tribunal, or regulatory or administrative body, wherever located, for the recognition of this Order and for assistance in carrying out the terms of this Order, and the Monitor may act as a representative in respect of the within proceedings for the purpose of having these proceedings recognized in a jurisdiction outside Canada.
- 63. The schedules to the Key Employee Bonus Agreement (Schedule B to this Order) shall be and remain sealed and kept confidential until further Order of this Court.

64. Any interested party, including the Applicant and the Monitor, may apply to this Court to vary or amend this Order on such notice required under the *Civil Procedure Rules* or as this Court may order.

Issued

, 2023

Prothonotary

Schedule "A" – Service List

SUPREME COURT OF NOVA SCOTIA

IN THE MATTER OF: Application by IMV Inc., Immunovaccine Technologies Inc. and IMV USA Inc. (the "Applicants"), for relief under the *Companies' Creditors Arrangement Act*

SERVICE LIST

Party	Method of Service
McCarthy Tétrault LLP 1000 De La Gauchetière Street West Suite MZ400 Montréal, QC H3B 0A2 <i>Counsel to the Applicants.</i>	M ^{tre} Alain N. Tardif Tel : 514.397.4274 E-mail : atardif@mccarthy.ca M ^{tre} François Alexandre Toupin Tel: 514.397.4210 E-mail: fatoupin@mccarthy.ca
Stewart McKelvey Lawyers Queen's Marque 600-1741 Lower Water Street Halifax, NS B3J 0J2 <i>NS Counsel to the Applicants</i>	M^{tre} Sara L. Scott Tel : 514.397.4274 E-mail : sscott@stewartmckelvey.com
Aird & Berlis LLP Brookfield Place 181 Bay Street, Suite 1800 Toronto, ON M5J 2T9 <i>Counsel to the Horizon Technology Finance</i> <i>Corporation, as agent</i>	M ^{tre} Miranda Spence Tel : 416.865.3414 E-mail : mspence@airdberlis.com M ^{tre} Graham Topa Tel: 416.865.4739 E-mail : gtopa@airdberlis.com M ^{tre} Kyle Plunkett Tel: 416.865.3406 E-mail : kplunkett@airdberlis.com
Ropes & Gray LLP Prudential Tower 800 Boylston Street Boston, MA 02199-3600 US Counsel to the secured creditor Horizon Technology Finance Corporation, as agent	M ^{tre} Cristine Schwarzman Tel: 212.596.9635 E-mail : Cristine.Schwarzman@ropesgray.com M ^{tre} Patricia I. Chen Tel : 617.951.7553 E-mail: patricia.chen@ropesgray.com

FTI Consulting Canada Inc. TD South Tower 79 Wellington Street West Suite 2010, P.O. Box 104 Toronto, ON M5K 1G8 <i>Monitor</i>	Jodi Porepa Tel : 437.332.5743 E-mail : Jodi.Porepa@fticonsulting.com Jeffrey Rosenberg Tel: 416.649.8073 E-mail: Jeffrey.Rosenberg@fticonsulting.com Adsaran Vithiyananthan E-mail : Adsaran.Vithiyananthan@fticonsulting.com
Stikeman Elliot LLP 5300 Commerce Court West 199 Bay Street Toronto ON M5L 1B9	M ^{tre} Maria Konyukhova Tel : 416.869.5230 E-mail : mkonyukhova@stikeman.com <i>Counsel to the Appointed Monitor</i>
Atlantic Canada Opportunities Agency Legal Services Department of Justice / Government of Canada PO Box 6051 644 Main Street Moncton, NB E1C 9J8	M^{tre} Karine LeBlanc Tel: 506.851.2153 Email: <u>karine.leblanc@acoa-apeca.gc.ca</u>
Canada Revenue Agency	General address for service notificationPGC-AGC.fiscal-tax@justice.gc.ca notificationPGC-AGC.civil@justice.gc.ca

Email distribution list:

<u>Atardif@mccarthy.ca; fatoupin@mccarthy.ca; sscott@stewartmckelvey.com;</u> <u>Adsaran.Vithiyananthan@fticonsulting.com; Jodi.Porepa@fticonsulting.com;</u> <u>Jeffrey.Rosenberg@fticonsulting.com; mkonyukhova@stikeman.com;</u> <u>mspence@airdberlis.com; gtopa@airdberlis.com; kplunkett@airdberlis.com;</u> <u>Cristine.Schwarzman@ropesgray.com; patricia.chen@ropesgray.com; karine.leblanc@acoaapeca.gc.ca; notificationPGC-AGC.civil@justice.gc.ca; notificationPGC-AGC.fiscaltax@justice.gc.ca</u> Schedule "B" – KERP

KEY EMPLOYEE BONUS AGREEMENT (THE "AGREEMENT") dated as of May •, 2023

BETWEEN:	IMV Inc. , a corporation incorporated under the <i>Canada Business Corporations Act;</i>
AND:	Immunovaccine Technologies Inc., a corporation incorporated under the <i>Companies' Act</i> (Nova Scotia); (the "Canadian Employer")
AND	IMV USA Inc. , a corporation incorporated under the laws of Delaware, United States;
	(the " US Employer " and together with IMV Inc. and the Canadian Employer, the " Company ")
AND:	Each person other than the Company having executed this Agreement
	(each, a "Participant" and collectively, the "Participants")

WHEREAS:

- A. Each Participant currently holds the position set out in each Participant's signature page with the Company;
- B. The Company acknowledges that each Participant is essential in ensuring the continued success and growth of the Company's business and is key to the conclusion of a Transaction (as defined hereinafter) beneficial to all of the Company's stakeholders;
- C. Taking into account these considerations, the Company wishes to offer each Participant certain incentives and payments detailed herein; and
- D. Each Participant's terms of employment with the Company shall continue in full force and effect and unamended, save as provided in this Agreement.

NOW, THEREFORE, the parties hereto agree as follows:

1. Confidentiality

Each Participant recognizes and acknowledges that the incentives and payments provided for herein are strictly confidential and that he (or she) may not discuss the contents of this Agreement, or the incentives and payments described herein, with anyone except his or her legal and financial advisors, and spouse if any (and, then, only if they agree first to preserve such confidentiality).

2. Key Employee Bonus

(a) Subject to the terms set forth in this Agreement, each Participant will receive his/her share as set out in each Participant's signature page of a key employee bonus of an amount up to USD \$425,000 ("Key Employee Bonus"), which will be payable by the Company on the earlier date between (A) the occurrence of a Liquidity Event and (B) October 31, 2023 (the "Payment Date").

- (b) Subject to the terms of this Agreement, certain members of senior management will, in addition to the Key Employee Bonus, receive a bonus equivalent to the percentage set out in such Participant's signature page of the gross consideration received by the Company in connection with the closing of a Transaction, if any ("Transaction Bonus").
- (c) "Liquidity Event" means (A) the closing of a Transaction, (B) the completion of the winding up of the Company's operations, including its ongoing clinical trials, (C) an assignment in bankruptcy under the *Bankruptcy and Insolvency Act* (Canada) (the "BIA") (D) the termination of the Participant's employment by the Canadian Employer or the US Employer, as applicable, without just cause and/or serious reason between the date of this Agreement and the Payment Date or (E) an order of the Court authorizing the payment of the Key Employee Bonus.
- (d) "Transaction" means (A) the direct or indirect sale of all or a majority of the equity securities of the Company to a third party (an "Acquirer"), (B) the merger or combination of the Company or a member of the Company Group with an Acquirer or (C) an Acquirer's acquisition of all or a significant portion of the assets, properties or business of the Company and the other members of the Company Group. "Transaction" shall include any sale or disposition (regardless of form) of all or more than 2/3 of the equity securities, assets, properties or business of the Companies' Creditors Arrangement Act (Canada) (the "CCAA"), the BIA or any federal or provincial corporate laws in Canada, or compromise or arrangement of the Company Group's debt obligations. For greater certainty, a Transaction does not involve the acquisition of the Company's business by the Company's secured lender pursuant to a credit bid.

3. Conditions

- (a) Subject to the terms of this Agreement, the Key Employee Bonus is not earned until paid. Further, in order to be eligible to receive the Key Employee Bonus a Participant must be "Actively Employed" by the Canadian Employer or the US Employer, as applicable, on the Payment Date. "Actively Employed" means, subject to subparagraph 3(b) and the minimum applicable requirements (if any) of applicable employment standards legislation, that the Participant has not resigned, given notice of resignation, been terminated, or been given notice of termination. In the event of termination of the Participant's employment by the Canadian Employer or the US Employer, as applicable, "Actively Employed" shall not, subject only to the minimum requirements (if any) of applicable employment standards legislation, include any notice of termination period to which the Participant may be entitled under statute, contract, common law, or civil law.
- (b) Notwithstanding the foregoing paragraph, in the event of termination of the Participant's employment by the Canadian Employer or the US Employer, as applicable, without just cause (or serious reason for Quebec employees) between the date of this Agreement and the Payment Date, the Participant will be entitled to his/her full share of the Key Employee Bonus in accordance with Section 2.
- (c) However, if the Participant voluntarily resigns, provides notice of resignation, is given notice of termination for cause, or is terminated for just cause (or serious

reason for Quebec employees) at any time between the date of this Agreement and the Payment Date, the Participant will not, subject only to the minimum requirements (if any) of applicable employment standards legislation, be entitled to receive his/her share of the Key Employee Bonus or any damages or payments in lieu of such Key Employee Bonus, including, without limitation, any damages for wrongful dismissal.

- (d) Further, the Participant's performance with the Company must be fully satisfactory between now and the Payment Date, as determined by the Company in its sole discretion.
- (e) Without limiting the generality of the foregoing, the Participant agrees to continue to fully perform his or her duties and responsibilities in an honest, diligent and efficient manner and in accordance with the terms of the Participant's employment agreement with the Canadian Employer or the US Employer, as applicable, and current employment practices (the "Employment Agreement") until the Payment Date.

4. Conditions For the Payment of the Transaction Bonus

- (a) Subject to the terms of this Agreement, the Transaction Bonus is not earned until consideration pursuant to any Transaction is paid and received by the Company. Further, in order to be eligible to receive the Transaction Bonus a Participant must be "Actively Employed" by the Canadian Employer or the US Employer, as applicable, on the closing of a Transaction. "Actively Employed" means, subject to subparagraph 4(b), that the Participant has not resigned, given notice of resignation, been terminated, or been given notice of termination. In the event of termination of the Participant's employed" shall not, subject only to the minimum requirements (if any) of applicable employment standards legislation, include any notice of termination period to which the Participant may be entitled under statute, contract, common law, or civil law.
- (b) Notwithstanding the foregoing paragraph, in the event of termination of the Participant's employment by the Canadian Employer or the US Employer, as applicable, without just cause (or serious reason for Quebec employees) between the date of this Agreement and the closing of a Transaction, the Participant will be entitled to his/her full share of the Transaction Bonus in accordance with Section 2.
- (c) Further, the Participant's performance with the Company must be fully satisfactory between now and the closing of the Transaction, as determined by the Company in its sole discretion.
- (d) Without limiting the generality of the foregoing, the Participant agrees to continue to fully perform his or her duties and responsibilities in an honest, diligent and efficient manner and in accordance with the terms of the Participant's Employment Agreement until the closing of a Transaction.

5. Deductions and Taxes

- (a) Any amount payable pursuant to the terms hereof shall be less all applicable withholdings and deductions required by law.
- (b) The Key Employee Bonus and the Transaction Bonus are not included as eligible income for the purpose of calculating any employer benefit or incentive programs, including but not limited to Group Insurance Benefits, Group RRSP, and Employee Share Purchase Plan.

6. Severance

- (a) The Participants acknowledge and agree that they shall have the right to file a claim for unpaid severance against the Company Group in the context of the claims process to be approved by the Nova Scotia Supreme Court pursuant to the CCAA, which claim shall be limited to 50% of the contractual severance provided in the Employment Agreement or the amount provided for in a Participant's signature page, as the case may be, and shall be accepted as a provable claim for such amount.
- (b) The Participants further acknowledge and agree that this Agreement serves as an appropriate and sufficient written notice of termination of the Employment Agreement from the Canadian Employer or the US Employer, as applicable, between now and the Payment Date.

7. Other Provisions

- (a) As a Company employee, each Participant agrees to fully perform his or her duties and responsibilities in an honest, diligent and efficient manner and in accordance with the terms of the Participant's Employment Agreement.
- (b) This Agreement and all payments made thereunder constitute a full and final settlement and transaction of any and all claims or recourses related to, directly or indirectly, the Participants' employment and termination thereof. Subject to the terms of this Agreement, including for greater certainty section 6(a) hereof, the Participant irrevocably renounces to any and all claims, actions or recourses against the Company and the Acquirer, as the case may be, notably and without limitation, pursuant to the Labour Standards Code (Nova Scotia), the Human Rights Act (Nova Scotia), the Occupational Health and Safety Act (Nova Scotia), the Act respecting labour Standards (Québec), the Civil Code of Quebec, the Charter of human rights and freedoms (Québec), the Act respecting industrial accidents and occupational diseases (Québec) and their respective regulations or any other similar laws applicable in the Participants' specific jurisdiction.
- (c) Other than with regard to any payment in connection with a Transaction, in the event of any conflict between the terms of this Agreement and the terms of the Employment Agreement, the terms of the Employment Agreement shall prevail.
- (d) No amendment or waiver of this Agreement shall be effective unless in writing signed by the Canadian Employer, the US Employer and the Participants.

- (e) This Agreement shall be governed by and interpreted and enforced in accordance with the laws of the Province of Québec and the federal laws of Canada applicable therein.
- (f) The provisions of this Agreement shall be binding on and enure to the benefit of the undersigned and their respective successors and permitted assigns.
- (g) This Agreement may be executed in any number of counterparts and all of such counterparts taken together shall be deemed to constitute one and the same instrument.

[Signature pages follow]

- S-1 -

IMV Inc.

Per:

Name:

Immunovaccine Technologies Inc.

Per:

Name:

IMV USA Inc.

Per:

Name:

Name	
Position	
Key Employee Bonus	
Transaction Bonus	
CCAA Claim	

I, •, having had a reasonable and sufficient opportunity to ask questions and seek independent legal advice, understand and accept the above Agreement.

I have requested that this Agreement be drafted in English. J'ai requis que la présente entente soit rédigée en anglais.

PARTICIPANT:

•

Schedule "C" – SISP Procedures

PROCEDURES OF THE SALE AND INVESTMENT SOLICITATION PROCESS IMV INC. ET AL

Recitals

- A. On May 1st, 2023, IMV Inc., Immunovaccine Technologies Inc. and IMV USA Inc. (collectively, "IMV") obtained protection from their creditors under the *Companies' Creditors Arrangement Act*, R.S.C. 1985, c. C-36 (as amended, the "CCAA") pursuant to the provisions of an order (as amended and restated on May 5, 2023, and as it may be further amended, restated or supplemented from time to time, the "Initial Order") of the Nova Scotia Supreme Court (the "Court").
- B. Pursuant to the Initial Order, FTI Consulting Canada Inc. was appointed by the Court as monitor (in its capacity as monitor and not in its personal capacity, the "**Monitor**") of IMV to the proceedings under the CCAA commenced by the Initial Order, in Court file Hfx No. 523334 (the "**CCAA Proceedings**").
- C. Pursuant to paragraphs [44] to [46] of the Amended and Restated Initial Order dated May 5, 2023 (as it may be amended, restated or supplemented from time to time, the "**SISP Approval Order**"), the Court approved a sale and investment solicitation process to be conducted in respect of the business and/or assets of IMV (the "**SISP**"), in accordance with the procedures, terms and conditions set out herein (the "**SISP Procedures**").
- D. The SISP Procedures sets out the manner in which (i) bids and proposals for a broad range of executable transaction alternatives (including, without limitation, a sale of assets and/or shares, restructuring, recapitalization and/or refinancing) involving the business of IMV, as more particularly described in the Teaser Letter (the "**Business**"), and the property, assets and undertaking of IMV (the "**Property**"), whether *en bloc* or any portion(s) thereof, will be solicited from interested parties, (ii) any bids received will be negotiated, (iii) any Successful Bid(s) will be selected and, (iv) the Court's approval of any Successful Bid(s) will be sought.
- E. An investment in the Business may involve, among other things, a restructuring, recapitalization, or other form of reorganization of the business and affairs of the Business or any part thereof, and such investment may be consummated pursuant to a plan of compromise or arrangement (a "**Plan**"), a reverse vesting order, an arrangement pursuant to the *Canada Business Corporations Act*, R.S.C., 1985, c. C-44 (respectively an "**Arrangement**" and the "**CBCA**") or otherwise.
- F. The SISP Approval Order, the SISP Procedures, and any other orders of the Court made in the CCAA Proceedings relating to the SISP shall exclusively govern the process for soliciting and selecting bids for the sale of the Property or investment in the Business pursuant to a broad range of executable transaction alternatives.
- G. Unless otherwise stated or unless the subject matter or context otherwise requires, the capitalized terms used in the SISP Procedure have the meaning ascribed to them at **Schedule A** hereof.

Section 1. Conduct of the SISP

- 1.1 Before the SISP is launched and before any documentation is used in the context of the SISP, the Monitor will provide IMV and the Lenders' Agent with copies of the documents which will be used, including the Teaser Letter, the Contact List, the Confidentiality Agreement etc., the whole for comments and approval by IMV and the Lenders' Agent.
- 1.2 The SISP will be carried out by the Monitor with the assistance of Stonegate, as the case may be, and in consultation with IMV and the Lenders' Agent. Unless otherwise provided for herein, and in accordance with the SISP Approval Order, the Monitor is fully and exclusively authorized, empowered and directed to take any and all actions and steps pursuant to the SISP, subject to the terms and conditions contained herein.
- 1.3 The Monitor, with the assistance of Stonegate and IMV, shall be responsible for contacting Prospective Bidders, communicating with Prospective Bidders and Bidders, negotiating with Bidders, providing them with the Teaser Letter, coordinating the execution of any Confidentiality Agreements executed by a Prospective Bidder pursuant to the SISP, managing the process of answering all reasonable inquiries from Prospective Bidders and Bidders and Bidders and Bidders and Bidders and arranging for visits, when applicable, by Bidders.
- 1.4 The Monitor, in consultation with IMV and the Lenders' Agent, shall review and assess LOIs and Bids.
- 1.5 At any time during the SISP, the Monitor may consult with IMV, the Lenders' Agent and such parties as it considers appropriate in respect of the conduct of the SISP.
- 1.6 After the issuance of the SISP Approval Order, the Monitor may at any time and from time to time and in consultation with IMV and with the consent of the Lenders' Agent, modify, amend, vary or supplement the SISP or the SISP Procedures, without the need for obtaining an order of the Court or providing notice to any Bidders, provided that the Monitor determines that such modification, amendment, variation or supplement would not be materially prejudicial to the interests of the Prospective Bidders and Bidders and is necessary or useful in order to give effect to the substance of the SISP, the SISP Procedures or the SISP Approval Order. The Monitor shall post on the Monitor's Website, as soon as possible, any such modification, amendment, variation or supplement to the SISP Procedures and inform Prospective Bidders and Bidders reasonably impacted by any such modification, amendment, variation or supplement to the SISP Procedures.
- 1.7 The Monitor, may, at any time and on notice to the service list in the CCAA Proceedings as posted on the Monitor's website, as it may be updated from time to time (the "**Service List**"), apply to the Court for directions in connection with the implementation of the SISP or the SISP Procedures.
- 1.8 The implementation of the SISP by the Monitor shall commence at the Commencement Date. In the event that the Monitor, in consultation with IMV and the Lenders' Agent, considers it necessary or appropriate to postpone the Commencement Date, it shall select a new Commencement Date to be published on the Monitor's website and notified forthwith to the Service List.

Section 2. Sale or Investment Opportunities

- 2.1 Qualified Bidders will have the opportunity to submit a bid consisting in either a Sale Proposal or an Investment Proposal. Sale Proposals and Investment Proposals may be in respect of only some of the Property and any such proposal will not be precluded from consideration as an acceptable Bid.
- 2.1 In the event of a Sale Proposal for any or all of the Property, all of IMV's relevant right, title and interest in and to the Property may be acquired pursuant to an approval and vesting order of the Court, including pursuant to a reverse vesting order, free and clear of all pledges, liens, security interests, charges, options, hypothecs, mortgages and interest thereon, except to the extent otherwise set forth in a definitive purchase agreement executed with a Successful Bidder.
- 2.2 In the event of an Investment Proposal for any or all of the Business, same can be implemented by way of a combined Plan and Arrangement.

Section 3. "As is, Where Is"

- 3.1 Any Sale Proposal or Investment Proposal (either being a "**Proposal**") shall be made on an "as is, where is" basis, without surviving representations or warranties of any kind or nature.
- 3.2 IMV and the Monitor are not responsible for, and will have no liability with respect to, any information obtained by any Prospective Bidder or Bidder in connection with the Business or Property. IMV, the Monitor and their advisors, as applicable, do not make any representations or warranties whatsoever as to the information or the materials provided through the due diligence process or otherwise made available to any Prospective Bidder and Bidder including any information contained in the Teaser Letter or Data Room.

Section 4. Solicitation of Interest

- 4.1 The Monitor will by no later than **5:00 p.m. (Halifax Time)** on **May 15, 2023**, or such later date or time as the Monitor, in accordance with the SISP Procedures, may determine appropriate (the "**Commencement Date**"):
 - (a) compile a listing (the "Contact List") of prospective purchasers and investors (collectively, "Prospective Bidders"). The Monitor will use all reasonable commercial efforts to contact all parties identified in the Contact List as well as any additional parties identified as prospective purchasers or investors;
 - (b) post of a copy of the SISP Approval Order and the SISP Procedures on the Monitor's website;
 - (c) determine the appropriate advertising to be directed at Prospective Bidders, which may include newspaper, trade publication, internet or other advertising;
 - (d) send to each Prospective Bidder teaser materials, including a solicitation letter summarizing the acquisition and investment opportunity with respect to the Business and Property (the "Teaser Letter");

- (e) set up and update an electronic data room with confidential information in respect of the Business and Property (the "**Data Room**");
- (f) send to each Prospective Bidder upon request a form of Confidentiality Agreement and written acknowledgement of receipt of the SISP Procedures wherein such Prospective Bidder agrees to accept and be bound by the provisions of the SISP Procedures (the "Written Acknowledgement"). The Prospective Bidders will be required, among other things, to sign a Confidentiality Agreement in order to gain access to confidential information (including access to the Data Room). For greater certainty, only Prospective Bidders who submit an executed Confidentiality Agreement and Written Acknowledgement shall have access to the Data Room and other confidential information and management presentations, if available;
- (g) give access to the Data Room and coordinate the communication of information to each Prospective Bidder who has executed a Confidentiality Agreement and provided the Written Acknowledgement; and
- (h) prepare the form of a template asset purchase agreement (the "Template APA") to be used by Prospective Bidders to submit a Sale Proposal and post same in the Data Room.
- 4.2 The Monitor, in consultation with IMV and the Lenders' Agent, reserves the right to limit any Prospective Bidder's or Bidder's access to any confidential information (including any information in the Data Room) and to customers and suppliers of IMV, where, in the Monitor's sole discretion, such access could negatively impact the SISP, the ability to maintain the confidentiality of the confidential information, the Business or the value of the Property.
- 4.3 Any and all requests for additional information are to be made to the Monitor and IMV.

Section 5. Submission of Non-Binding Letters of Intent & Other Participation Requirements

- 5.1 Unless otherwise provided for herein, ordered by the Court or agreed to by the Monitor, in order to participate in the SISP and be considered for qualification as a Qualified Bidder, a Prospective Bidder must deliver to the Monitor, so as to be received by the Monitor no later than 5:00 p.m. (Halifax Time) on June 19, 2023, or such later date or time as the Monitor, in consultation with IMV and approved by the Lenders' Agent and in accordance with the SISP Procedures, may determine appropriate (the "LOI Deadline") the following:
 - (a) an executed Confidentiality Agreement, which shall inure to the benefit of any purchaser of any part of the Property or any investor in IMV;
 - (b) an executed Written Acknowledgement;
 - (c) a non-binding letter of intent (a "**LOI**") which specifies whether the Prospective Bidder anticipates submitting a Sale Proposal or an Investment Proposal, and which complies with the requirements of paragraph 5.2 and 5.3 or 5.4 below, as applicable; and

- (d) a letter setting forth the identity of the Prospective Bidder, the contact information for such Prospective Bidder, and the contact information for any business, financial or legal advisors retained or to be retained in connection with the contemplated transaction, and full disclosure of the direct and indirect owners of the Prospective Bidder and their principals, and a description of the Prospective Bidder's plans regarding the business of IMV.
- 5.2 An LOI in respect of a Sale Proposal or an Investment Proposal must include:
 - (a) an acknowledgment that the Sale Proposal will be made on an "as is, where is" basis;
 - (b) a detailed description of any remaining due diligence required by the Prospective Bidder to be completed before seeking to make a Qualified Bid and an estimated timeline for the completion of such due diligence (including with respect to any environmental due diligence);
 - (c) any anticipated regulatory and other approvals required to close the proposed transaction and the anticipated time frame and any anticipated impediments for obtaining any such approvals;
 - (d) all material conditions to closing that the Prospective Bidder may wish to impose, including any financing condition;
 - (e) confirmation that the Prospective Bidder will be responsible for its own costs incurred in connection with its investigation of IMV and any transaction, including those of its advisors, attorneys, and agents;
 - (f) the proposed target closing date and a timeline to closing with critical milestones;
 - (g) an indication as to whether the Prospective Bidder is intending to effect the Sale Proposal through a special purpose vehicle;
 - (h) any other terms and conditions which the Prospective Bidder believes are material to the transaction;
 - (i) that the LOI is governed by the laws of the Province of Nova Scotia and the laws of Canada applicable therein; and
 - (j) such other information reasonably requested by the Monitor.
- 5.3 In addition to the requirements set out in paragraph 5.2 hereof, an LOI in respect of a Sale Proposal must include:
 - (a) a detailed listing and description of the Property to be included in the Sale Proposal and a detailed listing of the Property to be excluded from the Sale Proposal;
 - (b) the low and high range of the proposed purchase price for such Sale Proposal, the proposed allocation of purchase price among the applicable Property and an explanation of what contingencies and variables may influence the range in which the final purchase price will fall;

- (c) details as to the form of consideration for the Sale Proposal;
- (d) a list of the key material contracts and leases, if any, the Prospective Bidder wishes to acquire and the Prospective Bidder's proposed treatment of any related "cure costs";
- (e) the structure and financing of the transaction, including a sources and uses analysis; and
- (f) a description of any liabilities to be assumed by the Prospective Bidder and the Prospective Bidder's estimated value of such assumed liabilities.
- 5.4 In addition to the requirements set out in paragraph 5.2 hereof, an LOI in respect of an Investment Proposal must include:
 - (g) a description of the structure of the Investment Proposal;
 - (h) a description of the type and amount of consideration, including equity, if any, to be allocated to secured creditors, unsecured creditors and shareholders of IMV;
 - (i) the proposed treatment of IMV's stakeholders;
 - (j) the structure and financing of the transaction, including a sources and uses analysis; and
 - (k) a description of any liabilities to be assumed by the Prospective Bidder and the Prospective Bidder's estimated value of such assumed liabilities.
- 5.5 For greater certainty, the Monitor shall be entitled, either prior to or following the LOI Deadline, to seek to clarify the terms of an LOI or with respect to any of the other requirements of paragraphs 5.1, 5.2 and 5.3 or 5.4 above, and the Monitor may accept a revised and/or clarified LOI, provided that the initial LOI was received prior to the LOI Deadline. The Monitor may grant extensions to the LOI Deadline with respect to the Business or Property.
- 5.6 The Monitor, after consulting IMV and with the approval of the Lenders' Agent, may waive compliance with any one or more of the requirements specified in paragraphs 5.1, 5.2 and 5.3 or 5.4 and deem any non-compliant LOI to be a qualifying LOI.

Section 6. Identification of Qualified Bidders

6.1 The Monitor and IMV, in consultation with the Lenders' Agent, shall review and consider each LOI and the other materials submitted by a Prospective Bidder pursuant to paragraph 5.1 and if (a) the LOI meets requirements specified in paragraphs 5.1, 5.2 and 5.3 or 5.4, (b) it is determined that it will be in the best interests of IMV and its stakeholders to permit the Prospective Bidder to continue to participate in the SISP based upon the terms set out in the applicable LOI and (c) such Prospective Bidder's financial information and credit support or enhancement demonstrate to the satisfaction of the Monitor the capability of such Prospective Bidder to consummate a transaction and that such Prospective Bidder is likely (based on availability of financing, experience and other considerations) to consummate either a Sale Proposal or an Investment Proposal, such Prospective Bidder

shall be a "**Qualified Bidder**". For greater certainty, an LOI may be in respect of only a part or parts of the Business or Property.

- 6.2 The determination by IMV and the Monitor as to whether a Prospective Bidder is a Qualified Bidder will be made as promptly as practicable in consultation with the Lenders' Agent after such Prospective Bidder has satisfied the requirements described in paragraph 5.1, 5.2 and 5.3 or 5.4, as applicable, (subject to any waiver thereof under paragraph 5.6), and any clarification that may be sought by the Monitor pursuant to paragraph 5.6 and in any event prior to **June 30, 2023** (the "**Qualification Deadline**"). If it is determined that a Prospective Bidder is a Qualified Bidder, the Monitor will promptly notify the Prospective Bidder that it is a Qualified Bidder.
- 6.3 If at any point before or after the LOI Deadline, IMV and the Monitor, in consultation with the Lenders' Agent, determines that there are or will be no Qualified Bidders with respect to the Business or Property, or that it will not be in the best interests of IMV to continue with the SISP with respect to all or any of the Business or Property, IMV as soon as reasonably practicable file a motion with the Court on notice to the Service List for advice and directions with respect to the modification, suspension or termination of the SISP in respect of the Business or Property.

Section 7. Due diligence

- 7.1 Each Qualified Bidder shall have such access to due diligence materials and information relating to the Business and Property, and the debt and equity interests in IMV, as the Monitor, in consultation with IMV, deems appropriate. At the request of a Qualified Bidder, such confidential due diligence information shall also be provided to a proposed lender of such Qualified Bidder that is reasonably acceptable to the Monitor, with the consent of IMV.
- 7.2 At the discretion of the Monitor, in consultation with IMV, due diligence access may include presentations (as may be scheduled by the Monitor), access to physical and secure online electronic data rooms, on-site inspections and such other matters as a Qualified Bidder may reasonably request and as to which the Monitor, in its reasonable business judgment deems appropriate. The Monitor shall not be obligated to furnish any due diligence materials or information after the Bid Deadline.
- 7.3 Unless otherwise agreed in writing by the Monitor, with the consent of IMV, no Prospective Bidder or Bidder shall be permitted to have any discussions with any counterparty to any contract with IMV or with any regulatory authority responsible for IMV or any other Prospective Bidder or Bidder in connection with any bid submitted in accordance with the terms hereof or in contemplation thereof.

Section 8. Submission of Qualified Bids

8.1 In order to continue to participate in the SISP, a Qualified Bidder must deliver a Qualified Purchase Bid or Qualified Investment Bid to the Monitor and such bids must be received by the Monitor by no later than **5:00 p.m. (Halifax Time)** on **July 10, 2023**, or at such later date or time as the Monitor, in consultation with IMV and approved by the Lenders' Agent and in accordance with the SISP Procedures, may determine appropriate (the "Bid Deadline").

- 8.2 A Sale Proposal submitted by a Qualified Bidder will be considered a "**Qualified Purchase Bid**" only if the Sale Proposal complies with all of the following:
 - (a) it includes a letter stating that the Sale Proposal is irrevocable until 45 Business Days following the Bid Deadline; provided, however, that if such Sale Proposal is selected as the Successful Bid or the Backup Bid, it shall remain irrevocable until the closing of the Successful Bid or the Backup Bid, as the case may be;
 - (b) it includes a duly authorized and executed purchase and sale agreement, together with a mark up outlining and highlighting all proposed changes from the Template APA, specifying the purchase price, expressed in Canadian dollars, including the cash component thereof and/or the liabilities to be assumed by the Bidder (or the combination of both - the "**Purchase Price**"), together with all exhibits and schedules thereto, and such ancillary agreements as may be required by the Qualified Bidder with all exhibits and schedules thereto (or term sheets that describe the material terms and provisions of such ancillary agreements) and such ancillary agreements;
 - (c) it contains a detailed listing and description of the Property to be included in the Sale Proposal or a detailed listing of the Property to be excluded from the Sale Proposal, as well as the value and breakdown of the allocation of the Purchase Price;
 - (d) it does not include any request or entitlement to any break-fee, expense reimbursement or similar type of payment. Further, by submitting a Sale Proposal, a Qualified Bidder shall be deemed to waive its right to pursue a substantial contribution claim in any way related to the submissions of its Sale Proposal or the SISP;
 - (e) it includes written evidence of a firm, irrevocable commitment for all required funding and/or financing from a creditworthy bank or financial institution to consummate the proposed transaction, or other evidence satisfactory to the Monitor in its sole discretion, to allow the Monitor to make a reasonable determination as to the Qualified Bidder's financial, technical, operational and other capabilities to consummate the transaction contemplated by the Sale Proposal and operate IMV 's operations;
 - (f) it is not conditioned on (i) the outcome of unperformed due diligence by the Qualified Bidder and/or (ii) obtaining any financing capital and includes an acknowledgement and representation that the Qualified Bidder has had an opportunity to conduct any and all required due diligence prior to making its Sale Proposal;
 - (g) it fully discloses the identity of each entity that is bidding or otherwise that will be sponsoring or participating in the Sale Proposal, including the identification of the Qualified Bidder's direct and indirect owners and their principals, and the complete terms of any such participation;
 - (h) it includes an acknowledgement and representation that the Qualified Bidder:
 (i) has relied solely upon its own independent review, investigation and/or inspection of any documents and/or the assets to be acquired and liabilities to be

assumed in making its Sale Proposal; (ii) did not rely upon any written or oral statements, representations, promises, warranties or guaranties whatsoever, whether express or implied (by operation of law or otherwise), regarding the assets to be acquired or liabilities to be assumed or the completeness of any information provided in connection therewith, including by IMV or the Monitor, except as expressly stated in the purchase and sale agreement submitted by it; (iii) is a sophisticated party capable of making its own assessments in respect of making its Sale Proposal; and (iv) has had the benefit of independent legal advice in connection with its Sale Proposal;

- (i) it includes a waiver of all claims against IMV, its officers, directors or employees, the Monitor, or its advisors in respect of any present, past and future activities of IMV, or any use or durability of IMV's assets, their quality, value, or sustainability;
- (j) it includes evidence, in form and substance reasonably satisfactory to the Monitor of authorization and approval from the Qualified Bidder's board of directors (or comparable governing body) with respect to the submission, execution, delivery and closing of the transaction contemplated by the Sale Proposal;
- (k) except in the case of a Credit Bid, it is accompanied by a refundable deposit (the "Deposit") in the form of a wire transfer (to a trust account specified by the Monitor), or such other form acceptable to the Monitor, payable to the order of *FTI Consulting Canada Inc.* in trust, in an amount equal to 10% of the proposed gross purchase price, to be held and dealt with in accordance with the SISP Procedures;
- (I) it contains full details of the proposed number of employees of IMV who will become employees of the Qualified Bidder and the proposed terms and conditions of employment to be offered to those employees;
- (m) it includes an acknowledgement and representation that the Qualified Bidder will assume the obligations of IMV under executory contracts, unexpired leases, and licences proposed to be assigned (or identifies clearly the particular contracts, leases, and licenses of IMV, as applicable, that the Qualified Bidder wishes not to assume, or alternatively wishes to assume), contains full details of the Qualified Bidder's proposal for the treatment of related cure costs; and which the assumption of which is a condition of closing;
- to the extent not addressed elsewhere, it includes the proposed treatment of stakeholders;
- it provides for closing of the Qualified Purchase Bid by no later than July 28, 2023, or such later date or time as the Monitor, in accordance with the SISP Procedures, may determine appropriate (the "Target Closing Date");
- (p) if the Qualified Bidder is an entity newly formed for the purpose of the transaction, the bid shall contain an equity or debt commitment letter from the parent entity or sponsor, which is satisfactory to the Monitor that names IMV as a third-party beneficiary of any such commitment letter with recourse by IMV and the Monitor against such parent entity or sponsor;

- (q) it includes evidence, in form and substance reasonably satisfactory to the Monitor of compliance or anticipated compliance with any and all applicable regulatory approvals (including, if applicable, anti-trust regulatory approval), the anticipated time frame for such compliance and any anticipated impediments for obtaining such approvals;
- (r) it contains other information reasonably requested by the Monitor and IMV;
- (s) it is governed by the laws of the Province of Nova Scotia and the laws of Canada applicable therein; and
- (t) it is received by no later than the Bid Deadline.
- 8.3 An Investment Proposal submitted by a Qualified Bidder will be considered a "**Qualified Investment Bid**" only if the Investment Proposal complies with all of the following:
 - (a) it includes a letter stating that the Investment Proposal is irrevocable for a period of 45 Business Days following the Bid Deadline; provided, however, that if such Investment Proposal is selected as the Successful Bid or the Backup Bid, it shall remain irrevocable until the earlier of (i) the closing of the Successful Bid or the Backup Bid, as the case may be, and (ii) the outside date stipulated in the Successful Bid or the Backup Bid, as applicable;
 - (b) it includes duly authorized and executed binding definitive documentation setting out the terms and conditions of the proposed transaction, including the aggregate amount of the proposed equity and/or debt investment and details regarding the proposed equity and/or debt structure of IMV, if applicable, following completion of the proposed transaction (a "Definitive Investment Agreement");
 - it includes a description of the type and amount of consideration, including equity, if any, to be allocated to secured creditors, unsecured creditors and shareholders of IMV;
 - (d) it does not include a request or entitlement to a break-fee, expense reimbursement or any other similar type of payment. Further, by submitting an Investment Proposal, the Qualified Bidder shall be deemed to waive its right to pursue a substantial contribution claim in any way related to the submission of its Investment Proposal or the SISP;
 - (e) it includes written evidence of a firm, irrevocable commitment for all required funding and/or financing from a creditworthy bank or financial institution to consummate the proposed transaction, or other evidence satisfactory to the Monitor in its sole discretion, to allow the Monitor to make a reasonable determination as to the Qualified Bidder's financial, technical, operational and other capabilities to consummate the transaction contemplated by the Investment Proposal and operate IMV's operations;
 - (f) it is not conditioned on (i) the outcome of unperformed due diligence by the Qualified Bidder and/or (ii) obtaining any financing capital and includes an acknowledgement and representation that the Qualified Bidder has had an

opportunity to conduct any and all required due diligence prior to making its Investment Proposal;

- (g) it fully discloses the identity of each entity that is bidding or that is sponsoring or participating in the Investment Proposal, including the identification of the Qualified Bidder's direct and indirect owners and their principals, and the complete terms of any such participation;
- (h) it includes an acknowledgement and representation that the Qualified Bidder: (i) has relied solely upon its own independent review, investigation and/or inspection of any documents in making its Investment Proposal; (ii) did not rely upon any written or oral statements, representations, promises, warranties or guaranties whatsoever, whether express or implied (by operation of law or otherwise), regarding the business of IMV or the completeness of any information provided in connection therewith, including by IMV and the Monitor, except as expressly stated in the Definitive Investment Agreement; (iii) is a sophisticated party capable of making its own assessments in respect of making its Investment Proposal; and (iv) has had the benefit of independent legal advice in connection with its Investment Proposal;
- (i) it includes a waiver of all claims against IMV, its officers, directors or employees and the Monitor, and its advisors in respect of any present, past and future activities of IMV, or any use or durability of IMV's assets, their quality, value, or sustainability;
- (j) it includes evidence, in form and substance reasonably satisfactory to the Monitor, of authorization and approval from the Qualified Bidder's board of directors (or comparable governing body) with respect to the submission, execution, delivery and closing of the transaction contemplated by the Investment Proposal;
- (k) it is accompanied by a Deposit in the form of a wire transfer (to a trust account specified by the Monitor), or such other form acceptable to the Monitor, payable to the order of *FTI Consulting Canada Inc.*, in trust, in an amount equal to 10% of the total proposed investment, to be held and dealt with in accordance with the SISP Procedures;
- (I) it provides for closing of the Qualified Investment Bid by no later than the Target Closing Date;
- (m) if the Qualified Bidder is an entity newly formed for the purpose of the transaction, the Investment Proposal shall contain an equity or debt commitment letter from the parent entity or sponsor, and satisfactory to the Monitor, that names IMV as a thirdparty beneficiary of any such commitment letter with recourse by IMV and the Monitor against such parent entity or sponsor;
- it includes evidence, in form and substance reasonably satisfactory to the Monitor, of compliance or anticipated compliance with any and all applicable regulatory approvals (including, if applicable, anti-trust regulatory approval), the anticipated time frame for such compliance and any anticipated impediments for obtaining such approvals;

- (o) it contains other information reasonably requested by the Monitor;
- (p) it is governed by the laws of the Province of Nova Scotia and the laws of Canada applicable therein; and
- (q) it is received by no later than the Bid Deadline.
- 8.4 For greater certainty, Sale Proposals and Investment Proposals may be in respect of only a part or parts of the Business or Property and such proposal shall constitute a "Qualified Portion Bid" if it satisfies the requirements in paragraph 8.2 or 8.3 hereof, as applicable, in respect of the Business or Property subject to such proposal, and in such case, such bidder shall constitute a "Qualified Portion Bidder". Each Qualified Portion Bid shall be deemed to be a Qualified Bid, and each Qualified Portion Bidder shall be deemed to be a Qualified Bid, and each Qualified Portion Bidder shall be deemed to be a Qualified Bid, and each Qualified Portion Bidder shall be deemed to be a Qualified Bid, and each Qualified Portion Bidder shall be deemed to be a Qualified Bidder, for all purposes of the SISP.

Section 9. Credit Bid

- 9.1 The Secured Lenders may use their secured debt as consideration for a Proposal in respect of any portion of the Business and/or Property subject to that Proposal.
- 9.2 In the event that no LOI is received that contemplates a purchase price which is sufficient to repay in cash all outstanding amounts owing to the Secured Lenders, the Se-cured Lenders shall be authorized to submit a Credit Bid under the SISP up to the value of the Secured Lenders' collateral and, in such case, and subject to a re-view of the validity and enforceability of the Secured Lenders' security. For the purpose of this Section 9, the Secured Lenders will be deemed Qualified Bidders and such Credit Bid shall be deemed a Qualified Bid. The Secured Lenders re-serve the right to submit a Credit Bid in the event that the SISP is terminated or otherwise fails to result in a Qualified Bid.
- 9.3 In the event that a Secured Lender, or any of their affiliates, notifies the Monitor and IMV, in writing, of their intention to make a Credit Bid, the Lenders' Agent shall not benefit from the consultation and approval rights set out hereunder.

Section 10. Qualified Bid

- 10.1 Qualified Purchase Bids and Qualified Investment Bids shall hereinafter be referred to as "Qualified Bids" and each a "Qualified Bid" and each bidder who has submitted a Qualified Bid shall hereinafter be referred to as a "Authorized Bidder".
- 10.2 Notwithstanding paragraph 8.2 and paragraph 8.3 the Monitor, in consultation with IMV and the approval of the Lenders' Agent, may waive compliance with any one or more of the Qualified Bid requirements specified herein, and deem such non-compliant bids to be Qualified Purchase Bids or Qualified Investment Bids, as the case may be.

Section 11. Assessment of Qualified Bids

11.1 The Monitor and IMV, in consultation with the Lenders' Agent, will review and assess the Qualified Bids in respect of a Sale Proposal, and in making such assessment will consider, among other things, the following (the "Sale Proposal Bid Criteria"):

- the purchase price and net value (including all assumed liabilities and other obligations to be performed by the Qualified Bidder) provided by such Qualified Bid and the proposed allocation of the purchase price among the applicable Property;
- (b) the firm, irrevocable commitment for financing the transaction or other evidence of ability to consummate the Sale Proposal;
- (c) the claims, if any, likely to be created against IMV by the transaction contemplated by the Sale Proposal, relative to alternatives available to IMV ;
- (d) the nature and amount of debt and other liabilities to be assumed or acquired by the Qualified Bidder;
- (e) the counterparties to the Sale Proposal;
- (f) the proposed revisions to the Template APA and the terms of the proposed sale transaction documents;
- (g) the assets included in or excluded from the Sale Proposal and the transaction costs and risks associated with closing multiple transactions versus a single sale transaction for all or substantially all of the Property;
- (h) any transition services required from IMV post-closing and any related restructuring costs;
- (i) the planned treatment of stakeholders; and
- (j) other factors affecting the speed, certainty and value of the Sale Proposal (including any regulatory approvals and other conditions required to close the Sale Proposal by the applicable Target Closing Date), including the likelihood of closing the Sale Proposal on or before the applicable Target Closing Date.
- 11.2 The Monitor and IMV, in consultation with the Lenders' Agent, will review and assess the Qualified Bids in respect of an Investment Proposal, and in making such assessment will consider, among other things, the following (the "Investment Proposal Bid Criteria"):
 - the type and amount of consideration, including equity, if any, to be allocated to secured creditors, unsecured creditors and shareholders of IMV and the planned treatment of such persons under the proposed Investment Proposal;
 - (b) the firm, irrevocable commitment for financing the investment or other evidence of ability to consummate the Investment Proposal;
 - (c) the counterparties to the proposed Investment Proposal;
 - (d) the cost, risks and timing associated with obtaining the approval of the requisite majority of creditors and approval of the Court in respect of a Plan, if needed;

- the estimated number of employees of IMV that will be offered post-closing employment by the Bidder and any proposed measures associated with their continued employment;
- (f) the transition services required from IMV post-closing and any related costs;
- (g) the planned treatment of stakeholders; and
- (h) other factors affecting the speed, certainty and value of the Investment Proposal (including any regulatory approval and other conditions required to close the Investment Proposal by the applicable Target Closing Date), including the likelihood of closing the Investment Proposal on or before the applicable Target Closing Date.
- 11.3 For greater certainty, the Monitor and IMV, in consultation with the Lenders' Agent, shall be entitled, either prior to or following the Bid Deadline, to seek to clarify the terms of Qualified Bid and the Monitor may accept a revised and/or clarified Qualified Bid provided that the initial Qualified Bid was received prior to the Bid Deadline.
- 11.4 The Monitor and IMV, with the approval of the Lenders' Agent, may waive compliance with any one or more of the requirements specified in paragraph 8.2 and paragraph 8.3, as applicable and deem any non-compliant Bid to be a Qualified Bid.
- 11.5 The Monitor and IMV shall apply the Sale Proposal Bid Criteria and Investment Proposal Bid Criteria, as applicable, and consider each Qualified Bid upon its submission for determination. Such determination will be made as promptly as practicable after the Bid Deadline.
- 11.6 If IMV, in consultation with the Monitor and with the approval of the Lenders' Agent, determines that any Qualified Bid was received that is in the best interests of IMV's stakeholders (or any combination of non-overlapping Qualified Portion Bids was received that is in the best interests of IMV's stakeholders), IMV, in consultation with the Monitor and the Lenders' Agent, may choose to accept such Qualified Bid (in which case, such Qualified Bid shall be a "**Successful Bid**" and the Qualified Bidder making the Successful Bid shall be a "**Successful Bidder**") and take such steps as are necessary to finalize and complete an agreement for the Successful Bid with the Successful Bidder. For greater certainty, IMV, in consultation with the Monitor and the Lenders' Agent, may accept a combination of non-overlapping Qualified Portion Bids (collectively, an "**Aggregated Bid**") to create one Successful Bid and in such case, the applicable Authorized Bidders will become "**Successful Bidders**".
- 11.7 If IMV, in consultation with the Monitor and the Lenders' Agent, determine that more than one Qualified Bid (and/or more than one Aggregated Bid) should be considered, then the Monitor <u>may</u>, <u>without being obligated to do so</u>, conduct an auction (the "**Auction**"), to determine the highest and/or best Sale Proposal or Investment Proposal or Aggregated Bid. In the event that an Auction is to be held, all Authorized Bidders who submitted a Qualified Bid that IMV, in consultation with the Monitor determines entitles such Authorized Bidder to participate in the Auction (each, an "**Auction Bidder**") will be advised by the Monitor of such determination.

- 11.8 An Authorized Bidder not identified as an Auction Bidder will no longer be able to participate in the SISP or any Auction. In the event an Authorized Bidder is selected as an Auction Bidder, such Auction Bidder's Qualified Bid shall remain binding, irrevocable and open for acceptance until the acceptance of a Successful Bid in accordance with section 12.2(j) and, where applicable, the selection of the Backup Bid further to the Auction. For greater certainty, if such Auction Bidder's Bid is accepted as the Successful Bid or selected as the Backup Bid it shall remain binding and irrevocable and dealt with as such in accordance with the SISP Procedures.
- 11.9 The Monitor and IMV, in consultation and with the approval of the Lenders' Agent, may at any time (including prior to or during an Auction), (a) reject any Bid that is (i) inadequate or insufficient, (ii) not in conformity with the requirements of the CCAA, the SISP Procedures or any orders of the Court applicable to IMV; (b) in accordance with the terms hereof, accept Bids not in conformity with the SISP Procedures that is more favourable; (c) in accordance with the terms hereof, extend the Bid Deadline, and/or change the Auction Date; and/or (d) reject all Bids. For greater certainty, IMV shall be under no obligation to accept the highest or best offer and the selection of the Successful Bid shall be entirely in the discretion of IMV, in consultation with the Monitor and the Lenders' Agent.

Section 12. Auction

- 12.1 If the Auction is to be conducted pursuant to paragraph 11.7, the Auction shall commence on a date as the Monitor, may determine is appropriate (the "Auction Date"). All Auctions shall be conducted virtually through a platform to be determined by the Monitor, at the offices of *FTI Consulting Canada Inc.* or such other location as the Monitor may determine. Notice of the platform or place, date and time of the Auction will be delivered to all Auction Bidders by the Monitor not less than three (3) Business Days before the date of the Auction.
- 12.2 If there is an Auction, the Auction shall be conducted according to the following procedures:
 - (a) At least one (1) Business Day prior to the Auction, each Auction Bidder who has been notified that it has qualified as an Auction Bidder must inform the Monitor whether it intends to attend and participate in the Auction provided that, for greater certainty, such Auction Bidder's Qualified Bid shall remain binding, irrevocable and open for acceptance until the acceptance of a Successful Bid in accordance with section 12.2(j) and, where applicable, the selection of the Backup Bid further to the Auction.
 - (b) The Monitor shall direct and preside over the Auction. Only Auction Bidders are eligible to participate in the Auction. Only the authorized representatives (including legal counsel and other advisors) of each of the Auction Bidders, IMV, the Lenders' Agent and the Monitor shall be permitted to attend the Auction.
 - (c) Each Auction Bidder shall be required to confirm that it has not engaged, and will not engage, in any collusion with respect to the bidding or any Sale Proposal or Investment Proposal, and if such Auction Bidder is a special purpose vehicle, each of the direct or indirect equity holders of such Auction Bidder shall be required to confirm that it has not engaged, and will not engage, in any collusion with respect

to the bidding or any Sale Proposal or Investment Proposal, such confirmation, in each case, in form and substance satisfactory to the Monitor in its sole discretion.

- (d) Bidding at an Auction shall be conducted in rounds. In each round, an Auction Bidder may submit no more than one Overbid. If at the end of any round of bidding, an Auction Bidder (other than the Auction Bidder who submitted the "**Opening Bid**" for such round and any Qualified Portion Bidder who is an Auction Bidder) did not submit an Overbid, then such Auction Bidder shall be barred from participating in any further round of bidding at the Auction. Any Auction Bidder who submits an Overbid in a round, as well as the Auction Bidder who submitted the Opening Bid for such round and any Qualified Portion Bidder who is an Auction Bidder, shall be entitled to participate in the next round of bidding at the Auction.
- (e) IMV, in consultation with the Monitor and the Lenders' Agent, shall apply the Sale Proposal Bid Criteria and Investment Proposal Bid Criteria, as applicable, to determine which Qualified Bid is the highest and/or best bid received by the Bid Deadline, which shall constitute the Opening Bid for the first round of an Auction. IMV shall follow the same process to determine the highest and/or best Overbid submitted in each round of an Auction, which shall constitute the Opening Bid for the following round. For greater certainty, an Aggregated Bid may be determined to be the "Opening Bid" for any round. As soon as practicable prior to the start of the Auction, the Monitor shall distribute a copy of the Opening Bid for the first round to all Auction Bidders eligible to participate in the applicable Auction.
- (f) All bids made at an Auction shall be Overbids and shall be made and received on an open, non-confidential basis and the identity of each Auction Bidder and all material terms of each Overbid shall be fully disclosed to all other Auction Bidders participating in the applicable round of the applicable Auction. The Monitor shall maintain a transcript of the Opening Bids and all Overbids made and announced at an Auction.
- (g) A Sale Proposal or Investment Proposal submitted at an Auction will be considered an "**Overbid**" only if it complies with the following requirements:
 - (i) Minimum Consideration. Subject to subparagraph (I) below in respect of Qualified Portion Bids, the amount of the purchase price (in the case of a Sale Proposal), or the amount of the consideration to be allocated to secured creditors, unsecured creditors and shareholders of IMV (in the case of an Investment Proposal) shall not be less than the purchase price or consideration of the Opening Bid of the applicable round of such Auction, plus an amount (the "Minimum Overbid Increment") to be set by the Monitor; and
 - (ii) Qualified Bid Criteria. Except as modified herein, an Overbid shall comply with all requirements for a Qualified Bid as set forth in paragraph 8.2 in the case of Sale Proposals, or paragraph 8.3 in the case of Investment Proposals, (in each case including in respect of its binding and irrevocable nature, and being open for acceptance until the acceptance of a Successful Bid in accordance with section 12.2(j) and, where applicable, the selection of the Backup Bid further to the Auction) provided, however, that the Bid

Deadline shall not apply and Overbids need not be accompanied by additional cash deposits during the Auction.

- (h) At the end of each round of bidding, the Monitor shall (i) review each Overbid made in such round, (ii) identify the highest and/or best such Overbid in accordance with subparagraph (e), and (iii) announce to all Auction Bidders entitled to participate in the next round of bidding the terms of the highest and/or best Overbid and the identity of the Auction Bidder who submitted such Overbid. Such highest and/or best Overbid shall be the Opening Bid for the next round of such Auction.
- (i) The Monitor, in consultation with IMV and the Lenders' Agent, reserves the right to make one or more adjournments in an Auction to, among other things: (i) allow individual Auction Bidders to consider how they wish to proceed; (ii) consider and determine the current highest and/or best Overbid at any given time during the Auction; and (iii) give Auction Bidders the opportunity to provide the Monitor with such additional evidence as it may require to show that the Auction Bidder's bid complies with the requirements of an Overbid (including in respect of the required internal corporate or credit committee approvals and evidence of sufficient funding commitments or other financial capability to consummate the proposed transaction).
- (j) If, in any round of bidding, no new Overbid is made, such Auction shall be closed and the Monitor shall declare the last Opening Bid as a "Successful Bid" and the Auction Bidder submitting such Successful Bid a "Successful Bidder" and advise such Successful Bidder of such determination and all other applicable Auction Bidders that they are not a Successful Bidder.
- (k) To the extent not already provided, the Successful Bidder shall, within two (2) Business Days of the conclusion of the Auction, provide the Monitor with an additional Deposit to increase its original Deposit to equal ten percent (**10%**) of the total cash purchase price or investment contemplated by the Successful Bid.
- (I) Each Qualified Portion Bidder that is an Auction Bidder shall be entitled to submit Overbids at the applicable Auction (in a minimum increment to be determined by the Monitor) with respect to the portion of the Business or Property it is bidding on, and is not individually subject to the full Minimum Overbid Increment; provided that one or more Qualified Portion Bids forming an Aggregated Bid in any round of the Auction shall collectively be subject to the full Minimum Overbid Increment. For greater certainty, the Monitor may accept an Aggregated Bid as a "Successful Bid" and in such case, the applicable Auction Bidders will become "Successful Bidders".
- (m) For greater certainty, the Monitor and IMV shall be entitled during an Auction, to discuss and clarify the terms of all Overbids and accept a revised, clarified Overbid, provided it is submitted before the end of the applicable round of bidding. The Monitor and IMV, in consultation with the Lenders' Agent, may waive compliance with any one or more of the requirements specified in subparagraph (g), and deem any non-compliant Overbid to be a qualifying Overbid.
- (n) The Monitor may adopt rules for the Auction at or prior to the Auction that will better promote the goals of the Auction and that are not inconsistent with any of the provisions of the SISP Procedures or the SISP Approval Order; provided that no

such rules may change the requirement that all Overbids shall be made and received on an open, non-confidential basis, and all Auction Bidders entitled to participate in a further round of bidding shall be entitled to be present for all such bidding.

Section 13. Backup Bid

13.1 In the event a Successful Bid is accepted in accordance with section 11.6 or further to an Auction in accordance with section 12.2(j), IMV, in consultation with the Monitor and the Lenders' Agent, may also select any Qualified Bid, Aggregated Bid or Overbid, as the case may be, as the "**Backup Bid**" and take such steps as are necessary to finalize and complete an agreement for the Backup Bid with the Backup Bidder. In the event the closing of the Successful Bid accepted in accordance with sections 11.6 or 12.2(j), as the case may be, does not occur, the Backup Bid shall, upon confirmation of the Monitor, become the Successful Bid and be dealt with as such in accordance with the SISP Procedures.

Section 14. Approval Motion

- 14.1 After a definitive agreement(s) in respect of a Successful Bid has been finalized in accordance with the SISP Procedures, if such Successful Bid relates to the Business or Property, IMV shall apply to the Court as soon as reasonably practicable for an order approving such Successful Bid and authorizing IMV to enter into any and all necessary agreements with respect to such Successful Bid and to undertake such other actions as may be necessary or appropriate to give effect to such Successful Bid, including for the approval of any Plan(s) pursuant to the CCAA or Arrangement pursuant to the CBCA, as applicable (an "Approval Motion").
- 14.2 An Approval Motion will be held on a date to be scheduled by the Court and to be heard as soon as possible. IMV reserves its right, as may be necessary or appropriate, to seek to proceed on an expedited basis and abridge any notice period provided for in any Order. An Approval Motion may be adjourned or rescheduled by IMV by an announcement of the adjourned date at an Approval Motion or by notice to the Service List and no further notice shall be required.
- 14.3 All Bids (other than the Successful Bid and the Backup Bid, as the case may be) will be deemed rejected at 11:59 p.m. (Halifax Time) on the Business Day after the acceptance of the Successful Bid relating to the same Business and/or Property.
- 14.4 For the avoidance of doubt, the approvals required pursuant to the terms hereof are in addition to, and not in substitution for, any other approvals required by the CCAA or any other statute or as otherwise required at law in order to implement a Successful Bid.

Section 15. Treatment of Deposit

- 15.1 If there is a Successful Bid, the Deposit (plus accrued interest) paid by a Successful Bidder whose bid is approved by the Court will be released by the Monitor and applied to the purchase price to be paid, or investment to be made, by such Successful Bidder upon closing of the approved transaction or as otherwise set out in the definitive agreement.
- 15.2 The Deposits of Bidders not selected as a Successful Bidder, will be returned to such Bidders within ten (10) Business Days of the date of closing of the Successful Bid. If there

is no Successful Bid with respect to the Business or the Property, subject to the following paragraph 15.3, all Deposits with respect to such Business or Property will be returned to all Bidders with respect to that Business or Property, within ten (10) Business Days of the date on which the SISP with respect to that Business or Property is terminated in accordance with the SISP.

15.3 If (i) a Successful Bidder breaches any of its obligations under the terms of the SISP Procedures or any definitive transaction documentation; (ii) a Bidder fails to complete the transaction contemplated by its Bid if required by the Monitor to complete such transaction; or (iii) a Bidder fails to provide proof of its ability to complete the transaction to the Monitor (other than with respect to conditions specifically provided in its Bid), within five (5) Business Days of a request to that effect from the Monitor, then, in each case, such Bidder's Deposit will be forfeited to IMV as liquidated damages and not as a penalty. IMV shall apply and use any forfeited Deposit in a manner agreed upon by the Monitor, provided, however, that the forfeiture of such Deposit shall be in addition to, and not in lieu of, any other rights in law or equity that IMV and the Monitor have or may have against such breaching entity.

Section 16. Reservation of Rights and Conduct of the SISP

- 16.1 The SISP does not and will not be interpreted to create any contractual or other legal relationship between IMV or the Monitor and any Prospective Bidder and Bidder, other than as specifically set forth in a definitive agreement that any such Bidder may enter into with the Monitor.
- 16.2 The Monitor and IMV, in consultation and with the approval of the Lenders' Agent, may (a) reject, at any time any Bid that is (i) inadequate or insufficient, (ii) not in conformity with the requirements of the SISP Procedures, or (iii) contrary to the best interests of IMV, its estate, and stakeholders as determined by IMV and the Monitor; (b) in accordance with the terms hereof accept bids not in conformity with the SISP Procedures to the extent that IMV and the Monitor determine, in their reasonable business judgment, that doing so would benefit IMV, its estate, and stakeholders; and (c) reject all Bids. IMV shall not be required to accept the highest Bid.
- 16.3 The Monitor, in its reasonable discretion and in consultation with the Lenders' Agent, may shorten the dates provided for herein or terminate the SISP if there are no credible opportunities for the conclusion of a Transaction in the process.

Section 17. Notice to IMV and the Monitor

17.1 Any notice or other communication to be given to IMV in connection with this SISP shall be given in writing and shall be given by personal delivery (in which case it shall be left with a responsible officer of the recipient) or by electronic communication addressed to IMV as follows:

McCarthy Tétrault LLP 1000 De La Gauchetière Street West, MZ400 Montréal, Québec, H3B 0A2

Attention: Alain N. Tardif François Alexandre Toupin

atardif@mccarthy.ca fatoupin@mccarthy.ca

17.2 Any notice or other communication to be given to the Monitor in connection with this SISP shall be given in writing and shall be given by personal delivery (in which case it shall be left with a responsible officer of the recipient) or by electronic communication addressed to the Monitor as follows:

FTI Consulting Inc.

TD South Tower, 79 Wellington Street W Toronto-Dominion Centre, Suite 2010, P.O. Box 104 Toronto, ON M5K 1G8 Canada

Attention: Jeffrey Rosenberg Jodi Porepa

jeffrey.rosenberg@fticonsulting.com jodi.porepa@fticonsulting.com

With a copy to :

Stikeman Elliott LLP

5300 Commerce Court West 199 Bay Street Toronto, Ontario M5L 1B9 Canada

Attention: Maria Konyukhova

mkonyukhova@stikeman.com

SCHEDULE A DEFINED TERMS

"Aggregated Bid" has the meaning ascribed to it in paragraph 11.6.

"Approval Motion" has the meaning ascribed to it in paragraph 14.1.

"Arrangement" has the meaning ascribed to it in Recital E.

"IMV" has the meaning described thereto in Recital A.

"Auction" has the meaning ascribed to it in paragraph 11.7.

"Auction Bidder" has the meaning ascribed to it in paragraph 11.7.

"Auction Date" has the meaning ascribed to it in paragraph 12.1.

"Authorized Bidder" means each bidder who has submitted a Qualified Bid.

"Backup Bid" has the meaning ascribed to it in paragraph 13.1.

"Backup Bidder" means any Bidder whose Bid is selected as the Backup Bid.

"Bid Deadline" has the meaning ascribed to it in paragraph 8.1.

"**Bidders**" means collectively Qualified Bidders, Authorized Bidders, Auction Bidders, Backup Bidder and Successful Bidder, each a "**Bidder**".

"Bids" mean collectively Qualified Bids, Aggregated Bids, Backup Bids and Successful Bids, each a "Bid".

"Business" has the meaning ascribed to it in Recital D.

"**Business Day**" means any day other than (i) a Saturday or a Sunday or (ii) a day which is a statutory holiday in Halifax, Nova Scotia.

"CBCA" has the meaning ascribed to it in Recital E.

"CCAA" has the meaning ascribed to it in Recital A.

"CCAA Proceedings" has the meaning ascribed to it in Recital B.

"Commencement Date" has the meaning ascribed to it in paragraph 4.1.

"**Confidentiality Agreement**" means a form of confidentiality agreement satisfactory to the Monitor.

"**Contact List**" has the meaning ascribed to it in subparagraph 4.1(a).

"Court" has the meaning ascribed to it in Recital A.

"Credit Bid" means a Bid that meets the conditions set out in paragraph9.1.

"Data Room" has the meaning ascribed to it in subparagraph 4.1(e).

"Definitive Investment Agreement" has the meaning ascribed to it in subparagraph 8.3(b).

"Deposit" has the meaning ascribed to it in subparagraph 8.2(k).

"Initial Order" has the meaning ascribed to it in Recital A.

"**Investment Proposal**" means an offer for a broad range of executable transaction alternatives (restructuring, recapitalization and/or refinancing) involving an investment in IMV.

"Investment Proposal Bid Criteria" has the meaning ascribed to it in paragraph 11.2.

"Lenders' Agent" means Horizon Technology Finance Corporation, in its capacity as collateral agent for the Secured Lenders under a Venture Loan and Security Agreement dated as of December 17, 2021.

"LOI" has the meaning ascribed to it in paragraph 5.1(b).

"LOI Deadline" has the meaning ascribed to it in paragraph 5.1.

"Minimum Overbid Increments" has the meaning ascribed to it in subparagraph 12.2(g)(i).

"Monitor" has the meaning ascribed to it in Recital B.

"Opening Bid" has the meaning ascribed to it in subparagraph 12.2(d).

"Overbid" has the meaning ascribed to it in subparagraph 12.2(g).

"Plan" has the meaning described thereto in Recital E.

"Property" has the meaning ascribed to it in Recital D.

"Proposal" means any Sale Proposal or Investment Proposal.

"Prospective Bidders" has the meaning ascribed to it in subparagraph 4.1(a).

"Purchase Price" has the meaning ascribed to it in subparagraph 8.2(b).

"Qualification Deadline" has the meaning ascribed to it in paragraph 6.2.

"**Qualified Bids**" means Qualified Purchase Bids and Qualified Investment Bids, each a Qualified Bid. A Qualified Portion Bid shall be deemed a Qualified Bid.

"Qualified Bidder" has the meaning ascribed to it in paragraph 6.1.

"Qualified Investment Bid" has the meaning ascribed to it in paragraph 8.3.

"Qualified Portion Bid" has the meaning ascribed to it in paragraph 8.4.

"Qualified Portion Bidder" has the meaning ascribed to it in paragraph 8.4.

"Qualified Purchase Bid" has the meaning ascribed to it in paragraph 8.2.

"Sale Proposal" means an offer to acquire all or part of the Property.

"Sale Proposal Bid Criteria" has the meaning ascribed to it in paragraph 11.1

"Secured Lenders" means Horizon Technology Finance Corporation, Horizon Credit II LLC, Horizon Funding I, LLC, Powerscourt Investments XXV LP, and Powerscourt Investments XXV Trust.

"Service List" has the meaning ascribed to it in paragraph 1.7.

"SISP" has the meaning ascribed to it in Recital C.

"SISP Approval Order" has the meaning ascribed to it in Recital C.

"SISP Procedures" has the meaning ascribed to it in Recital C.

"Successful Bid" has the meaning ascribed to it in paragraph 11.6.

"Successful Bidder" means the Qualified Bidder making the Successful Bid.

"Target Closing Date" has the meaning ascribed to it in subparagraph 8.2(o).

"Teaser Letter" means a solicitation letter summarizing the acquisition and investment opportunity with respect to the Property and the Business.

"Written Acknowledgement" has the meaning ascribed to it in subparagraph 4.1(f).

"Template APA" has the meaning ascribed to it in subparagraph 4.1(h).

Schedule "C" Draft Claims Process Order

SUPREME COURT OF NOVA SCOTIA

IN THE MATTER OF: Application by IMV Inc. and Immunovaccine Technologies Inc. and IMV USA Inc. (the "Applicants"), for relief under the *Companies' Creditors Arrangement Act*

Claims Process Order

Before the Honourable

in chambers:

The Applicants propose to make a compromise or arrangement under the *Companies' Creditors Arrangement Act*, R.S.C. 1985, c. C-36 as amended (the "**CCAA**") and they applied for an initial order and, now or in the future, other relief under the CCAA as may be sought on notice of motion.

The following parties received notice of this application: see attached at Schedule "A".

The following parties, represented by the following counsel, made submissions:

Counsel

Party 199

Applicants

McCarthy Tétrault LLP Alain N. Tardif François Alexandre Toupin

Stewart McKelvey Lawyers Sara L. Scott

Monitor, FTI Consulting Canada Inc.

Maria Konyukhova

Stikeman Elliott LLP

Horizon Technology Finance Corporation, as agent Aird & Berlis LLP Miranda Spence Kyle Plunkett

On motion of the Applicants, the following is ordered and declared:

Service

1. The service of the Notice of Motion, the associated pleadings, and the supporting documents, as set out in the affidavit of service is hereby abridged and validated so that this Motion is properly returnable today and further service thereof is hereby dispensed with.

2. The service of the present Order (this "**Order**") at any time and place and by any means whatsoever, including email.

Definitions

- 3. The following terms in this Order shall, unless otherwise indicated, have the following meanings ascribed thereto:
 - 3.1 **"Appeal Application**" means a written application referred to in paragraph 9 of this Order setting out in detail the grounds of appeal from a Notice of Revision or Disallowance;
 - 3.2 **"Assessment**" means any right or claim of Her Majesty the Queen in Right of Canada or of any Province or Territory or Municipality or any other taxation authority in any Canadian or foreign jurisdiction against any of the Applicants, including, without limitation, amounts which may arise, have arisen under, or would arise under, in connection with, or as a result of any notice of assessment, notice of objection, notice of reassessment, notice of appeal, audit, investigation, demand or similar request from any taxation authority, without regard to whether the statute of limitation with respect to any particular tax Claim has expired or whether any audit or investigation has already been conducted;
 - 3.3 "BIA" means the Bankruptcy and Insolvency Act, RSC 1985, c B-3;
 - 3.4 **"Business Day**" means a day, other than a Saturday, Sunday or a statutory holiday in the Province of Nova Scotia;
 - 3.5 **"CCAA**" means the *Companies' Creditors Arrangement Act*, RSC 1985, c C-36;
 - 3.6 **"CCAA Proceedings**" means the proceedings in respect of the Applicants before the Court commenced pursuant to the CCAA in Court file Hfx No. 523334;
 - 3.7 "Claim" means any right of any Person against any of the Applicants, arising in or in connection with any jurisdiction including but not limited to the United States of America and Canada, in connection with any indebtedness, right to payment, or obligation of any kind of the Applicants, whether or not such right is reduced to judgment, present, future, due or accruing due to such Person and any corresponding interest accrued thereon or costs, damages or equivalents, payable in respect thereof, whether liquidated, unliquidated, fixed, contingent, matured, unmatured, disputed, undisputed, secured, unsecured, known or unknown, including, inter alia, any Assessment, any right of ownership of or title to property or assets or right to a trust or deemed trust (statutory, express, implied, resulting, constructive or otherwise), any right to an equitable remedy for breach of performance if such breach gives rise to a right to payment, whether or not such right to an equitable remedy is reduced to judgment, fixed, contingent, matured, unmatured, disputed, undisputed, secured, or unsecured, any executory or non-executory guarantee or surety,

and (i) the right or ability of any Person to advance a claim for contribution, indemnity or otherwise with respect to any matter, action or cause, which indebtedness, liability or obligation is based in whole or in part on facts existing as at the Determination Date; (ii) any Equity Claim; and (iii) any claim which would constitute a claim under the CCAA as at the Determination Date. A Claim shall include, without limitation, (a) any Unaffected Claim; (b) any Claim against the Beneficiaries; (c) any Claim against the Directors and Officers; and (d) any Restructuring Claim, provided however, that in no case shall a Claim include an Excluded Claim;

3.8 **"Claims Bar Date**" means:

- (i) for Employees, 5:00 p.m. (Halifax Time) on June 30, 2023;
- (ii) for a Creditor with a Restructuring Claim, the latest of (a) 5:00 p.m. (Halifax Time) on July 31, 2023 and (b) thirty (30) days after the date of receipt by the Creditor of a notice from the Applicants giving rise to the Restructuring Claim, it being understood that at no time shall such a notice from the Applicants be sent to the Creditor less than thirty (30) days before the date of the Creditors' Meeting
- (iii) for a Creditor with a Claim, 5:00 p.m. (Halifax Time) on July 31, 2023; and
- (iv) for a Creditor with a Claim against the Directors and Officers, 5:00 p.m. (Halifax Time) on August 29, 2023;
- "Claim against the Directors and Officers" means a claim as defined in 3.9 paragraph 11.03(1) of the CCAA as well as any right of any Person against any of the Directors and Officers whether asserted against them in their personal capacity or in their capacity as directors, officers, shareholders, trustees, beneficiaries of any of the Applicants or of any entities related to, or affiliated with, the Applicants or the Directors and Officers, in connection with any indebtedness or obligation of any kind of the Directors and Officers, present. future, due or accruing due to such Person and any interest accrued thereon or costs payable in respect thereof, whether liquidated, unliquidated, contingent, matured, unmatured, disputed, undisputed, secured, unsecured, known or unknown, including, inter alia, any executory or non-executory guarantee or surety and (i) the right or ability of any Person to advance a claim for contribution, indemnity or otherwise with respect to any matter, action or cause, which indebtedness, liability or obligation is based in whole or in part on facts existing as at the Determination Date; (ii) any Equity Claim; (iii) any Restructuring Claim and (iv) any claim which would constitute a claim under the CCAA as at the Determination Date;
- 3.10 "Court" means the Nova Scotia Supreme Court;
- 3.11 "Creditor" means any Person having a Claim and may, where the context requires, include the assignee of a Claim or a trustee, interim receiver, receiver, receiver and manager, or other Person acting on behalf of such

Person and includes a Known Creditor. A Creditor shall not, however, include an Excluded Creditor in respect of that Person's claim resulting from an Excluded Claim;

- 3.12 **"Creditors' Instructions**" means (i) for Creditors other than Employees, the instructions for Creditors, including a Proof of Claim, an Instruction Letter explaining how to complete same, and a copy of this Order and (ii) for Employees, the Employee Claim Statement, which includes the instructions to Employees;
- 3.13 **"Creditors' List**" means a list of all Known Creditors;
- 3.14 **"Creditors' Meeting**" means any meeting of the Applicants' Creditors to be convened, with leave of the Court, for the purposes of voting on the Plan, and any adjournment or suspension thereof;
- 3.15 **"Directors and Officers**" means any of the present or former, de jure or de facto, directors or officers of any of the Applicants as well as any other individuals legally entitled to administer the affairs of any of the Applicants;
- 3.16 **"Designated Newspapers**" means La Presse+, The Globe and Mail (National Edition) and the Wall Street Journal (National Edition);
- 3.17 **"Determination Date**" means May 1st, 2023;
- 3.18 "**Employee(s)**" means a Creditor who is an employee or a former employee the Applicants;
- 3.19 **"Employee Claim Statement**" means the employee instruction letter and claim statement, being substantially in the form of **Schedule "B"** hereto.
- 3.20 **"Equity Claim**" has the meaning ascribed thereto in the definition contained in the BIA and the CCAA;
- 3.21 "Excluded Claim" means (a) any right of any Person against the Applicants in connection with any indebtedness or obligation of any kind which came into existence after the Determination Date and any interest thereon, including any obligation of the Applicants toward creditors who have supplied or shall supply services, utilities, goods or materials or who have or shall have advanced funds to the Applicants after the Determination Date, but only to the extent of their claims in respect of the supply of such services, utilities, goods, materials or funds after the Determination Date and to the extent that such claims are not otherwise affected by the Plan; and (b) any right or claim that may be asserted by any beneficiary of any charges granted by the Court under the CCAA in the present proceedings, with respect to such charges;
- 3.22 "Excluded Creditor" means a Person having a Claim in respect of an Excluded Claim but only in respect of such Excluded Claim and to the extent that the Plan does not otherwise affect such Claim;

- 3.23 "Initial Order" means the order of the Court rendered on May 1st, 2023 under the CCAA, as amended and restated from time to time, including on May 5, 2023;
- 3.24 **"Instruction Letter**" means the instruction letter sent to Creditors, other than Employees, in a document substantially in the form of **Schedule "C"** hereto;
- 3.25 **"Known Creditor**" means a Creditor whose Claim is included in the Applicants' books and records;
- 3.26 **"Monitor**" means FTI Consulting Canada Inc., acting in its capacity as Monitor of the Applicants pursuant to the Initial Order;
- 3.27 **"Newspaper Notice**" means the notice of this Order to be published in the Designated Newspapers on the Publication Date in accordance with paragraph 4, which shall set out the Claims Bar Date and the Creditors' Instructions, and being substantially in the form of **Schedule "D**" hereto;
- 3.28 **"Notice of Revision or Disallowance**" means the notice referred to in paragraph 9 hereof, advising a Creditor that the Monitor has revised or rejected all or part of such Creditor's Claim set out in its Proof of Claim and setting out the reasons for such revision or disallowance, and being substantially in the form of **Schedule "E"** hereto;
- 3.29 "Notice of Dispute" means the notice referred to in paragraph 10 that may be delivered to the Monitor by a Creditor who is an Employee, who is disputing the amount of its Claim as stated in the Employee Claim Statement, and being substantially in the form of Schedule "F" hereto;
- 3.30 **"Person**" means any individual, corporation, limited or unlimited liability company, general or limited partnership, association, trust, unincorporated organization without legal personality, joint venture, governmental body or agency, or any other entity;
- 3.31 **"Plan**" means a plan of compromise or arrangement filed or to be filed by the Applicants pursuant to the CCAA, as such plan may be amended or supplemented from time to time;
- 3.32 **"Proof of Claim**" means (i) for Creditors, other than Employees, the form of Proof of Claim for Creditors referred to in paragraph 9 hereof, in the form of **Schedule "G"** hereto, and (ii) for Employees, the Employee Claim Statement referred to in paragraph 10 hereof;
- 3.33 **"Proven Claim**" means the amount of any Claim of any Creditor as of the Determination Date, determined in accordance with the provisions of the CCAA and this Order, and proven by delivering a Proof of Claim to the Monitor or stated in the Employee Claim Statement, as applicable;
- 3.34 **"Publication Date**" means the date on which the publication of the Newspaper Notice in all of the Designated Newspapers has been completed;

- 3.35 **"Restructuring Claim**" means any right of any Person against any of the Applicants in connection with any indebtedness or obligation of any kind owed to such Person arising out of the restructuring, repudiation, or termination of any contract, lease, employment agreement, collective agreement or other agreement, whether written or oral, after the Determination Date, including any right of any Person who receives a notice of disclaimer, repudiation or termination from any of the Applicants; provided however, that a Restructuring Claim may not include an Excluded Claim;
- 3.36 **"Service List**" means the service list in these CCAA Proceedings; and
- 3.37 **"Unaffected Claim**" shall have the meaning ascribed to such term in the Plan.

Notification Procedure

- 4. The form of Newspaper Notice shall be published by the Monitor in the Designated Newspapers as soon as possible following the issuance of this Order, but in any event no later than on May 17, 2023.
- 5. The Monitor shall publish on its website at <u>http://cfcanada.fticonsulting.com/imv</u>, on or before the day that is five Business Days after the date of this Order, a copy of the Creditors' List, the Creditors' Instructions and this Order.
- 6. The Monitor shall send, by regular mail or by email, a copy of the Creditors' Instructions to each Known Creditor at their last known address or at the address listed on the Service List no later than on May 15, 2023.
- 7. Any notices of disclaimer or resiliation under section 32 of the CCAA delivered by the Applicants after the date of this Order shall be accompanied by a copy of the Creditors' Instructions.

Claims Bar Date

8. Unless otherwise authorized by the Court, a Creditor who does not file a Proof of Claim by the Claims Bar Date (i) shall not be entitled to any further notice; (ii) shall be forever barred from pursuing a Claim against the Applicants or the Directors and Officers; (iii) shall not be entitled to participate as a Creditor in these proceedings; (iv) shall not be entitled to vote on any matter in these proceedings, including the Plan; (v) shall not be entitled to file a Claim against the Applicants or the Directors and Officers; or (vi) shall not be entitled to receive a distribution under the Plan.

Claims Procedure

- 9. The following procedure shall apply where a Creditor files a Proof of Claim before the Claims Bar Date:
 - 9.1 the Monitor, together with the Applicants, shall review the Proof of Claim to value the amounts and terms set out therein for voting and distribution purposes. Where applicable, the Monitor shall send the Creditor, or its legal counsel, a Notice of Revision or Disallowance;

- 9.2 the Creditor who receives a Notice of Revision or Disallowance, personally or through its legal counsel, and wishes to dispute it shall, within ten (10) calendar days of the Notice of Revision or Disallowance, file an Appeal Application with the Court and serve a copy of such Appeal Application to the Applicants and the Monitor;
- 9.3 unless otherwise authorized by the Court, if the Creditor does not file an Appeal Application within the delay provided in paragraph 9.2 above, such Creditor shall be deemed to have accepted the nature and value of its Claim as set out in the Notice of Revision or Disallowance; and
- 9.4 where a Creditor files an Appeal Application, the appeal shall be treated as a true appeal on the record and not an appeal de novo, unless the Court decides that proceeding as a true appeal on the record would result in an injustice to the Creditor.

Employee Claims Procedure

- 10. The following procedure shall apply where an Employee received an Employee Claim Statement:
 - 10.1 if the Employee agrees with the amount of its Claim, as set out in the Employee Claim Statement, no further action will be required by the Employee;
 - 10.2 if the Employee disagrees with the amount of its Claim as set out in the Employee Claim Statement, it shall, before the Claims Bar Date, send by email a Notice of Dispute to the Monitor setting out the basis for its dispute;
 - 10.3 unless otherwise authorized by this Court, if the Employee does not provide a Notice of Dispute within the time period provided for above, the Employee shall be deemed to have accepted the determination of its Claim as set out in the Employee Claim Statement;
 - 10.4 the Monitor, with the assistance of the Applicants, shall attempt to consensually resolve the disputed Claim following the receipt by the Monitor of the Notice of Dispute by the Employee; and
 - 10.5 if, after the expiration of such period of time as the Monitor believes appropriate, the disputed Claim has not been resolved, the Monitor, after consultation with the Applicants, shall refer the Claim to the Court and either the Employee, the Monitor or the Applicants may bring an application for the resolution of such Claim by the Court.

Notices and Communications

11. Any notice, service or other communication to be given under this Order by any Creditor to the Monitor or the Applicants shall be in writing in substantially the form provided for in this Order, where applicable, and will be sufficiently given only if delivered by email, or if a Creditor is unable to do so, and after notifying the Monitor of the method of delivery via the telephone hotline available on the Monitor's Website, by

prepaid registered mail, courier, or personal delivery addressed to:

Monitor:	FTI Consulting Inc. TD South Tower, 79 Wellington Street W Toronto-Dominion Centre Suite 2010, P.O. Box 104 Toronto, ON M5K 1G8 Canada
	C/O
	Mr. Jeffrey Rosenberg Ms. Jodi Porepa
	Email:
	jeffrey.rosenberg@fticonsulting.com jodi.porepa@fticonsulting.com
With a Copy to:	Stikeman Elliott LLP 5300 Commerce Court West 199 Bay Street Toronto, Ontario M5L 1B9 Canada
	C/O
	M ^{tre} Maria Konyukhova
	Email:
	mkonyukhova@stikeman.com
Applicants:	McCarthy Tétrault LLP 1000 De La Gauchetière Street West Suite MZ400 Montréal, Québec, H3B 0A2
	C/O
	M ^{tre} Alain N. Tardif M ^{tre} François Alexandre Toupin
	Email:

atardif@mccarthy.ca fatoupin@mccarthy.ca

12. Any document sent by the Monitor pursuant to this Order may be sent by email, ordinary mail, registered mail or courier. A Creditor shall be deemed to have received any document sent pursuant to this Order two (2) Business Days after the document is sent by mail and one (1) Business Day after the document is sent by courier or email.

Documents shall not be sent by ordinary or registered mail during a postal strike or work stoppage of general application.

Notice to Transferees

- 13. From the date of this Order until seven (7) days prior to the date fixed by the Court for the first distribution in the CCAA Proceedings or any other proceeding, including a bankruptcy, to the extent required, leave is hereby granted to permit a Creditor to provide to the Monitor notice of assignment or transfer of a Claim to any third party.
- 14. Subject to the terms of any subsequent Order of this Court, if, after the Determination Date, the holder of a Claim transfers or assigns its Claim to another Person, none of the Monitor nor any of the Applicants shall be obligated to give notice to or otherwise deal with the transferee or assignee of such Claim in respect thereof unless and until written notice of such transfer or assignment, together with satisfactory evidence of such transfer or assignment, shall have been received by the Monitor and acknowledged by the Applicants or the Monitor in writing and thereafter such transferee or assignee shall, for the purposes hereof, constitute the "Creditor" in respect of such Claim and the Applicants and the Monitor shall thereafter only be required to deal with such transferee or assignee and not the original Creditor. Any such transferee or assignee of a Claim shall be bound by any notices given or steps taken or not taken in respect of such Claim in accordance with this Order prior to receipt by the Monitor and acknowledgement by the Applicants or the Monitor of satisfactory evidence of such transfer or assignment. A transferee or assignee of a Claim takes the Claim subject to any rights of set-off to which the Applicants and/or the applicable Directors and Officers may be entitled with respect to such Claim. For greater certainty, a transferee or assignee of a Claim shall not be entitled to set-off, apply, merge, consolidate or combine any Claim assigned or transferred to it against or on account or in reduction of any amounts owing by such Person to any of the Applicants or the applicable Directors and Officers.
- 15. No transfer or assignment shall be effective for voting purposes at any Creditors' Meeting unless sufficient notice and evidence of such transfer or assignment has been received by the Monitor no later than 5:00 p.m. on the date that is seven (7) days prior to the date fixed by the Court for any Creditors' Meeting, failing which the original Creditor shall have all applicable rights as the "Creditor" with respect to such Claim as if no transfer or assignment of the Claim had occurred.

Aid and Assistance of Other Courts

16. The aid and recognition of any Court, tribunal, regulatory or administrative body in Canada, the United States of America or elsewhere, to give effect to this Order and to assist the Applicants, the Monitor and their respective agents in carrying out the terms of this Order. All Courts, tribunals, regulatory and administrative bodies are hereby respectfully requested to make such orders and to provide such assistance to the Applicants, and the Monitor as may be necessary or desirable to give effect to this Order, to grant representative status to the Monitor or the authorized representative of the Applicants in any foreign proceeding, to assist the Applicants, and the Monitor, and to act in aid of and to be complementary to this Court, in carrying out the terms of this Order.

General Provisions

- 17. For the purposes of this Order, all Claims shall be denominated in United States dollars and all Claims that are not denominated in such currency shall be converted to United States dollars at the Bank of Canada daily average exchange rate in effect on the Determination Date (USD 1 : CAD 1.35). If no currency is specified for the Claim, the currency shall be presumed to be denominated in Canadian dollars.
- 18. The Monitor shall use reasonable discretion as to the forms of the documents to be sent out and received pursuant to this Order and may modify such forms if needed to accomplish the goals of this Order. The Monitor shall use reasonable discretion as to the adequacy of completion and execution of any document completed and executed pursuant to this Order and, where the Monitor is satisfied that any matter to be proven under this Order has been adequately proven, the Monitor may waive strict compliance with the requirements of this Order as to the completion and execution of documents.
- 19. The Monitor may apply to the Court for advice and direction in connection with the discharge or variation of its powers and duties under this Order.
- 20. The provisional execution of this Order notwithstanding appeal, and without requirement to provide any security or provision for costs whatsoever, is hereby ordered.

Issued

, 2023

Prothonotary

Schedule A Service List

SUPREME COURT OF NOVA SCOTIA

IN THE MATTER OF: Application by IMV Inc., Immunovaccine Technologies Inc. and IMV USA Inc. (the "Applicants"), for relief under the *Companies' Creditors Arrangement Act*

SERVICE LIST

Party	Method of Service
McCarthy Tétrault LLP 1000 De La Gauchetière Street West Suite MZ400 Montréal, QC H3B 0A2 <i>Counsel to the Applicants.</i>	M ^{tre} Alain N. Tardif Tel : 514.397.4274 E-mail : atardif@mccarthy.ca M ^{tre} François Alexandre Toupin Tel: 514.397.4210 E-mail: fatoupin@mccarthy.ca
Stewart McKelvey Lawyers Queen's Marque 600-1741 Lower Water Street Halifax, NS B3J 0J2 <i>NS Counsel to the Applicants</i>	M^{tre} Sara L. Scott Tel : 514.397.4274 E-mail : sscott@stewartmckelvey.com
Aird & Berlis LLP Brookfield Place 181 Bay Street, Suite 1800 Toronto, ON M5J 2T9 <i>Counsel to the Horizon Technology Finance</i> <i>Corporation, as agent</i>	M ^{tre} Miranda Spence Tel : 416.865.3414 E-mail : mspence@airdberlis.com M ^{tre} Graham Topa Tel: 416.865.4739 E-mail : gtopa@airdberlis.com M ^{tre} Kyle Plunkett Tel: 416.865.3406 E-mail : kplunkett@airdberlis.com
Ropes & Gray LLP Prudential Tower 800 Boylston Street Boston, MA 02199-3600 US Counsel to the secured creditor Horizon Technology Finance Corporation, as agent	M ^{tre} Cristine Schwarzman Tel: 212.596.9635 E-mail : Cristine.Schwarzman@ropesgray.com M ^{tre} Patricia I. Chen Tel : 617.951.7553 E-mail: patricia.chen@ropesgray.com

FTI Consulting Canada Inc. TD South Tower 79 Wellington Street West Suite 2010, P.O. Box 104 Toronto, ON M5K 1G8 <i>Monitor</i>	Jodi Porepa Tel : 437.332.5743 E-mail : Jodi.Porepa@fticonsulting.com Jeffrey Rosenberg Tel: 416.649.8073 E-mail: Jeffrey.Rosenberg@fticonsulting.com Adsaran Vithiyananthan E-mail : Adsaran.Vithiyananthan@fticonsulting.com
Stikeman Elliot LLP 5300 Commerce Court West 199 Bay Street Toronto ON M5L 1B9	M ^{tre} Maria Konyukhova Tel : 416.869.5230 E-mail : mkonyukhova@stikeman.com <i>Counsel to the Appointed Monitor</i>
Atlantic Canada Opportunities Agency Legal Services Department of Justice / Government of Canada PO Box 6051 644 Main Street Moncton, NB E1C 9J8	M^{tre} Karine LeBlanc Tel: 506.851.2153 Email: <u>karine.leblanc@acoa-apeca.gc.ca</u>
Canada Revenue Agency	General address for service notificationPGC-AGC.fiscal-tax@justice.gc.ca notificationPGC-AGC.civil@justice.gc.ca

Email distribution list:

<u>Atardif@mccarthy.ca; fatoupin@mccarthy.ca; sscott@stewartmckelvey.com;</u> <u>Adsaran.Vithiyananthan@fticonsulting.com; Jodi.Porepa@fticonsulting.com;</u> <u>Jeffrey.Rosenberg@fticonsulting.com; mkonyukhova@stikeman.com;</u> <u>mspence@airdberlis.com; gtopa@airdberlis.com; kplunkett@airdberlis.com;</u> <u>Cristine.Schwarzman@ropesgray.com; patricia.chen@ropesgray.com; karine.leblanc@acoaapeca.gc.ca; notificationPGC-AGC.civil@justice.gc.ca; notificationPGC-AGC.fiscaltax@justice.gc.ca</u>

Schedule B Employee Claim Statement

EMPLOYEE CLAIM STATEMENT

(Letterhead of the Monitor)

[DATE], 2023

[•]

Dear:

RE: EMPLOYEE CLAIM STATEMENT IN THE CCAA PROCEEDINGS OF IMV INC., IMMMUNOVACCINE TECHNOLOGIES INC. AND IMV USA INC.

(COURT FILE NO. HFX NO. 523334)

As you know, IMV Inc., Immunovaccine Technologies Inc., and IMV USA Inc. (collectively, the "**Applicants**") filed for and were granted creditor protection under the *Companies' Creditors Arrangement Act* (Canada) (the "**CCAA**"), pursuant to an order (as amended, the "**Initial Order**") of the Supreme Court of Nova Scotia (the "**Court**") (the "**CCAA Proceedings**"). Pursuant to the Initial Order, the Court appointed FTI Consulting Canada Inc. as monitor of the Applicants (in such capacity, the "**Monitor**") to, among other things, oversee the CCAA Proceedings. Terms not otherwise defined herein have the meaning ascribed to them in the Claims Procedure Order dated May [X] 2023 (the "**Claims Procedure Order**"). Copies of the Initial Order, the Claims Procedure Order and other information relating to the CCAA Proceedings has been posted to: <u>http://cfcanada.fticonsulting.com/imv</u> (the "**Monitor's Website**").

The purpose of this notice is to inform you about the employee claims process (the "**Employee Claims Process**") that was approved by the Court pursuant to the Claims Procedure Order. The Claims Process governs employee claims to be asserted against the Applicants and their Directors and Officers in the CCAA Proceedings (collectively, the "**Employees**" and each an "**Employee**"). The Employee Claims Process, as it relates to Employee claims (each an "**Employee Claim**"), is described below.

EMPLOYEE CLAIMS PROCESS

• Under the Claims Process, the Monitor is required to send a notice prepared by the Applicants, in consultation with the Monitor, to each Employee outlining the quantum of their Employee Claim against the Applicants and the Directors and Officers that the Monitor is prepared to allow (each an "**Employee Claim Statement**").

 <u>This notice is your Employee Claim Statement</u>, and your total Employee Claim has been assessed by the Monitor, in consultation with the Applicants. Please refer to Appendix [•] for a copy of your Employee Claim Statement as of May 1, 2023.

If you agree with the Employee Claim Statement, you need take no further action.

IF YOU WISH TO DISPUTE THE ASSESSMENT OF YOUR EMPLOYEE CLAIM, YOU MUST TAKE THE STEPS OUTLINED BELOW.

Disagreement with Assessment:

If you disagree with the assessment of your Employee Claim Statement set out in this Employee Claim Statement, you must complete and return to the Monitor a completed Notice of Dispute asserting a claim in a different amount supported by the appropriate documentation. A blank Notice of Dispute is enclosed. The Notice of Dispute with supporting documentation disputing the within assessment of your claim **must** be received by the Monitor no later than 5:00 p.m. (Halifax Time) on June 30, 2023 (the "Employee Claims Bar Date").

If no such Notice of Dispute is received by the Monitor by that date, the amount of your claim will be, subject to further order of the Court, conclusively deemed to be as shown in this Employee Claim Statement.

The Notice of Dispute of Employee Claim Statement *will be sufficiently received <u>only</u> if delivered by email*, or if an Employee is unable to do so, and after notifying the Monitor of the method of delivery via the telephone hotline (416.649.8121 or 1.833.860.8353), by prepaid registered mail, courier, or personal delivery addressed to:

FTI Consulting Canada Inc. TD Waterhouse Tower 79 Wellington Street West Suite 2010, P.O. Box 104 Toronto, ON M5K 1G8

Attention: Jeff Rosenberg / Jodi Porepa

Email: imv@fticonsulting.com

with copies to:

Stikeman Elliott LLP 5300 Commerce Court West 199 Bay Street Toronto, Ontario M5L 1B9

Attention: Maria Konyukhova

Email: mkonyukhova@stikeman.com

Any Notice of Dispute shall be deemed to be received upon actual receipt thereof before 5:00 p.m. (Halifax Time) on a Business Day or if delivered outside of normal business hours, the next Business Day.

IMPORTANT DEADLINE:

If you do not file a Notice of Dispute by the Employee Claims Bar Date, you will have no further right to dispute your Employee Claim Statement against any of the Applicants or the Directors and Officers, as assessed by Monitor, in consultation with the Applicants, and you will be barred from filing any such dispute in the future.

More Information:

If you have questions regarding the foregoing, you may contact the Monitor at <u>imv@fticonsulting.com</u> or via the telephone hotline (416.649.8121 or 1.833.860.8353).

Schedule C Instruction Letter

SCHEDULE "C"

IN THE MATTER OF THE COMPANIES' CREDITORS ARRANGEMENT ACT, R.S.C. 1985, c. C-36, AS AMENDED

AND IN THE MATTER OF A PLAN OF COMPROMISE OR ARRANGEMENT OF IMV INC., IMMUNOVACCINE TECHNOLOGIES INC. AND IMV USA INC. (collectively, the "**Applicants**")

INSTRUCTION LETTER FOR THE CLAIMS PROCEDURE

I. CLAIMS PROCEDURE

By Order of the Supreme Court of Nova Scotia dated May 5, 2023 (the "**Claims Procedure Order**"), FTI Consulting Canada Inc., in its capacity as the Court-appointed monitor (in such capacity, the "**Monitor**") of the Applicants has been authorized, with the assistance of the Applicants, to conduct a claims procedure (the "**Claims Procedure**") with respect to Claims against the Applicants and their present or former Directors and Officers. The Claims Procedure Order governs the filing and determination of all Claims against the Applicants.

Unless otherwise defined, all capitalized terms used herein shall have the meanings ascribed to them in the Claims Procedure Order.

The Claims Procedure Order, the Claims Package, a Proof of Claim form and related materials may be accessed from the Monitor's Website at <u>http://cfcanada.fticonsulting.com/imv</u>

This letter provides instructions for responding to or completing the Proof of Claim. Reference should be made to the Claims Procedure Order for a complete description of the Claims Procedure.

The Claims Procedure is intended for any Person with any Claims, other than Excluded Claims, of any kind or nature whatsoever against the Applicants, the Directors or Officers or any of them, whether liquidated, unliquidated, contingent or otherwise. Please review the enclosed material for the complete definitions of "Claim", "Claims Bar Date", "Claims against the Directors and Officers", "Creditor", "Equity Claim", "Excluded Claim", "Excluded Creditor", "Known Creditor", "Restructuring Claim", "Unaffected Claim" to which the Claims Procedure applies.

All enquiries with respect to the Claims Procedure should be addressed to the Monitor at: <u>imv@fticonsulting.com</u> or via the telephone hotline (416.649.8121 or 1.833.860.8353), provided however, that formal notices to the Monitor must be delivered as set out below.

II. FOR CLAIMANTS SUBMITTING A PROOF OF CLAIM

If you believe that you have a Claim (other than an Excluded Claim) that you wish to assert against the Applicants, the Directors or Officers or any of them, you **MUST** file a Proof of Claim with the Monitor.

All Proofs of Claim for (i) Pre-Filing Claims, which for greater certainty are Claims against the Applicants arising prior to the Filing Date of May 1, 2023; and (ii) D&O Pre-Filing Claims must be received by the Monitor before 5:00 p.m. (Halifax Time) on July 31, 2023 (the "Pre-Filing Claims Bar Date") and before 5:00 p.m. (Halifax Time) on August 29, 2023 (the "D&O Claims Bar Date"), respectively.

All Proofs of Claim for (i) Restructuring Claims, which for greater certainty are Claims arising out of the restructuring, disclaimer, resiliation, termination or breach by the Applicants on or after the Filing Date of May 1, 2023 of any contract, lease or other agreement or arrangement whether written or oral, and (ii) D&O Restructuring Claims, must be received by the Monitor by the later of (i) the Pre-Filing Claims Bar Date and (ii) 5:00 p.m. (Halifax Time) on the date that is thirty (30) days after the date of receipt of a notice from the Applicants giving rise to the Restructuring Claim (the "Restructuring Claims Bar Date").

Any notice or communication required to be provided or delivered, including for greater certainty any Proof of Claim, shall be in writing in substantially the form, if any, provided for in the Claims Procedure Order and *will be sufficiently given <u>only</u> if delivered by email*, or if a Claimant is unable to do so, and after notifying the Monitor of the method of delivery via the telephone hotline (416.649.8121or 1.833.860.8353), by prepaid registered mail, courier, or personal delivery, addressed to:

If to the Applicants:

McCarthy Tétrault LLP 1000 De La Gauchetière Street West, MZ400 Montreal, QC H3B 0A2

Attention: Alain N. Tardif/ François Alexandre Toupin

Email: atardif@mccarthy.ca / fatoupin@mccarthy.ca

If to the Monitor:

FTI Consulting Canada Inc. TD Waterhouse Tower 79 Wellington Street West Suite 2010, P.O. Box 104 Toronto, ON M5K 1G8

Attention: Jeff Rosenberg / Jodi Porepa

Email: <u>imv@fticonsulting.com</u>

with copies to:

Stikeman Elliott LLP 5300 Commerce Court West 199 Bay Street Toronto, Ontario M5L 1B9

Attention: Maria Konyukhova

Email: mkonyukhova@stikeman.com

Any such notice or communication delivered by a Claimant shall be deemed to be received upon actual receipt thereof before 5:00 p.m. (Halifax Time) on a Business Day or if delivered outside of normal business hours, the next Business Day.

PROOFS OF CLAIM MUST BE RECEIVED BY THE PRE-FILING CLAIMS BAR DATE OR THE RESTRUCTURING CLAIMS BAR DATE, AS APPLICABLE, OR THE APPLICABLE CLAIM WILL BE FOREVER BARRED AND EXTINGUISHED. If you are required to file a Proof of Claim pursuant to the Claims Procedure but do not file a Proof of Claim in respect of a Claim by the Pre-Filing Claims Bar Date or the Restructuring Claims Bar Date, as applicable, you shall not be entitled to vote at any Meeting regarding a Plan or participate in any distribution under a Plan or otherwise in respect of such Claims.

All Claims denominated in foreign currency shall be converted to United States dollars at the Bank of Canada daily average exchange rate in effect on the Filing Date of May 1, 2023 (USD 1 : CAD 1.35).

Additional Proof of Claim forms can be obtained by contacting the Monitor at the telephone number and address indicated above and providing particulars as to your name, address and facsimile number or email mail address. In addition, Proofs of Claim and related materials may be accessed from the Monitor's Website at <u>http://cfcanada.fticonsulting.com/imv</u>

DATED at Toronto, Ontario this $[X^{st}]$ day of May, 2023.

FTI Consulting Canada Inc., solely in its capacity as Monitor of the Applicants and not in its personal capacity.

Schedule D Newspaper Notice

SCHEDULE "D"

IN THE MATTER OF THE COMPANIES' CREDITORS ARRANGEMENT ACT, R.S.C. 1985, c. C-36, AS AMENDED

AND IN THE MATTER OF A PLAN OF COMPROMISE OR ARRANGEMENT OF IMV INC., IMMUNOVACCINE TECHNOLOGIES INC. AND IMV USA INC. (collectively, the "Applicants")

NOTICE LETTER FOR THE CLAIMS PROCEDURE

RE: NOTICE OF CLAIMS PROCEDURE, PRE-FILING CLAIMS BAR DATE & RESTRUCTURING CLAIMS BAR DATE

This notice is being published pursuant to an Order of the Supreme Court of Nova Scotia dated May 5, 2023 (the "**Claims Procedure Order**"), in the Applicants' proceedings under the *Companies' Creditors Arrangement Act*, R.S.C., 1985, c. C-36, as amended. Pursuant to the Initial Order dated May 1, 2023, FTI Consulting Canada Inc. was appointed as monitor of the Applicants (in such capacity, the "**Monitor**"), and pursuant to the Claims Procedure Order will, with the assistance of the Applicants, conduct a claims procedure (the "**Claims Procedure**") with respect to Claims against the Applicants and their present and former Directors and Officers. Additionally, the Monitor is required to send Claims Packages to, among others, the Applicants' Known Creditors. All capitalized terms not defined herein shall have the meanings ascribed to them in the Claims Procedure Order.

The Claims Procedure Order, the Claims Package, a Proof of Claim and related materials may be accessed from the Monitor's Website at http://cfcanada.fticonsulting.com/imv

I. SUBMISSION OF A PROOF OF CLAIM

All persons wishing to assert a Claim (other than an Excluded Claim) against the Applicants or their Directors or Officers **MUST** file a Proof of Claim with the Monitor.

The Pre-Filing Claims Bar Date is 5:00 p.m. (Halifax Time) on July 31, 2023 (the "Pre-Filing Claims Bar Date") and before 5:00 p.m. (Halifax Time) on August 29, 2023 (the "D&O Claims Bar Date"). Proofs of Claim in respect of Pre-Filing Claims, including D&O Pre-Filing Claims must be completed and filed with the Monitor on or before the Pre-Filing Claims Bar Date and the D&O Claims Bar Date, respectively.

The Restructuring Claims Bar Date is the later of (i) the Pre-Filing Claims Bar Date and (ii) 5:00 p.m. (Halifax Time) on the date that is thirty (30) days after the date of receipt of a notice from the Debtors giving rise to the Restructuring Claim (the "Restructuring Claims Bar Date"). Proofs of Claim in respect of Restructuring Claims and D&O Restructuring Claims must be completed and filed with the Monitor on or before the Restructuring Claims Bar Date.

Any notice or communication required to be provided or delivered, including for greater certainty any Proof of Claim, shall be in writing in substantially the form, if any, provided for in the Claims Procedure Order

and *will be sufficiently given <u>only</u> if delivered by email*, or if a Claimant is unable to do so, and after notifying the Monitor of the method of delivery via the telephone hotline (416.649.8121 or 1.833.860.8353), by prepaid registered mail, courier, or personal delivery, addressed to:

If to the Applicants:

McCarthy Tétrault LLP 1000 De La Gauchetière Street West, MZ400 Montreal, QC H3B 0A2

Attention: Alain N. Tardif/ François Alexandre Toupin

Email: atardif@mccarthy.ca / fatoupin@mccarthy.ca

If to the Monitor:

FTI Consulting Canada Inc. TD Waterhouse Tower 79 Wellington Street West Suite 2010, P.O. Box 104 Toronto, ON M5K 1G8

Attention: Jeff Rosenberg / Jodi Porepa

Email: <u>imv@fticonsulting.com</u>

with copies to:

Stikeman Elliott LLP 5300 Commerce Court West 199 Bay Street Toronto, Ontario M5L 1B9

Attention: Maria Konyukhova

Email: mkonyukhova@stikeman.com

Any such notice or communication delivered by a Claimant shall be deemed to be received upon actual receipt thereof before 5:00 p.m. (Halifax Time) on a Business Day or if delivered outside of normal business hours, the next Business Day.

PROOFS OF CLAIM MUST BE RECEIVED BY THE MONITOR BY THE PRE-FILING CLAIMS BAR DATE OR RESTRUCTURING CLAIMS BAR DATE, AS APPLICABLE, OR THE CLAIM WILL BE FOREVER BARRED AND EXTINGUISHED. If you are required to file a Proof of Claim

pursuant to the Claims Procedure but do not file a Proof of Claim in respect of a Claim by the Pre-Filing Claims Bar Date or the Restructuring Claims Bar Date, as applicable, you shall not be entitled to vote at any Meeting regarding a Plan or participate in any distribution under a Plan, if any, or otherwise in respect of such Claims.

Reference should be made to the enclosed material for the complete definitions of "Claim", "Claims Bar Date", "Claims against the Directors and Officers", "Creditor", "Equity Claim", "Excluded Claim", "Excluded Creditor", "Known Creditor", "Restructuring Claim", "Unaffected Claim" to which the Claims Procedure applies.

III. MONITOR CONTACT INFORMATION

All enquiries with respect to the Claims Procedure should be addressed to the Monitor at: Email: <u>imv@fticonsulting.com</u> or via the telephone hotline (416.649.8121 or 1.833.860.8353), provided however, that formal notices to the Monitor must be delivered as set out above.

DATED at Toronto, Ontario this $[X^{st}]$ day of May, 2023.

FTI Consulting Canada Inc., solely in its capacity as Monitor of the Applicants and not in its personal capacity.

Schedule E Notice of Revision or Disallowance

SCHEDULE "E"

IN THE MATTER OF THE COMPANIES' CREDITORS ARRANGEMENT ACT, R.S.C. 1985, c. C-36, AS AMENDED

AND IN THE MATTER OF A PLAN OF COMPROMISE OR ARRANGEMENT OF IMV INC., IMMUNOVACCINE TECHNOLOGIES INC. AND IMV USA INC. (collectively, the "**Applicants**")

NOTICE OF REVISION OR DISALLOWANCE

TO: [name and address of Claimant]

PLEASE TAKE NOTICE that this Notice of Revision or Disallowance is being sent pursuant to an order of the Supreme Court of Nova Scotia dated May 5, 2023 (the "**Claims Procedure Order**"). All capitalized terms not otherwise defined in this Notice of Revision or Disallowance shall have the meaning ascribed to them in the Claims Procedure Order, which is available on the Monitor's Website at http://cfcanada.fticonsulting.com/imv

The Monitor has reviewed your Proof of Claim dated ______2023, and has revised or disallowed your Claim for the following reasons:

Subject to further dispute by you in accordance with the provisions of the Claims Procedure Order, your Claim will be as follows:

Claim Against	Type of Claim per Proof of Claim	Amount of Claim per Proof of Claim	Type of Claim per this Notice of Revision or Disallowance	Amount of Claim per this Notice of Revision or Disallowance
[Insert name of appropriate party]	[Pre-Filing Claim/ Restructuring Claim/D&O Claim] [Unsecured Claim/ Unsecured Priority Claim/ Secured Claim]	CA\$	[Pre-Filing Claim/ Restructuring Claim/D&O Claim] [Unsecured Claim/ Unsecured Priority Claim / Secured Claim]	CA\$

If you intend to dispute this Notice of Revision or Disallowance, you shall, within ten (10) calendar days of the Notice of Revision or Disallowance, file an Appeal application with the Court and serve a copy of such Appeal Application to the Debtors and the Monitor in writing which *will be sufficiently given only if delivered by email*, or if you are unable to do so, and after notifying the Monitor of the method of delivery via the telephone hotline (416.649.8121 or 1.833.860.8353), by prepaid registered mail, courier, or personal delivery, addressed to:

If to the Applicants:

McCarthy Tétrault LLP 1000 De La Gauchetière Street West, MZ400 Montreal, QC H3B 0A2

Attention: Alain N. Tardif/ François Alexandre Toupin

Email: <u>atardif@mccarthy.ca</u> / <u>fatoupin@mccarthy.ca</u>

If to the Monitor:

FTI Consulting Canada Inc. TD Waterhouse Tower 79 Wellington Street West Suite 2010, P.O. Box 104 Toronto, ON M5K 1G8

Attention: Jeff Rosenberg / Jodi Porepa Email: <u>imv@fticonsulting.com</u>

with copies to:

Stikeman Elliott LLP 5300 Commerce Court West 199 Bay Street Toronto, Ontario M5L 1B9

Attention: Maria Konyukhova

Email: mkonyukhova@stikeman.com

If you agree with this Notice of Revision or Disallowance, there is no need to file anything further with the Monitor.

DATED this _____ day of _____, 2023.

FTI Consulting Canada Inc., solely in its capacity as Monitor of the Applicants and not in its personal capacity.

Schedule F Notice of Dispute

SCHEDULE "F"

IN THE MATTER OF THE *COMPANIES' CREDITORS* ARRANGEMENT ACT, R.S.C. 1985, c. C-36, AS AMENDED

AND IN THE MATTER OF A PLAN OF COMPROMISE OR ARRANGEMENT OF IMV INC., IMMUNOVACCINE TECHNOLOGIES INC. AND IMV USA INC. (collectively, the "**Applicants**")

NOTICE OF DISPUTE

(With Respect to the Applicants and their Directors & Officers)

Capitalized terms not defined herein have the meaning ascribed to them in the Order of the Supreme Court of Nova Scotia dated May 5, 2023 (the "**Claims Procedure Order**").

Claims Reference Number:

1. Particulars of Claimant:

Full Legal Name of Employee Claimant (include trade name, if applicable)

(the "Employee Claimant")

Full Mailing Address of the Employee Claimant:

Other Contact Information of the Claimant:

 2. Particulars of original Employee from whom you acquired the Employee Claim (if applicable):

Have you acquired this purported Employee Claim from an Employee by assignment?

Yes: No:

If yes and if not already provided, attach documents evidencing assignment.

Full Legal Name of original Employee Claimant:

3. Dispute of Employee Claim Statement:

The Employee Claimant hereby disagrees with the value of its Employee Claim as set out in the Employee Claim Statement and asserts a Claim as follows¹:

	Amount in Employee Claim Statement	Amounts Claimed by Employee Claimant	Entity or Director(s) and/or Officers Against which Employee Claim is Asserted
Total Claim	\$	\$	

4. Reasons for Dispute:

Please describe the reasons and basis for your dispute of the amount set out in your Employee Claim Statement. You may attach a separate schedule if more space is required.

¹All Employee Claims shall be denominated in Canadian dollars. Any Employee Claims denominated in a foreign currency shall be converted United States dollars at the Bank of Canada daily average exchange rate on the Filing Date of May 1, 2023 (USD 1 : CAD 1.35).

DATED this _____ day of _____, 2023.

Signature of Claimant or its Authorized Signatory

If in response to an Employee Claim Statement, this Notice of Dispute MUST be delivered to the Monitor at the below address such that it is received by the Monitor by no later than 5:00 p.m. (Halifax Time) on June 30, 2023 (the "Employee Claims Bar Date").

This Notice of Dispute must be delivered in writing and *will be sufficiently given <u>only</u> if delivered by email*, or if you are unable to do so, and after notifying the Monitor of the method of delivery via the telephone hotline (416.649.8121 or 1.833.860.8353), by prepaid registered mail, courier, or personal delivery, addressed to:

FTI Consulting Canada Inc. TD Waterhouse Tower 79 Wellington Street West Suite 2010, P.O. Box 104 Toronto, ON M5K 1G8

Attention: Jeff Rosenberg / Jodi Porepa Email: <u>imv@fticonsulting.com</u>

with copies to:

Stikeman Elliott LLP 5300 Commerce Court West 199 Bay Street Toronto, Ontario M5L 1B9

Attention : Maria Konyukhova Email : <u>mkonyukhova@stikeman.com</u>

Any Notice of Dispute delivered shall be deemed to be received upon actual receipt thereof before 5:00 p.m. (Halifax Time) on a Business Day or if delivered outside of normal business hours, the next Business Day.

If a completed Notice of Dispute is not received by the Monitor by the dates set out in the Claims Procedure Order and described herein, the Claimant shall be forever barred from disputing the classification, amount or nature of the Employee Claim Statement.

IMPORTANT DEADLINE:

If you do not file a Notice of Dispute by the Employee Claims Bar Date, you will have no further right to dispute your Employee Claim Statement against any of the Applicants or the Directors and Officers, as assessed by Monitor, in consultation with the Applicants, and you will be barred from filing any such dispute in the future. THE CLAIM AS SET OUT IN THE EMPLOYEE CLAIM STATEMENT SENT TO YOU WILL BE DEEMED TO BE YOUR CLAIM AND WILL BE FINAL AND BINDING.

Schedule G Proof of Claim

SCHEDULE "G"

IN THE MATTER OF THE COMPANIES' CREDITORS ARRANGEMENT ACT, R.S.C. 1985, c. C-36, AS AMENDED

AND IN THE MATTER OF A PLAN OF COMPROMISE OR ARRANGEMENT OF IMV INC., IMMUNOVACCINE TECHNOLOGIES INC. AND IMV USA INC. (collectively, the "**Applicants**")

PROOF OF CLAIM

Please read carefully the enclosed Instruction Letter for completing this Proof of Claim. All capitalized terms not defined herein have the meaning ascribed to them in the Claims Procedure Order dated May 5, 2023.

I. PARTICULARS OF CLAIMANT:

1. Full Legal Name of Claimant:

(the "Claimant")

2. Full Mailing Address of the Claimant:

3. Telephone Number:

5. Facsimile Number:

6. Attention (Contact Person):

7. Have you acquired this Claim by assignment?

Yes: \Box No: \Box (if yes, attach documents evidencing assignment)

If Yes, Full Legal Name of Original Claimant(s):

II. **PROOF OF CLAIM:**

E-Mail Address:

4.

1. I, _____ (name of Claimant or Representative of the Claimant), of _____

			do	hereby cer	tify:		
(city	and pro	ovince)					
(a)	that	I check (\checkmark) one					
		am the Claimant; OR					
		am	(state	position	or	title)	of
	(nam	e of Claimant)					

- (b) that I have knowledge of all the circumstances connected with the Claim referred to below;
- (c) that complete documentation in support of the Claim referred to below is attached; and
- (d) that the Applicants and/or one or more of the Directors or Officers of the Applicants were and still are indebted to the Claimant as follows:¹

III. SUBMISSION OF A PRE-FILING CLAIM PROOF OF CLAIM

Debtor	Pre-Filing Claim Amount	Whether Claim is Secured, Priority Unsecured, or Unsecured	Value of Security Held, if any:
IMV Inc.			
Directors and Officers of IMV Inc.			
(insert names above)			
Immunovaccine Technologies Inc.			
Directors and Officers of Immunovaccine Technologies Inc.			
(insert names above)			

¹ All Claims denominated in foreign currency shall be converted to United States dollars at the Bank of Canada daily average exchange rate in effect on the Filing Date of May 1, 2023 (USD 1 : CAD 1.35).

Debtor	Pre-Filing Claim Amount	Whether Claim is Secured, Priority Unsecured, or Unsecured	Value of Security Held, if any:
IMV USA Inc.			
Directors and Officers of IMV USA Inc.			
(insert names above)			

IV. SUBMISSION OF A RESTRUCTURING CLAIM PROOF OF CLAIM

Debtor	Restructuring Claim Amount	Whether Claim is Secured, Priority Unsecured, or Unsecured	Value of Security Held, if any:
IMV Inc.			
Directors and Officers of IMV Inc.			
(insert names above)			
Immunovaccine Technologies Inc.			
Directors and Officers of Immunovaccine Technologies Inc.			
(insert names above)			
IMV USA Inc.			

Directors and Officers of IMV USA Inc.		
(insert names above)		

V. PARTICULARS OF CLAIM

The particulars of the undersigned's total Claim (including Pre-Filing Claims, Restructuring Claims and D&O Claims) are attached.

(Provide full particulars of the Claim and supporting documentation, including amount, description of transaction(s) or agreement(s) or legal breach(es) giving rise to the Claim, name of any guarantor(s) which has guaranteed the Claim, particulars and copies of any security and amount of Claim allocated thereto, date and number of all invoices, particulars of all credits, discounts, etc. claimed. If a Claim is made against any Directors or Officers, specify the applicable Directors or Officers and the legal basis for the Claim against each of them.)

VI. FILING OF CLAIM

For Pre-Filing Claims (including D&O Pre-Filing Claims), this Proof of Claim MUST be received by the Monitor before 5:00 p.m. (Halifax Time) on July 31, 2023 (the "Pre-Filing Claims Bar Date") and before 5:00 p.m. (Halifax Time) on August 29, 2023 (the "D&O Claims Bar Date").

For Restructuring Claims (including D&O Restructuring Claims), this Proof of Claim **MUST** be received by the Monitor **before the later of (i) the Pre-Filing Claims Bar Date and (ii) 5:00 p.m. (Halifax Time) on the date that is thirty (30) days after the date of receipt of a notice from the Debtors giving rise to the Restructuring Claim** (the "**Restructuring Claims Bar Date**").

In either case, this Proof of Claim shall be delivered in writing and *will be sufficiently given <u>only</u> if delivered by email*, or you unable to do so, and after notifying the Monitor of the method of delivery via the telephone hotline (416.649.8121 or 1.833.860.8353), by prepaid registered mail, courier, or personal delivery, addressed to:

FTI Consulting Canada Inc. TD Waterhouse Tower 79 Wellington Street West Suite 2010, P.O. Box 104 Toronto, ON M5K 1G8

Attention: Jeff Rosenberg / Jodi Porepa

Email: <u>imv@fticonsulting.com</u>

with copies to:

Stikeman Elliott LLP 5300 Commerce Court West 199 Bay Street Toronto, Ontario M5L 1B9

Attention: Maria Konyukhova

Email: mkonyukhova@stikeman.com

Any Proof of Claim delivered shall be deemed to be received upon actual receipt thereof before 5:00 p.m. (Halifax Time) on a Business Day or if delivered outside of normal business hours, the next Business Day.

Failure to file your Proof of Claim as directed by the Pre-Filing Claims Bar Date, Restructuring Claims Bar Date or D&O Claims Bar Date, as applicable, will result in your Claim being extinguished and forever barred and in you being prevented from making or enforcing a Claim against the Applicants or any of its present or former Directors and Officers.

DATED at ______ this _____ day of _____, 2023.

Signature of Claimant

Schedule "D" Redline Draft Amended and Restated Initial Order to Model Order

<u>20</u> 2023	– No. <u>Hfx No. 523334</u>
	SUPREME COURT OF NOVA SCOTIA
	Supreme Court of Nova Scotia
	-Application by

IN THE MATTER OF: Application by IMV Inc. and Immunovaccine Technologies Inc. and IMV USA Inc. (the "Applicants"), for relief under the Companies' Creditors Arrangement Act

Amended and Restated Initial Order

Before the Honourable Justice [name or blank] in chambers :

<u>Party</u>

The <u>Applicant proposes Applicants propose</u> to make a compromise or arrangement under the *Companies' Creditors Arrangement Act*, R.S.C. 1985, c. C-36 as amended (the "**CCAA**") and <u>itthey</u> applied for an initial order and, now or in the future, other relief under the CCAA as may be sought on notice of motion.

The following parties received notice of this application: see attached at Schedule "A".

The following parties, represented by the following counsel, made submissions:

Counsel

Applicant	
<u>Applicants</u>	McCarthy Tétrault LLP Alain N. Tardif François Alexandre Toupin
	<u>Stewart McKelvey Lawyers</u> Sara L. Scott
<u>Monitor, FTI</u> Consulting Canada Inc.	Stikeman Elliott LLP Maria Konyukhova
Horizon Technology Finance Corporation, as agent	Aird & Berlis LLP Miranda Spence Kyle Plunkett
Ŧ	

On motion of the <u>ApplicantApplicants</u>, the following is ordered and declared:

Service

1. The service of the <u>noticeNotice</u> of <u>application in chambersMotion</u>, the <u>associated</u> <u>pleadings</u>, and the supporting documents, as set out in the affidavit of service is hereby deemed adequate notice¹ <u>abridged and validated</u> so that <u>the motion</u> this Motion is properly returnable today <u>-</u>[and further service thereof is hereby dispensed with]².

Application

2. The <u>Applicant is a companyApplicants are affiliated debtor companies within the</u> meaning of the CCAA and are companies to which the CCAA applies³.

Effective Time

3. <u>This Order and all of its provisions are effective as of 12:01 a.m. Halifax time, province of Nova Scotia, on May 1, 2023 (the "Effective Time")</u>.

Plan of Arrangement

4. <u>3. The ApplicantApplicants</u>, in consultation with the Monitor, shall have the authority to file and may, subject to further order of this Court, file with this Court a plan of compromise or arrangement (the ""Plan"").

Administrative Consolidation

- 5. <u>The CCAA proceedings of the Applicants shall be consolidated under one single Court</u> <u>file, in file Hfx No. 523334.</u>
- 6. <u>All proceedings, filings, and other matters in the CCAA proceedings shall be filed jointly</u> and together by the Applicants under file Hfx No. 523334.
- 7. The consolidation of these CCAA proceedings in respect of the Applicants shall be for administrative purposes only and shall not effect a consolidation of the assets and property or of the debts and obligations of each of the Applicants including, without limitation, for the purposes of any Plan or Plans that may be hereafter proposed.

Possession of Property and Operations

4. The <u>ApplicantApplicants</u> shall remain in possession and control of <u>itstheir</u> current and future assets, undertakings, and properties of every nature and kind whatsoever, and wherever <u>situatesituated</u> including all proceeds thereof (the "Property"). Subject to

⁴ The Applicant should seek to have service deemed adequate if it was done in a manner other than as authorized by the Rules of Court.

² This provision should only be used when all parties entitled to notice have been served with notice of the application. If all parties entitled to notice have not been served then the section should be deleted and the Initial Order should provide for a motion hearing.

³ If there are multiple applicants, the Order should confirm that the applicants are "affiliated debtor companies" within the meaning of the CCAA.

further order of this Court, the <u>ApplicantApplicants</u> shall continue to carry on business in a manner consistent with the preservation of <u>itstheir</u> business (the <u>"</u>"Business"") and Property. The <u>ApplicantApplicants</u> shall be authorized and empowered to continue to retain and employ consultants, agents, experts, accountants, counsel, and such other persons (collectively

- ""Assistants") and the employees currently retained or employed by it, with liberty to retain such further Assistants as it deems reasonably necessary or desirable in the ordinary course of business or for the carrying out of the terms of this Order.
- 9. The Applicants shall be entitled to continue to utilize the central cash management system currently in place as described in the Affidavit of Andrew Hall dated April 28, 2023, or replace it with another substantially similar central cash management system (the "Cash Management System") and that any present or future bank providing the Cash Management System shall not be under any obligation whatsoever to inquire into the propriety, validity or legality of any transfer, payment, collection or other action taken under the Cash Management System, or as to the use or application by the Applicant of funds transferred, paid, collected or otherwise dealt with in the Cash Management System, shall be entitled to provide the Cash Management System without any liability in respect thereof to any Person (as defined hereinafter) other than the Applicants, pursuant to the terms of the documentation applicable to the Cash Management System, an unaffected creditor under the Plan with regard to any claims or expenses it may suffer or incur in connection with the provision of the Cash Management System.
- 10. <u>5. 4 The Applicant Applicants</u> may pay the following expenses whether incurred prior to or after this Order:
 - (a) a.-all outstanding and future wages, salaries, employee and pension benefits, vacation pay, <u>severance</u> and expenses payable to employees who continue to provide service on or after the date of this Order ("Active Employees"), in each case incurred in the ordinary course of business and consistent with existing compensation policies and arrangements; and

⁴ If the Applicant has a central cash management system, the provision below may be inserted in advance of paragraph 5 above. This provision should only be utilized where necessary, in view of the fact that central cash management systems often operate in a manner that consolidates the cash of Applicant. Specific attention should be paid to cross border and inter company transfers of cash. If there are multiple Applicant companies, it may be appropriate to create an inter company charge that provides a charge against the assets of one applicant company for any amount advanced from another applicant company.

"5. _____The Applicant shall be entitled to continue to utilize the central cash management system currently in place as described in the Affidavit of _______ or replace it with another substantially similar central cash management system (the "Cash Management System") and that any present or future bank providing the Cash Management System shall not be under any obligation whatsoever to inquire into the propriety, validity or legality of any transfer, payment, collection or other action taken under the Cash Management System, or as to the use or application by the Applicant of funds transferred, paid, collected or otherwise dealt with in the Cash Management System, shall be entitled to provide the Cash Management System without any liability in respect thereof to any Person (as hereinafter defined) other than the Applicant, pursuant to the terms of the documentation applicable to the Cash Management System, and shall be, in its capacity as provider of the Cash Management System, an unaffected creditor under the Plan with regard to any claims or expenses it may suffer or incur in connection with the provision of the Cash Management System."

- (b) b.-all existing and future employee health, dental, life insurance, short and long term disability and related benefits (collectively, the "Group Benefits") payable on or after the date of this Order to Active Employeesemployees on or after the date of this Order, in each case incurred in the ordinary course of business and consistent with existing policies and arrangements or such amended policies and arrangements as are necessary or desirable to deliver the existing Group Benefits;
- (c) e. with prior written approval of the Monitor, the fees and disbursements for any Assistants retained or employed by the Applicant in respect of these proceedings, at their reasonable standard rates and charges; and
- (d) with the consent of the Monitor, amounts owing for goods or services actually provided to the Applicants prior to the Effective Time by third parties, if, in the opinion of IMV, such third party is critical to the Business and ongoing operations of IMV and such third party would sustain material prejudice if such payment is not made, up to a maximum aggregate amount of \$350,000.
- 11. 6. Except as otherwise provided to the contrary herein, the <u>ApplicantApplicants</u> may pay all reasonable expenses incurred by the <u>ApplicantApplicants</u> in carrying on the Business in the ordinary course after this Order, and in carrying out the provisions of this Order, which expenses shall include, without limitation:
 - (a) a. all expenses and capital expenditures reasonably necessary for the preservation of the Property or the Business including, without limitation, payments on account of insurance including directors and officers insurance, maintenance, and security services; and
 - (b) b. payment for goods or services actually supplied to the <u>ApplicantApplicants</u> following the date of this Order.
- 12. 7. The <u>Applicant Applicants</u> shall remit or pay, in accordance with legal requirements or on terms as may be agreed to between the <u>Applicant Applicants</u> and the applicable authority:
 - (a) a. any statutory deemed trust amounts in favour of the Crown in right of Canada or of any Province thereof or any other taxation authority which are required to be deducted from employees' wages, including, without limitation, amounts in respect of: i) employment insurance, ii) Canada Pension Plan, iii) Quebec Pension Plan, and iv) income taxes;
 - (b) b.-all goods and services or other applicable sales taxes (collectively, ""Sales **Taxes**") required to be remitted by the <u>ApplicantApplicants</u> in connection with the sale of goods and services by the <u>ApplicantApplicants</u>, but only where such Sales Taxes are accrued or collected after the date of this Order, or where such <u>Sales Taxes were accrued or collected prior to the date of this Order but not</u> required to be remitted until on or after the date of this Order; and
 - (c) e. any amount payable to the Crown in right of Canada or of any Province or any regulatory or administrative body or any other authority, in all cases in respect of municipal realty, municipal business, or other taxes, assessments or levies of

any nature or kind which are: (i) entitled at law to be paid in priority to claims of secured creditors; (ii) attributable to or in respect of the ongoing Business carried on by the <u>ApplicantApplicants</u>; and (iii) payable in respect of the period commencing on or after the date of this Order.

- 13. 8. Until such time as the Applicant [disclaims/Applicants disclaims or resiliates]⁵ resiliate a real property lease in accordance with the CCAA, the ApplicantApplicants shall pay all amounts constituting rent or payable as rent under real property leases, including, for greater certainty, common area maintenance charges, utilities and realty taxes, and any other amounts payable to the landlord under the lease, or as otherwise may be negotiated between the ApplicantApplicants and the landlord from time to time (""Rent""), for the period commencing from and including the date of this Order, in accordance with its existing lease agreements. On the date of the first of such payments, any arrears relating to the period commencing from and including the date of this Order shall also be paid.
- 14. 9. Except as specifically permitted herein or by further order of this Court⁶, the <u>Applicant</u> is<u>Applicants are</u> hereby directed, until further order of this Court: (i) to make no payments of principal, interest thereon or otherwise on account of amounts owing by the <u>ApplicantApplicants</u> to any of its<u>their</u> creditors as of this date without prior written consent of the Monitor; (ii) to grant no security interests, trusts, liens, charges, or encumbrances upon or in respect of any of its Property; and (iii) to not grant credit or incur liabilities except in the ordinary course of the Business or with the prior written approval of the Monitor.

Restructuring

- 15. <u>10.</u> The <u>ApplicantApplicants</u> shall, subject to such requirements as are imposed by the Monitor and under any agreements for debtor in possession financing which may be granted, have the right to:
 - (a) a.-permanently or temporarily cease, downsize or shut down any of its business or their operations;
 - b. [terminate the employment of such of its employees or temporarily lay off such of its employees or locations as it deems they deem appropriate and, as applicable, make provision for the consequences thereof in accordance with the terms of any collective agreement]Plan;⁷
 - (b) e. pursue all avenues of refinancing to finance or refinance, market, convey, transfer, assign or in any other manner dispose of its the Business or Property, in whole or part, subject to prior approval further order of this the Court being obtained before any refinancing; and sections 11.3 and 36 CCAA, and under reserve of subparagraph (c):

⁶ This language is inserted to clearly allow for payments which may be authorized by the Court under a companion charging order or otherwise.

⁷ Reference should be made to section 33 of the CCAA.

⁵ The term "resiliate" should be included if there are leased premises in the Province of Quebec, but can otherwise be removed.

- (c) d. in accordance with its convey, transfer assign, lease, or in any other manner dispose of the Property, outside the ordinary course of business, dispose of redundant or nonmaterial assets not exceeding \$ in value in whole or in part, provided that the price in each case does not exceed \$ 50 000 or \$ 150 000 in the aggregate;
- (d) <u>terminate the employment of such of their employees or temporarily or</u> permanently lay off such of their employees as they deem appropriate and, to the extent any amounts in lieu of notice, termination or severance pay or other amounts in respect thereof are not paid in the ordinary course, make provision, on such terms as may be agreed upon between the Applicants, as applicable, and such employee, or failing such agreement, make provision to deal with, any consequences thereof in the Plan, as the Applicants may determine;
- (e) <u>disclaim or resiliate agreements, subject to the provisions of section 32 CCAA</u> which are as follows:

32 (1) Subject to subsections (2) and (3), a debtor company may — on notice given in the prescribed form and manner to the other parties to the agreement and the monitor — disclaim or resiliate any agreement to which the company is a party on the day on which proceedings commence under this Act. The company may not give notice unless the monitor approves the proposed disclaimer or resiliation.

(2) Within 15 days after the day on which the company gives notice under subsection (1), a party to the agreement may, on notice to the other parties to the agreement and the monitor, apply to a court for an order that the agreement is not to be disclaimed or resiliated.

(3) If the monitor does not approve the proposed disclaimer or resiliation, the company may, on notice to the other parties to the agreement and the monitor, apply to a court for an order that the agreement be disclaimed or resiliated.

(4) In deciding whether to make the order, the court is to consider, among other things,

(a) whether the monitor approved the proposed disclaimer or resiliation;

(b) whether the disclaimer or resiliation would enhance the prospects of a viable compromise or arrangement being made in respect of the company; and

(c) whether the disclaimer or resiliation would likely cause significant financial hardship to a party to the agreement.

(5) An agreement is disclaimed or resiliated

(a) if no application is made under subsection (2), on the day that is 30 days after the day on which the company gives notice under subsection (1);

(b) if the court dismisses the application made under subsection (2), on the day that is 30 days after the day on which the company gives notice under subsection (1) or on any later day fixed by the court; or

(c) if the court orders that the agreement is disclaimed or resiliated under subsection (3), on the day that is 30 days after the day on which the company gives notice or on any later day fixed by the court.

(6) If the company has granted a right to use intellectual property to a party to an agreement, the disclaimer or resiliation does not affect the party's right to use the intellectual property — including the party's right to enforce an exclusive use — during the term of the agreement, including any period for which the party extends the agreement as of right, as long as the party continues to perform its obligations under the agreement in relation to the use of the intellectual property.

(7) If an agreement is disclaimed or resiliated, a party to the agreement who suffers a loss in relation to the disclaimer or resiliation is considered to have a provable claim.

(8) A company shall, on request by a party to the agreement, provide in writing the reasons for the proposed disclaimer or resiliation within five days after the day on which the party requests them.

(9) This section does not apply in respect of

(a) an eligible financial contract;

(b) a collective agreement;

(c) a financing agreement if the company is the borrower; or

(d) a lease of real property or of an immovable if the company is the lessor.

<u>and</u>

(f) <u>subject to section 11.3 CCAA, assign any rights and obligations of the</u> <u>Applicants.</u>

No Proceedings Against the Applicant or the Property

11. Until and including the day of July 17, 20 [no more than 30 days] 2023, or such 16. later date as this Court may order (the ""Stay Period""), no claim, grievance, application, action, suit, right or remedy, or proceeding or enforcement process in any court, tribunal, or arbitration association (each, a ""Proceeding") including but not limited to seizures, right to distrain, executions, writs of seizure or execution, any and all actions, applications, arbitration proceedings and other lawsuits existing at the time of this Order in which any of the Applicants is a defendant, party or respondent (either individually or with other Persons (as defined hereinafter)) shall be commenced, continued, or enforced against or in respect of any of the ApplicantApplicants or the Monitor, or affecting the Business or the Property, except with the written consent of the Applicant Applicants and the Monitor, or with leave of this Court, and any and all Proceedings currently under way against or in respect of the ApplicantApplicants or affecting the Business or the Property are hereby stayed and suspended pending further order of this Court. The entities related to or affiliated with the Monitor shall also be entitled to the protection, benefits and privileges afforded to the Monitor pursuant to this paragraph.

17. <u>The rights of His Majesty in right of Canada and His Majesty in right of a Province are</u> suspended in accordance with the terms and conditions of section 11.09 CCAA.

No Exercise of Rights or Remedies

18. 12. During the Stay Period, all rights and remedies, including, but not limited to modifications of existing rights and events deemed to occur pursuant to any agreement to which any of the Applicants is a party as a result of the insolvency of the Applicants and/or these CCAA proceedings, any events of default or non-performance by any of the Applicants as a result of the insolvency of the Applicants and/or these CCAA proceedings or any admissions or evidence in these CCAA proceedings, of any individual, firm, corporation, governmental body or agency, or any other entities (all of the foregoing, collectively being ""Persons" and each being a ""Person") against or in respect of the ApplicantApplicants or the Monitor, or affecting the Business or the Property, are hereby stayed and suspended except with the written consent of the Applicant Applicants and the Monitor, or leave of this Court, provided that nothing in this Order shall (i) empower the Applicant to carry on any business which the Applicant isApplicants are not lawfully entitled to carry on; (ii) affect such investigations, actions, suits or proceedings by a regulatory body as are permitted by section 11.1 of the CCAA; (iii) exempt the ApplicantApplicants from compliance with statutory or regulatory provisions relating to health, safety, or the environment; (iv) prevent the filing of any registration to preserve or perfect a security interest; or (v) prevent the registration of a claim for lien and the related filing of an action to preserve the right of a lien holder, provided that the Applicant Applicants shall not be required to file a defence during the stay period.

No Interference with Rights

19. 13. During the Stay Period, no Person shall discontinue, <u>fail to renew per the same</u> terms and conditions, fail to honour, alter, interfere with, repudiate, terminate, or cease to perform any right, renewal right, contract, agreement, licence, or permit in favour of or held by the <u>ApplicantApplicants</u>, including but not limited to renewal rights in respect of existing insurance policies on the same terms, except with the written consent of the <u>ApplicantApplicants</u> and the Monitor, or leave of this Court.

Exemption from Certain Reporting Obligations

20. Any and all continuous disclosure, reporting and filing obligations of, and audit committee requirements applicable to IMV Inc. as a result of its status as a reporting issuer in each of the provinces and territories of Canada subject to Canadian securities laws, rules, regulations and policy statements is hereby suspended until further order of this Court.

Shareholders' Meeting

21. The time limit to call and hold the Applicants' annual shareholders' meeting is extended until after the conclusion of the CCAA Proceedings, subject to further order of this Court.

Continuation of Services

22. <u>14.</u> During the Stay Period, all Persons having oral or written agreements with the <u>ApplicantApplicants</u> or statutory or regulatory mandates for the supply of goods or services, including without limitation all computer software, communication and other data services, centralized banking services, payroll services, insurance, transportation services, utility, or other services to the Business or the <u>ApplicantApplicants</u>, are hereby restrained until further order of this Court from discontinuing, <u>failing to renew per the same terms and conditions</u>, altering, interfering with, or terminating the supply of such goods or services as may be required by the <u>ApplicantApplicants</u>, and the <u>ApplicantApplicants</u> shall be entitled to the continued use of <u>itstheir</u> current premises, telephone numbers, facsimile numbers, internet addresses, and domain names, provided in each case that the normal prices or charges for all such goods or services received after the date of this Order are paid by the <u>ApplicantApplicants</u>, without having to provide any security deposit or any other security, in accordance with normal payment practices of the <u>ApplicantApplicants</u> or such other practices as may be agreed upon by the supplier or service provider and each of the <u>ApplicantApplicants</u> and the Monitor, or as may be ordered by this Court.

Non-Derogation of Rights

23. 15. Notwithstanding anything else contained herein, no Person shall be prohibited from requiring immediate payment for goods, services, use of leased or licensed property, or other valuable consideration provided on or after the date of this Order, nor shall any Person be under any obligation on or after the date of this Order to advance or re-advance any monies or otherwise extend any credit to the Applicant.⁸ Applicants.

Proceedings Against Directors and Officers

24. 16. During the Stay Period, and except as permitted by subsection 11.03(2) of the CCAA, no Proceeding may be commenced or continued against any of the former, current, or future, *de jure* or *de facto*, directors or officers of the ApplicantApplicants, nor against any person deemed to be a director or an officer of any of the Applicants under subsection 11.03(3) of the CCAA (the "Directors and Officers"), with respect to any claim against the directors or officers that arose beforeprior to the date hereofEffective Time and that relates to any obligations of the ApplicantApplicants whereby the directors or officers are alleged under any law to be liable in their capacity as directors or officers for the payment or performance of such obligations, until a compromise or arrangement in respect of the ApplicantApplicants, if one is filed, is sanctioned by this Court or is refused by the creditors of the ApplicantApplicants or this Court, these proceedings are dismissed by final order of this Court, or with leave of this Court.

Appointment of Monitor

- 25. <u>17. FTI Consulting Canada Inc.</u> is hereby appointed pursuant to the CCAA as the Monitor, an officer of this Court, to monitor the Business and financial affairs of the <u>ApplicantApplicants</u>, the Property, and the
- ApplicantApplicants's conduct of the Business with the powers and obligations set out in the CCAA or set forth herein and the ApplicantApplicants and itstheir shareholders, officers, directors, employees and Assistants shall advise the Monitor of all material steps taken

⁸ The Order must conform with the provisions of the CCAA. Particular attention should be paid when drafting the Order as a number of actions or steps cannot be stayed and the stay is subject to certain limits and restrictions under the CCAA. See, for example, CCAA sections 11.01, 11.04, 11.06, 11.07, and 11.08, and subsections 11.1(2) and 11.5(1).

by the <u>ApplicantApplicants</u> pursuant to this Order, cooperate fully with the Monitor in the exercise of its powers and discharge of its obligations, and provide the Monitor with the assistance that is necessary to enable the Monitor to adequately carry out the Monitor's functions.

- 26. 18. The Monitor, in addition to its prescribed rights and obligations under the CCAA, is hereby directed and empowered to:
 - (a) **a.** monitor the <u>Applicant's Applicants'</u> receipts and disbursements;

b.____

- (b) report to this<u>the</u> Court at such times and intervals as the Monitor may deem appropriate with respect to matters relating to the Property, the Business, the activities<u>on the state</u> of the
- Applicant, business and such other matters as may be relevant to the proceedings hereinfinancial affairs of the Applicants or developments in these proceedings or any related proceedings within the time limits set forth in the CCAA and at such time as considered appropriate by the Monitor or as the Court may order and may file consolidated reports for the Applicants;
- (c) e. advise the <u>ApplicantApplicants</u> in <u>itstheir</u> development of the Plan and any amendments to the Plan, and, to the extent deemed appropriate by the Monitor, assist in <u>itstheir</u> negotiations with creditors, customers, vendors, and other interested Persons;
- (d) d. assist the <u>ApplicantApplicants</u>, to the extent deemed appropriate by the Monitor, with the holding and administering of creditors' or shareholders' meetings for voting on the Plan;
- (e) retain and employ such agents, advisers and other assistants as are reasonably necessary for the purpose of carrying out the terms of this Order, including, without limitation, one or more entities related to or affiliated with the Monitor;
- (f) e. have full and complete access to the Property, including the premises, books, records, data, including data in electronic form, and other financial documents and to the Business of the <u>ApplicantApplicants</u>, to the extent that is necessary to adequately assess the Applicant's Business and financial affairs or to perform its duties arising under this Order;
- (g) f.-be at liberty to engage independent legal counsel or such other persons as the Monitor deems necessary or advisable respecting the exercise of its powers and performance of its obligations under this Order, including any affiliate of, or person related to the Monitor;
- (h) g. develop a claims process to ascertain the quantum of the claims of all creditors; and
- (i) h. be at liberty to perform such other duties as are required by this Order or by this Court from time to time.

- 27. The Monitor shall, as soon as practicable, (i) publish once a week for two (2) consecutive weeks or as otherwise directed by the Court, in La Presse+, the Globe & Mail National Edition and the Wall Street Journal and (ii) within five (5) business days after the date of this Order (A) post on the Monitor's website (the "Website") a notice containing the information prescribed under the CCAA, (B) make this Order publicly available in the manner prescribed under the CCAA, (C) send, in the prescribed manner, a notice to all known creditors having a claim against the Debtors of more than \$1,000, advising them that this Order is publicly available, and (D) prepare a list showing the names and addresses of such creditors and the estimated amounts of their respective claims, and make it publicly available in the prescribed manner, all in accordance with Section 23(1)(a) CCAA and the regulations made thereunder;
- 28. 19. The Monitor shall not take possession of the Property and shall take no part whatsoever in the management or supervision of the management of the Business and shall not, by fulfilling its obligations hereunder, be deemed to have taken or maintained possession or control of the Business or Property, or any part thereof.
- 29. 20. Nothing herein contained shall limit the protections afforded the Monitor at law including those protections set out in the CCAA.
- 30. 21. The Monitor shallmay provide any creditor of the Applicant or a potential Debtor In Possession lender ("DIP Lender"), with copy to counsel to the Applicants, creditors and other relevant stakeholders of the Applicants with information provided by the Applicant in response to reasonable requests for information made in writing by such creditor or a DIP Lenderrelevant stakeholder addressed to the Monitor. The Monitor shall not have any responsibility or liability with respect to the information disseminated by it pursuant to this paragraph. In the case of information that the Monitor has been advised by the ApplicantApplicants is confidential, the Monitor shall not provide such information to creditors or a DIP Lenderother relevant stakeholders unless otherwise directed by this Court or on such terms as the Monitor and the ApplicantApplicants may agree.
- 31. 22.-The Monitor, counsel to the Monitor, and all counsel to the <u>ApplicantApplicants</u> shall be paid their reasonable fees and disbursements, in each case not to exceed their standard rates and charges, by the <u>ApplicantApplicants</u> as part of the costs of these proceedings. The <u>Applicant isApplicants are</u> hereby authorized and directed to pay the accounts of the Monitor, counsel for the Monitor (<u>Stikeman Elliott LLP</u>) and counsel for the <u>Applicant on a [monthly/semi-annual/other]</u> <u>Applicants (McCarthy Tétrault LLP, Stewart McKelvey Lawyers and Troutman Pepper Hamilton Sanders LLP) on a weekly basis and, in addition, the <u>Applicant is hereby Applicants are</u> authorized to pay to the <u>Monitor, counsel to the Monitor, and counsel to the Applicant, retainers in the amounts of \$, \$, and \$ respectively to be held by them as security for payment of their respectiveprovide each with a reasonable retainer in advance on account of such fees and disbursements-outstanding from time to time, if so requested.</u></u>
- 32. 23. The Monitor and its legal counsel shall pass their accounts from time to time before a judge of this court or a referee appointed by a judge.

[Administrative Charge⁹

- 33. <u>24.</u> The Monitor, <u>counsel to</u> the Monitor<u>'s counsel</u>, and the <u>ApplicantApplicants</u>'s counsel <u>(collectively, the</u>)
- <u>"Chargees"</u>) shall be entitled to the benefit of and are hereby granted a charge, <u>security and</u> <u>hypothec</u> (the <u>"</u>Administration Charge<u>"</u>) on the Property, which charge shall not exceed an aggregate amount of <u>750,000</u>, as security for their professional fees and disbursements incurred at the standard rates and charges of the Monitor and such counsel, both before and after the making of this Order in respect of these proceedings. <u>The Administration Charge shall have the priority set out in paragraphs 39 and 41 herein.</u>

Directors' Charge

- 34. The Applicant shall indemnify their Directors and Officers against obligations and liabilities that they may incur as directors or officers of the Applicants after the commencement of the within proceedings, except to the extent that, with respect to any officer or director, the obligation or liability was incurred as a result of the Directors or Officers' gross negligence or wilful misconduct.
- 35. The Directors and Officers of the Applicants shall be entitled to the benefit of and are hereby granted a charge, security and hypothec (the "**Directors' Charge**") on the Property, which charge shall not exceed an aggregate amount of \$275,000, as security for the indemnity provided in paragraph 34 of this Order. The Directors' Charge shall have the priority set out in paragraphs 39 and 41 herein.
- 36. Notwithstanding any language in any applicable insurance policy to the contrary, (i) no insurer shall be entitled to be subrogated to or claim the benefit of the Directors' Charge, and (ii) the Directors and Officers shall only be entitled to the benefit of the Directors' Charge to the extent that they do not have coverage under any directors' and officers' insurance policy, or to the extent that such coverage is insufficient to pay amounts indemnified in accordance with paragraph 34 of this Order.

Key Employee Retention Plan

- 37. The Key Employee Retention Plan (the "KERP") attached as Schedule B to this Order (with its schedules filed under seal) is hereby approved and the Applicants are authorized to make peyments contemplated thereunder in accordance with the terms and conditions of the KERP with such minor alterations, changes, amendments, deletions or additions made thereto, with the consent of the Monitor, save and except with respect to the quantum of the KERP.
- 38. The employees referred to in the KERP (the "Key Employees") shall be entitled and are hereby granted a charge on the Property (the "KERP Charge") in the aggregate amount of \$575,000, to secure any payments to the Key Employees under the KERP. The KERP Charge shall have the priority set out in paragraphs 39 and 41 herein.

⁹ If a companion CCAA Charging Order is not granted, these provisions may be used to provide for an interim Administrative Charge.

Validity and Priority of the Charges Created by this Order

39. The priorities of the Administration Charge, the Directors' Charge and the KERP Charge, as among them, with respect to the Property to which they apply shall be as follows:

First – Administration Charge (to the maximum amount of \$750,000);

Second - Directors' Charge (to the maximum amount of \$275,000); and

Third – KERP Charge (to the maximum amount of \$575,000).

- 40. <u>25.</u> The filing, registration or perfection of the Administration Charge, <u>the Directors'</u> <u>Charge or the KERP Charge (collectively, the "Charges"</u>) shall not be required, and the <u>Administration ChargeCharges</u> shall be valid and enforceable for all purposes, including as against any right, title or interest filed, registered, recorded or perfected subsequent to the <u>Administration ChargeCharges</u> coming into existence, notwithstanding any such failure to file, register, record or perfect.
- 41. <u>26. The Each of the Administration Charge, the Directors' Charge and the KERP Charge</u> (all as constituted and defined herein) shall constitute a charge, security and hypothec on the Property and <u>such Charges</u> shall rank in priority to *claims of the following secured creditors:* [name secured creditors who have received notice as required by <u>CCAA section 11.52 and over whom priority is sought]</u> and in priority to anyall other hypothecs, mortgages, security interests, priorities, trusts (including deemed or <u>constructive trusts</u>), liens, charges, and encumbrances and claims of secured creditors, statutory or otherwise, of whatever nature or kind (collectively, the "Encumbrances") in favour of any Person.
- 27. The Applicant and the Chargees shall be entitled, upon giving notice to parties likely affected, to seek an order changing the amount of the Administration Charge or providing that the Administrative Charge shall rank in priority to secured creditors not named in paragraph 26.
- 42. 28. Except as otherwise expressly provided for herein, or as may be approved by this Court, the <u>ApplicantApplicants</u> shall not grant any <u>encumbranceEncumbrances</u> over any Property that <u>ranksrank</u> in priority to, or pari passu with, any of the Administration Charge, the Directors' Charge or the KERP Charge, unless the <u>Applicant alsoApplicants</u> obtains the prior written consent of the <u>Chargees</u>Monitor, the Secured Lenders and the <u>beneficiaries of the Administration Charge, the Directors' Charge</u>, the Directors' Charge or the KERP Charge, or further <u>orderOrder</u> of this Court.
- 43. 29. The <u>Administration Charge Charges</u> shall not be rendered invalid or unenforceable and the rights and remedies of the <u>chargees entitled to the benefit of the Charges</u> (collectively, the "Chargees") thereunder shall not otherwise be limited or impaired in any way by:- (ai) the pendency of these proceedings and the declarations of insolvency made herein; (bii) any application(s) for *a*-bankruptcy order(s) issued pursuant to *the* BIA, or any bankruptcy order made pursuant to such applications; (ciii) the filing of any

assignments for the general benefit of creditors made pursuant to the BIA; (iv) the provisions of any federal or provincial statutes; or ($d\underline{v}$) any negative covenants, prohibitions, or other similar provisions with respect to borrowings, incurring debt or the creation of *encumbrances* Encumbrances, contained in any existing loan documents, lease, sublease, offer to lease, or other agreement (collectively, an <u>"</u>"Agreement") which binds the <u>ApplicantApplicants</u>, and notwithstanding any provision to the contrary in any Agreement:

- (a) <u>neither</u> *a.* the creation of the <u>Administration Charge</u> Charges nor the execution, <u>delivery</u>, <u>perfection</u>, <u>registration or performance same</u> shall<u>not</u> create or be deemed to constitute a breach by the <u>ApplicantApplicants</u> of any Agreement to which it is a party;
- (b) b. none of the Chargees shall have any liability to any Person whatsoever as a result of any breach of any Agreement caused by or resulting from the <u>Applicant</u> seeking the creation of the <u>Administration ChargeCharges</u>; and
- (c) <u>e.</u> the payments made by the <u>ApplicantApplicants</u> pursuant to this Order <u>and the</u> <u>granting of the Charges</u>, do not and will not constitute preferences, fraudulent conveyances, transfers at undervalue, oppressive conduct, or other challengeable or voidable transactions under any applicable law.-]

Sale and Investment Solicitation Process

- 44. The sale and investment solicitation process ("**SISP**") procedures attached as **Schedule C** to this Order (the "**SISP Procedures**") are hereby approved, and the Monitor is hereby authorized to commence the SISP in accordance with the terms and conditions of the SISP Procedures. The Monitor and the Applicants are hereby authorized and directed to perfrom their respective obligations and to do all things reasonably necessary to perform their obligations under the SISP Procedures.
- 45. The Monitor and its affiliates, partners, directors, employees, agents and controlling persons shall have no liability with respect to any and all losses, claims, damages or liabilities, of any nature or kind, to any person in connection with or as a result of the conducting of the SISP, including any steps taken by the Monitor prior to the approval of the SISP and described in the First Report of the Monitor which are hereby approved nunc pro tunc, except to the extent such losses, claims, damages or liabilities result from the gross negligence or wilful misconduct of the Monitor in performing its obligations under the SISP Procedures, as determined by this Court.
- 46. Pursuant to clause 7(3)(c) of the Canada Personal Information Protection and Electronic Documents Act, the Applicants, the Monitor and their respective advisors are hereby authorized and permitted, subject to the execution of a non-disclosure agreement, to disclose and transfer to prospective SISP participants (each, a "SISP Participant") and their advisors personal information of identifiable individuals but only to the extent desirable or required to negotiate or attempt to complete a transaction pursuant to the SISP (a "Transaction"). Each SISP Participant to whom such personal information is disclosed shall maintain and protect the privacy of such information and limit the use of such information to its evaluation for the purpose of effecting a Transaction, and if it does not complete a Transaction, shall return all such information to the Applicants or the Monitor, or in the alternative destroy all such information and provide confirmation of its destruction if requested by the Applicants or the Monitor. Any Successful Party shall

maintain and protect the privacy of such information and, upon closing of the Transaction(s) contemplated in the Successful Bid(s), shall be entitled to use the personal information provided to it that is related to the Business and/or Property acquired pursuant to the SISP in a manner that is in all material respects identical to the prior use of such information by the Applicants, and shall return all other personal information to the Applicants or the Monitor, or ensure that all other personal information is destroyed and provide confirmation of its destruction if requested by the Applicants or the Monitor.

Service and Notice

30.— The Monitor shall: i) without delay, publish in a notice containing the information prescribed under the CCAA, ii) within five days after the date of this Order, (A) make this Order publicly available in the manner prescribed under the CCAA, (B) send, in the prescribed manner, a notice to every known creditor who has a claim against the Applicant of more than \$1,000, and (C) prepare a list showing the names and addresses of those creditors and the estimated amounts of those claims, and make it publicly available in the prescribed manner, all in accordance with section 23(1)(a) of the CCAA and the regulations made thereunder.

- 47. Subject to further Order of this Court, all applications in these CCAA proceedings are to be brought on not less than five (5) days' notice to all Persons on the service list. Each application shall specify a date (the "Initial Hearing Date") and time (the "Initial Hearing Time") for the hearing.
- 48. Any Person wishing to object to the relief sought on an application in these CCAA proceedings must serve a detailed written contestation stating the objection to the application and the grounds for such objection (a "**Contestation**") in writing to the moving party, the Applicants and the Monitor, with a copy to all Persons on the service list, no later than 5 p.m. Halifax time on the date that is three (3) days prior to the Initial Hearing Date (the "**Objection Deadline**").
- 49. If no Contestation is served by the Objection Deadline, the Judge having carriage of the application (the "Presiding Judge") may determine: (i) whether a hearing is necessary; (ii) whether such hearing will be in person, virtually, by telephone or by written submissions only; and (iii) the parties from whom submissions are required (collectively, the "Hearing Details"). In the absence of any such determination, a hearing will be held in the ordinary course.
- 50. If no Contestation is served by the Objection Deadline, the Applicants shall communicate with the Presiding Judge regarding whether a determination has been made by the Presiding Judge concerning the Hearing Details. The Applicants shall thereafter advise the service list of the Hearing Details and the Applicants shall report upon its dissemination of the Hearing Details to the Court in a timely manner.
- 51. If a Contestation is served by the Objection Deadline, the interested parties shall appear before the Presiding Judge on the Initial Hearing Date at the Initial Hearing Time, or such earlier or later time as may be directed by the Court, to, as the Court may direct: (i) proceed with the hearing on the Initial Hearing Date and at the Initial Hearing Time; or (ii) establish a schedule for the delivery of materials and the hearing of the contested application and such other matters, including interim relief, as the Court may direct.

- 52. Except as otherwise specified herein, the Applicants and the Monitor are at liberty to serve any notice, proof of claim form, proxy, circular or other document in connection with these CCAA proceedings by forwarding copies by prepaid ordinary mail, courier, personal delivery or electronic transmission to Persons or other appropriate parties at their respective given addresses as last shown on the records of the Applicants and that any such service shall be deemed to be received on the date of delivery if by personal delivery or electronic transmission, on the following business day if delivered by courier, or three business days after mailing if by ordinary mail.
- 53. 31. The ApplicantApplicants and the Monitor may give notice of this Order, any other materials and orders in these <u>CCAA</u> proceedings, and any notices, and provide correspondence, by forwarding <u>PDF copies</u> originals or true copies by prepaid ordinary mail, courier, personal delivery, or electronic transmission, including by email, to the <u>Applicant'sApplicants</u>' creditors or other interested parties at their respective addresses as last shown on the records of the <u>ApplicantApplicants</u> and any such notice by courier, personal delivery, or electronic transmission shall be deemed to be received on the next business day following the date of forwarding thereof, or if sent by ordinary mail, on the third business day after mailing.
- 54. 32. The <u>ApplicantApplicants</u> and the Monitor, and any party who has filed a demand of notice may serve any court materials in these proceedings by e-mailing a PDF or other electronic copy of such materials to counsel's e-mail addresses as recorded on the service list from time to time, and the Monitor may post a copy of any or all such materials on its website at http://cfcanada.fticonsulting.com/imv.com.
- 55. Unless otherwise provided herein, under the CCAA, or ordered by this Court, no document, order or other material need be served on any Person in respect of these CCAA proceedings, unless such Person has filed a demand of notice, or appears on the service list prepared by counsel for the Monitor, save and except when an order is sought against a Person not previously involved in these CCAA proceedings.

General

- 56. <u>33.</u> The <u>Applicant Applicants</u> or the Monitor may, from time to time, apply to this Court for <u>advice and</u> directions <u>inconcerning</u> the <u>dischargeexercise</u> of <u>itstheir respective</u> powers <u>and</u>, duties <u>and rights</u> hereunder <u>or in respect of the proper execution of this Order</u>.
- 57. 34. Nothing in this Order shall prevent the Monitor from acting as an interim receiver, a receiver, a receiver and manager, construction lien trustee, or a trustee in bankruptcy of the ApplicantApplicants, the Business or the Property.
- 58. This Order and all other orders in these proceedings shall have full force and effect in all provinces and territories in Canada.
- 59. IMV Inc. is authorized to apply as it may consider necessary or desirable, with or without notice, to any other court or administrative body, whether in Canada, the United States of America, or elsewhere, for orders which aid and complement this Order and any subsequent orders of this Court and, without limitation to the foregoing, any orders under Chapter 15 of the United States Bankruptcy Code, including an order for recognition of these CCAA proceedings as "Foreign Main Proceedings" in the United States of America pursuant to Chapter 15 of the United States Bankruptcy Code, and

for which IMV Inc. shall be the foreign representative of the Applicants (the "Foreign Representative"). All courts and administrative bodies of all such jurisdictions are hereby respectively requested to make such orders and to provide such assistance to the Applicants and the Foreign Representative as may be deemed necessary or appropriate for that purpose.

- 60. <u>35.</u> The aid and recognition of any <u>courtCourt</u>, tribunal, regulatory or administrative body having jurisdiction outside Nova Scotia, is requested in Canada, the United States of America or elsewhere, to give effect to this Order and to assist the <u>ApplicantApplicants</u>, the Monitor and their respective agents in carrying out the terms of this Order. All <u>courtsCourts</u>, tribunals, <u>and</u>-regulatory and administrative bodies are hereby respectfully requested to make such orders and to provide such assistance to the <u>ApplicantApplicants</u>, and to the Monitor, <u>as an officer of this Court</u>, as may be necessary or desirable to give effect to this Order, to grant representative status to the Monitor <u>or the authorized representative of the Applicants</u> in any foreign proceeding, or to assist the <u>ApplicantApplicants</u>, and the Monitor-<u>and their respective agents</u>, and to act in aid of and to be complementary to this Court, in carrying out the terms of this Order.
- 61. <u>For the purposes of any applications authorized by paragraphs 59 and 60, Applicants'</u> <u>"centre of main interest</u>" is located in the province of Nova Scotia, Canada.
- 62. <u>36.</u> Each of the <u>ApplicantApplicants</u> and the Monitor may apply to any court, tribunal, or regulatory or administrative body, wherever located, for the recognition of this Order and for assistance in carrying out the terms of this Order, and the Monitor may act as a representative in respect of the within proceedings for the purpose of having these proceedings recognized in a jurisdiction outside Canada.
- 63. <u>The schedules to the Key Employee Bonus Agreement (Schedule B to this Order) shall</u> be and remain sealed and kept confidential until further Order of this Court.
- 64. 37. Any interested party, including the Applicant and the Monitor, may apply to this Court to vary or amend this Order on such notice required under the *Civil Procedure Rules* or as this Court may order.

 38.—This Order and all of its provisions are effective as of
 [a.m./p.m.]
 [Atlantic

 Standard/Atlantic Daylight Saving]
 Time on the day of -, 20
 20

Issued , 20

lssued ,	
2023	

Prothonotary

Schedule "A" – Service List

Schedule "B" – KERP

Schedule "C" – SISP Procedures

Document comparison by Workshare Compare on jeudi, mai 04, 2023 10:29:47 Input:

-	
Document 1 ID	file://C:\Users\fatoupin\OneDrive - McCarthy Tétrault\Insolvabilité\IMV\NSSC_PM-8_Initial_Order_Form .doc
Description	NSSC_PM-8_Initial_Order_Form
Document 2 ID	iManage://mccarthytetrault-mobility-ca.imanage.work/mtdo cs/47684994/4
Description	#47684994v4 <mccarthytetrault-mobility-ca.imanage.work> - DRAFT - Amended and Restated Initial Order</mccarthytetrault-mobility-ca.imanage.work>
Rendering set	Standard

Legend:	
Insertion	
Deletion	
Moved from	
Moved to	
Style change	
Format change	
Moved deletion	
Inserted cell	
Deleted cell	
Moved cell	
Split/Merged cell	
Padding cell	

Statistics:	
	Count
Insertions	398
Deletions	296
Moved from	7
Moved to	7
Style changes	0
Format changes	0
Total changes	708

Schedule "E" Redline Draft Amended and Restated Initial Order to Initial Order

in

SUPREME COURT OF NOVA SCOTIA

IN THE MATTER OF: Application by IMV Inc. and Immunovaccine Technologies Inc. and IMV USA Inc. (the "Applicants"), for relief under the *Companies' Creditors Arrangement Act*

Amended and Restated Initial Order

Before the Honourable chambers:

The Applicants propose to make a compromise or arrangement under the *Companies' Creditors Arrangement Act*, R.S.C. 1985, c. C-36 as amended (the "**CCAA**") and they applied for an initial order and, now or in the future, other relief under the CCAA as may be sought on notice of motion.

The following parties received notice of this application: see attached at Schedule "A".

The following parties, represented by the following counsel, made submissions:

Party

<u>Counsel</u>

Applicant <u>Applicants</u>	McCarthy Tétrault LLP Alain N. Tardif François Alexandre Toupin
	Stewart McKelvey Lawyers Sara L. Scott
Proposed-Monitor, FTI Consulting Canada Inc.	Stikeman Elliott LLP Maria Konyukhova
<u>Horizon Technology Finance</u> Corporation, as agent	Aird & Berlis LLP Miranda Spence Kyle Plunkett

On motion of the Applicants, the following is ordered and declared:

Service

1. The service of the Notice of Application in ChambersMotion, the associated pleadings, and the supporting documents, as set out in the affidavit of service is hereby deemed adequate noticeabridged and validated so that the motionthis Motion is properly returnable today and further service thereof is hereby dispensed with.

Application

2. The Applicants are affiliated debtor companies within the meaning of the CCAA and are companies to which the CCAA applies.

Effective Time

3. This Order and all of its provisions are effective as of 12:01 a.m. Halifax time, province of Nova Scotia, on May 1, 2023 (the "**Effective Time**").

Plan of Arrangement

4. The Applicants, in consultation with the Monitor, shall have the authority to file and may, subject to further order of this Court, file with this Court a plan of compromise or arrangement (the "**Plan**").

Administrative Consolidation

- 5. The CCAA proceedings of the Applicants shall be consolidated under one single Court file, in file Hfx No. 523334.
- 6. All proceedings, filings, and other matters in the CCAA proceedings shall be filed jointly and together by the Applicants under file Hfx No. 523334.
- 7. The consolidation of these CCAA proceedings in respect of the Applicants shall be for administrative purposes only and shall not effect a consolidation of the assets and property or of the debts and obligations of each of the Applicants including, without limitation, for the purposes of any Plan or Plans that may be hereafter proposed.

Possession of Property and Operations

- 8. The Applicants shall remain in possession and control of their current and future assets, undertakings, and properties of every nature and kind whatsoever, and wherever situated including all proceeds thereof (the "**Property**"). Subject to further order of this Court, the Applicants shall continue to carry on business in a manner consistent with the preservation of their business (the "**Business**") and Property. The Applicants shall be authorized and empowered to continue to retain and employ consultants, agents, experts, accountants, counsel, and such other persons (collectively "**Assistants**") and the employees currently retained or employed by it, with liberty to retain such further Assistants as it deems reasonably necessary or desirable in the ordinary course of business or for the carrying out of the terms of this Order.
- 9. The Applicants shall be entitled to continue to utilize the central cash management system currently in place as described in the Affidavit of Andrew Hall dated April 28, 2023, or replace it with another substantially similar central cash management system (the "**Cash Management System**") and that any present or future bank providing the Cash Management System shall not be under any obligation whatsoever to inquire into the propriety, validity or legality of any transfer, payment, collection or other action taken under the Cash Management System, or as to the use or application by the Applicant of funds transferred, paid, collected or otherwise dealt with in the Cash Management System, shall be entitled to provide the Cash Management System without any liability in respect thereof to any Person (as defined hereinafter) other than the Applicants, pursuant to the terms of the documentation applicable to the Cash Management System, an unaffected creditor under the Plan with regard to any claims or expenses it may suffer or incur in connection with the provision of the Cash Management System.
- 10. The Applicants may pay the following expenses whether incurred prior to or after this Order

- (a) all outstanding and future wages, salaries, employee benefits, vacation pay, severance and expenses payable to employees on or after the date of this Order, in each case incurred in the ordinary course of business and consistent with existing compensation policies and arrangements; and
- (b) all existing and future employee health, dental, life insurance, short and long term disability and related benefits (collectively, the "**Group Benefits**") payable on or after the date of this Order to employees on or after the date of this Order, in each case incurred in the ordinary course of business and consistent with existing policies and arrangements or such amended policies and arrangements as are necessary or desirable to deliver the existing Group Benefits;
- (c) with prior written approval of the Monitor, the fees and disbursements for any Assistants retained or employed by the Applicant in respect of these proceedings, at their reasonable standard rates and charges; and
- (d) with the consent of the Monitor, amounts owing for goods or services actually provided to the Applicants prior to the Effective Time by third parties, if, in the opinion of IMV, such third party is critical to the Business and ongoing operations of IMV and such third party would sustain material prejudice if such payment is not made, up to a maximum aggregate amount of \$350,000.
- 11. Except as otherwise provided to the contrary herein, the Applicants may pay all reasonable expenses incurred by the Applicants in carrying on the Business in the ordinary course after this Order, and in carrying out the provisions of this Order, which expenses shall include, without limitation:
 - (a) all expenses and capital expenditures reasonably necessary for the preservation of the Property or the Business including, without limitation, payments on account of insurance including directors and officers insurance, maintenance, and security services; and
 - (b) payment for goods or services actually supplied to the Applicants following the date of this Order.
- 12. The Applicants shall remit or pay, in accordance with legal requirements or on terms as may be agreed to between the Applicants and the applicable authority:
 - (a) any statutory deemed trust amounts in favour of the Crown in right of Canada or of any Province thereof or any other taxation authority which are required to be deducted from employees' wages, including, without limitation, amounts in respect of: i) employment insurance, ii) Canada Pension Plan, iii) Quebec Pension Plan, and iv) income taxes;
 - (b) all goods and services or other applicable sales taxes (collectively, "Sales Taxes") required to be remitted by the Applicants in connection with the sale of goods and services by the Applicants, but only where such Sales Taxes are accrued or collected after the date of this Order; and
 - (c) any amount payable to the Crown in right of Canada or of any Province or any regulatory or administrative body or any other authority, in all cases in respect of municipal realty, municipal business, or other taxes, assessments or levies of any nature or kind which are: (i) entitled at law to be paid in priority to claims of

secured creditors; (ii) attributable to or in respect of the ongoing Business carried on by the Applicants; and (iii) payable in respect of the period commencing on or after the date of this Order.

- 13. Until such time as the Applicants disclaims or resiliate a real property lease in accordance with the CCAA, the Applicants shall pay all amounts constituting rent or payable as rent under real property leases, including, for greater certainty, common area maintenance charges, utilities and realty taxes, and any other amounts payable to the landlord under the lease, or as otherwise may be negotiated between the Applicants and the landlord from time to time ("**Rent**"), for the period commencing from and including the date of this Order, in accordance with its existing lease agreements. On the date of the first of such payments, any arrears relating to the period commencing from and including the date of this Order shall also be paid.
- 14. Except as specifically permitted herein or by further order of this Court, the Applicants are hereby directed, until further order of this Court: (i) to make no payments of principal, interest thereon or otherwise on account of amounts owing by the Applicants to any of their creditors as of this date without prior written consent of the Monitor; (ii) to grant no security interests, trusts, liens, charges, or encumbrances upon or in respect of any of its Property; and (iii) to not grant credit or incur liabilities except in the ordinary course of the Business or with the prior written approval of the Monitor.

Restructuring

- 15. The Applicants shall, subject to such requirements as are imposed by the Monitor and under any agreements for debtor in possession financing which may be granted, have the right to:
 - (a) permanently or temporarily cease, downsize or shut down any of their operations or locations as they deem appropriate and make provision for the consequences thereof in the Plan;
 - (b) pursue all avenues to finance or refinance, market, convey, transfer, assign or in any other manner dispose of the Business or Property, in whole or part, subject to further order of the Court and sections 11.3 and 36 CCAA, and under reserve of subparagraph (c);
 - (c) convey, transfer assign, lease, or in any other manner dispose of the Property, outside the ordinary course of business, in whole or in part, provided that the price in each case does not exceed \$ 50 000 or \$ 150 000 in the aggregate;
 - (d) terminate the employment of such of their employees or temporarily or permanently lay off such of their employees as they deem appropriate and, to the extent any amounts in lieu of notice, termination or severance pay or other amounts in respect thereof are not paid in the ordinary course, make provision, on such terms as may be agreed upon between the Applicants, as applicable, and such employee, or failing such agreement, make provision to deal with, any consequences thereof in the Plan, as the Applicants may determine;
 - (e) disclaim or resiliate agreements, subject to the provisions of section 32 CCAA which are as follows:

32 (1) Subject to subsections (2) and (3), a debtor company may — on notice given in the prescribed form and manner to the other parties to the agreement and the monitor — disclaim or resiliate any agreement to which the company is a party on the day on which proceedings commence under this Act. The company may not give notice unless the monitor approves the proposed disclaimer or resiliation.

(2) Within 15 days after the day on which the company gives notice under subsection (1), a party to the agreement may, on notice to the other parties to the agreement and the monitor, apply to a court for an order that the agreement is not to be disclaimed or resiliated.

(3) If the monitor does not approve the proposed disclaimer or resiliation, the company may, on notice to the other parties to the agreement and the monitor, apply to a court for an order that the agreement be disclaimed or resiliated.

(4) In deciding whether to make the order, the court is to consider, among other things,

(a) whether the monitor approved the proposed disclaimer or resiliation;

(b) whether the disclaimer or resiliation would enhance the prospects of a viable compromise or arrangement being made in respect of the company; and

(c) whether the disclaimer or resiliation would likely cause significant financial hardship to a party to the agreement.

(5) An agreement is disclaimed or resiliated

(a) if no application is made under subsection (2), on the day that is 30 days after the day on which the company gives notice under subsection (1);

(b) if the court dismisses the application made under subsection (2), on the day that is 30 days after the day on which the company gives notice under subsection (1) or on any later day fixed by the court; or

(c) if the court orders that the agreement is disclaimed or resiliated under subsection (3), on the day that is 30 days after the day on which the company gives notice or on any later day fixed by the court.

(6) If the company has granted a right to use intellectual property to a party to an agreement, the disclaimer or resiliation does not affect the party's right to use the intellectual property — including the party's right to enforce an exclusive use — during the term of the agreement, including any period for which the party extends the agreement as of right, as long as the party continues to perform its obligations under the agreement in relation to the use of the intellectual property.

(7) If an agreement is disclaimed or resiliated, a party to the agreement who suffers a loss in relation to the disclaimer or resiliation is considered to have a provable claim.

(8) A company shall, on request by a party to the agreement, provide in writing the reasons for the proposed disclaimer or resiliation within five days after the day on which the party requests them.

- (9) This section does not apply in respect of
 - (a) an eligible financial contract;
 - (b) a collective agreement;
 - (c) a financing agreement if the company is the borrower; or
 - (d) a lease of real property or of an immovable if the company is the lessor.

and

(f) subject to section 11.3 CCAA, assign any rights and obligations of the Applicants.

No Proceedings Against the Applicant or the Property

- 16. Until and including the day of May 5 July 17, 2023, or such later date as this Court may order (the "Stay Period"), no claim, grievance, application, action, suit, right or remedy, or proceeding or enforcement process in any court, tribunal, or arbitration association (each, a "Proceeding") including but not limited to seizures, right to distrain, executions, writs of seizure or execution, any and all actions, applications, arbitration proceedings and other lawsuits existing at the time of this Order in which any of the Applicants is a defendant, party or respondent (either individually or with other Persons (as defined hereinafter)) shall be commenced, continued, or enforced against or in respect of any of the Applicants or the Monitor, or affecting the Business or the Property, except with the written consent of the Applicants and the Monitor, or with leave of this Court, and any and all Proceedings currently under way against or in respect of the Applicants or affecting the Business or the Property are hereby stayed and suspended pending further order of this Court. The entities related to or affiliated with the Monitor shall also be entitled to the protection, benefits and privileges afforded to the Monitor pursuant to this paragraph.
- 17. The rights of His Majesty in right of Canada and His Majesty in right of a Province are suspended in accordance with the terms and conditions of section 11.09 CCAA.

No Exercise of Rights or Remedies

18. During the Stay Period all rights and remedies, including, but not limited to modifications of existing rights and events deemed to occur pursuant to any agreement to which any of the Applicants is a party as a result of the insolvency of the Applicants and/or these CCAA proceedings, any events of default or non-performance by any of the Applicants as a result of the insolvency of the Applicants and/or these CCAA proceedings or any admissions or evidence in these CCAA proceedings, of any individual, firm, corporation, governmental body or agency, or any other entities (all of the foregoing, collectively being "Persons" and each being a "Person") against or in respect of the Applicants or the Monitor, or affecting the Business or the Property, are hereby stayed and suspended except with the written consent of the Applicants and the Monitor, or leave of this Court, provided that nothing in this Order shall (i) empower the Applicant to carry on any business which the Applicants are not lawfully entitled to carry on; (ii) affect such investigations, actions, suits or proceedings by a regulatory body as are permitted by section 11.1 of the CCAA; (iii) exempt the Applicants from compliance with statutory or regulatory provisions relating to health, safety, or the environment; (iv) prevent the filing of any registration to preserve or perfect a security interest; or (v) prevent the

registration of a claim for lien and the related filing of an action to preserve the right of a lien holder, provided that the Applicants shall not be required to file a defence during the stay period.

No Interference with Rights

19. During the Stay Period, no Person shall discontinue, fail to renew per the same terms and conditions, fail to honour, alter, interfere with, repudiate, terminate, or cease to perform any right, renewal right, contract, agreement, licence, or permit in favour of or held by the Applicants, including but not limited to renewal rights in respect of existing insurance policies on the same terms, except with the written consent of the Applicants and the Monitor, or leave of this Court.

Exemption from Certain Reporting Obligations

20. Any and all continuous disclosure, reporting and filing obligations of, and audit committee requirements applicable to IMV Inc. as a result of its status as a reporting issuer in each of the provinces and territories of Canada subject to Canadian securities laws, rules, regulations and policy statements is hereby suspended until further order of this Court.

Shareholders' Meeting

21. The time limit to call and hold the Applicants' annual shareholders' meeting is extended until after the conclusion of the CCAA Proceedings, subject to further order of this Court.

Continuation of Services

22 During the Stay Period, all Persons having oral or written agreements with the Applicants or statutory or regulatory mandates for the supply of goods or services, including without limitation all computer software, communication and other data services, centralized banking services, payroll services, insurance, transportation services, utility, or other services to the Business or the Applicants, are hereby restrained until further order of this Court from discontinuing, failing to renew per the same terms and conditions, altering, interfering with, or terminating the supply of such goods or services as may be required by the Applicants, and the Applicants shall be entitled to the continued use of their current premises, telephone numbers, facsimile numbers, internet addresses, and domain names, provided in each case that the normal prices or charges for all such goods or services received after the date of this Order are paid by the Applicants, without having to provide any security deposit or any other security, in accordance with normal payment practices of the Applicants or such other practices as may be agreed upon by the supplier or service provider and each of the Applicants and the Monitor, or as may be ordered by this Court.

Non-Derogation of Rights

23. Notwithstanding anything else contained herein, no Person shall be prohibited from requiring immediate payment for goods, services, use of leased or licensed property, or other valuable consideration provided on or after the date of this Order, nor shall any Person be under any obligation on or after the date of this Order to advance or re-advance any monies or otherwise extend any credit to the Applicants.

Proceedings Against Directors and Officers

24. During the Stay Period, and except as permitted by subsection 11.03(2) of the CCAA, no Proceeding may be commenced or continued against any of the former, current, or future, *de jure* or *de facto*, directors or officers of the Applicants, nor against any person deemed to be a director or an officer of any of the Applicants under subsection 11.03(3) of the CCAA (the "**Directors and Officers**"), with respect to any claim against the directors or officers that arose prior to the Effective Time and that relates to any obligations of the Applicants whereby the directors or officers are alleged under any law to be liable in their capacity as directors or officers for the payment or performance of such obligations, until a compromise or arrangement in respect of the Applicants, if one is filed, is sanctioned by this Court or is refused by the creditors of the Applicants or this Court, these proceedings are dismissed by final order of this Court, or with leave of this Court.

Appointment of Monitor

- 25. FTI Consulting Canada Inc. is hereby appointed pursuant to the CCAA as the Monitor, an officer of this Court, to monitor the Business and financial affairs of the Applicants, the Property, and the Applicants' conduct of the Business with the powers and obligations set out in the CCAA or set forth herein and the Applicants and their shareholders, officers, directors, employees and Assistants shall advise the Monitor of all material steps taken by the Applicants pursuant to this Order, cooperate fully with the Monitor in the exercise of its powers and discharge of its obligations, and provide the Monitor with the assistance that is necessary to enable the Monitor to adequately carry out the Monitor's functions.
- 26. The Monitor, in addition to its prescribed rights and obligations under the CCAA, is hereby directed and empowered to:
 - (a) monitor the Applicants' receipts and disbursements;
 - (b) report to the Court on the state of the business and financial affairs of the Applicants or developments in these proceedings or any related proceedings within the time limits set forth in the CCAA and at such time as considered appropriate by the Monitor or as the Court may order and may file consolidated reports for the Applicants;
 - (c) advise the Applicants in their development of the Plan and any amendments to the Plan, and, to the extent deemed appropriate by the Monitor, assist in their negotiations with creditors, customers, vendors, and other interested Persons;
 - (d) assist the Applicants, to the extent deemed appropriate by the Monitor, with the holding and administering of creditors' or shareholders' meetings for voting on the Plan;
 - (e) retain and employ such agents, advisers and other assistants as are reasonably necessary for the purpose of carrying out the terms of this Order, including, without limitation, one or more entities related to or affiliated with the Monitor;
 - (f) have full and complete access to the Property, including the premises, books, records, data, including data in electronic form, and other financial documents and to the Business of the Applicants, to the extent that is necessary to

adequately assess the Applicant's Business and financial affairs or to perform its duties arising under this Order;

- (g) be at liberty to engage independent legal counsel or such other persons as the Monitor deems necessary or advisable respecting the exercise of its powers and performance of its obligations under this Order, including any affiliate of, or person related to the Monitor;
- (h) develop a claims process to ascertain the quantum of the claims of all creditors; and
- (i) be at liberty to perform such other duties as are required by this Order or by this Court from time to time.
- 27. The Monitor shall, as soon as practicable, (i) publish once a week for two (2) consecutive weeks or as otherwise directed by the Court, in La Presse+, the Globe & Mail National Edition and the Wall Street Journal and (ii) within five (5) business days after the date of this Order (A) post on the Monitor's website (the "**Website**") a notice containing the information prescribed under the CCAA, (B) make this Order publicly available in the manner prescribed under the CCAA, (C) send, in the prescribed manner, a notice to all known creditors having a claim against the Debtors of more than \$1,000, advising them that this Order is publicly available, and (D) prepare a list showing the names and addresses of such creditors and the estimated amounts of their respective claims, and make it publicly available in the prescribed manner, all in accordance with Section 23(1)(a) CCAA and the regulations made thereunder;
- 28. The Monitor shall not take possession of the Property and shall take no part whatsoever in the management or supervision of the management of the Business and shall not, by fulfilling its obligations hereunder, be deemed to have taken or maintained possession or control of the Business or Property, or any part thereof.
- 29. Nothing herein contained shall limit the protections afforded the Monitor at law including those protections set out in the CCAA.
- 30. The Monitor may provide, with copy to counsel to the Applicants, creditors and other relevant stakeholders of the Applicants with information in response to reasonable requests for information made in writing by such creditor or relevant stakeholder addressed to the Monitor. The Monitor shall not have any responsibility or liability with respect to the information disseminated by it pursuant to this paragraph. In the case of information that the Monitor has been advised by the Applicants is confidential, the Monitor shall not provide such information to creditors or other relevant stakeholders unless otherwise directed by this Court or on such terms as the Monitor and the Applicants may agree.
- 31. The Monitor, counsel to the Monitor, and all counsel to the Applicants shall be paid their reasonable fees and disbursements, in each case not to exceed their standard rates and charges, by the Applicants as part of the costs of these proceedings. The Applicants are hereby authorized and directed to pay the accounts of the Monitor, counsel for the Monitor (Stikeman Elliott LLP) and counsel for the Applicants (McCarthy Tétrault LLP, Stewart McKelvey Lawyers and Troutman Pepper Hamilton Sanders LLP) on a weekly basis and, in addition, the Applicants are authorized to provide each with a reasonable retainer in advance on account of such fees and disbursements, if so requested.

- 32. The Monitor and its legal counsel shall pass their accounts from time to time before a judge of this court or a referee appointed by a judge.
- 33. The Monitor, counsel to the Monitor and the Applicants' counsel shall be entitled to the benefit of and are hereby granted a charge, security and hypothec (the "Administration Charge") on the Property, which charge shall not exceed an aggregate amount of \$350,000750,000, as security for their professional fees and disbursements incurred at the standard rates and charges of the Monitor and such counsel, both before and after the making of this Order in respect of these proceedings. The Administration Charge shall have the priority set out in paragraphs 3739 and 3941 herein.

Directors' Charge

- 34. The Applicant shall indemnify their Directors and Officers against obligations and liabilities that they may incur as directors or officers of the Applicants after the commencement of the within proceedings, except to the extent that, with respect to any officer or director, the obligation or liability was incurred as a result of the Directors or Officers' gross negligence or wilful misconduct.
- 35. The Directors and Officers of the Applicants shall be entitled to the benefit of and are hereby granted a charge, security and hypothec (the "**Directors' Charge**") on the Property, which charge shall not exceed an aggregate amount of \$450,000275,000, as security for the indemnity provided in paragraph 34 of this Order. The Directors' Charge shall have the priority set out in paragraphs 3739 and 3941 herein.
- 36. Notwithstanding any language in any applicable insurance policy to the contrary, (i) no insurer shall be entitled to be subrogated to or claim the benefit of the Directors' Charge, and (ii) the Directors and Officers shall only be entitled to the benefit of the Directors' Charge to the extent that they do not have coverage under any directors' and officers' insurance policy, or to the extent that such coverage is insufficient to pay amounts indemnified in accordance with paragraph 34 of this Order.

Key Employee Retention Plan

- 37. The Key Employee Retention Plan (the "KERP") attached as Schedule B to this Order (with its schedules filed under seal) is hereby approved and the Applicants are authorized to make peyments contemplated thereunder in accordance with the terms and conditions of the KERP with such minor alterations, changes, amendments, deletions or additions made thereto, with the consent of the Monitor, save and except with respect to the quantum of the KERP.
- 38. The employees referred to in the KERP (the "Key Employees") shall be entitled and are hereby granted a charge on the Property (the "KERP Charge") in the aggregate amount of \$575,000, to secure any payments to the Key Employees under the KERP. The KERP Charge shall have the priority set out in paragraphs 39 and 41 herein.

Validity and Priority of the Charges Created by this Order

39. 37. The priorities of the Administration Charge and the <u>KERP</u> Charge, as among them, with respect to the Property to which they apply shall be as follows:

> First – Administration Charge (to the maximum amount of \$350,000,750,000). Second Effectors' Charge (to the maximum amount of \$450,000,275,000); and

Third – KERP Charge (to the maximum amount of \$575,000).

- 40. 38. The filing, registration or perfection of the <u>Administration Charge, the</u> Directors' Charge or the <u>AdministrationKERP</u> Charge (collectively, the "**Charges**") shall not be required, and the Charges shall be valid and enforceable for all purposes, including as against any right, title or interest filed, registered, recorded or perfected subsequent to the Charges coming into existence, notwithstanding any such failure to file, register, record or perfect.
- 41. 39. Each of the Administration Charge and, the Directors' Charge and the KERP Charge (all as constituted and defined herein) shall constitute a charge, security and hypothec on the Property.
- 40. The Applicants and the Chargees shall be entitled, upon giving notice to parties likely to be affected, to seek an order changing the amount of the Charges or providing that the<u>such</u> Charges shall rank in priority to secured creditors of the Applicants, including over the claims of the federal and provincial governments subject to a deemed trust.
- 41. Except as otherwise expressly provided for herein, or as may be approved by this Court, the Applicants shall not grant anyall other hypothecs, mortgages, security interests, priorities, trusts (including deemed or constructive trusts), liens, charges and encumbrances, claims of secured creditors, statutory or otherwise, of whatever nature or kind (collectively, the "Encumbrances") in favour of any Person.
- 42. Except as otherwise expressly provided for herein, or as may be approved by this Court, the Applicants shall not grant any Encumbrances over any Property that rank in priority to, or *pari passu* with, any of the Administration Charge-or, the Directors' <u>Charge or the KERP</u> Charge, unless the Applicants obtains the prior written consent of the Monitor, the <u>Secured Lenders</u> and the beneficiaries of the Administration Charge-and, the Directors' <u>Charge and the KERP</u> Charge, or further Order of this Court.
- **43**. **42**. The Charges shall not be rendered invalid or unenforceable and the rights and remedies of the chargees entitled to the benefit of the Charges (collectively, the "**Chargees**") thereunder shall not otherwise be limited or impaired in any way by (i) the pendency of these proceedings and the declarations of insolvency made herein; (ii) any application(s) for bankruptcy order(s) issued pursuant to BIA, or any bankruptcy order made pursuant to such applications; (iii) the filing of any assignments for the general benefit of creditors made pursuant to the BIA; (iv) the provisions of any federal or provincial statutes; or (v) any negative covenants, prohibitions or other similar provisions with respect to borrowings, incurring debt or the creation of Encumbrances, contained in any existing loan documents, lease, sublease, offer to lease or other agreement (collectively, an "**Agreement**") which binds the Applicants, and notwithstanding any provision to the contrary in any Agreement:

- (a) neither the creation of the Charges nor the execution, delivery, perfection, registration or performance same shall create or be deemed to constitute a breach by the Applicants of any Agreement to which it is a party;
- (b) none of the Chargees shall have any liability to any Person whatsoever as a result of any breach of any Agreement caused by or resulting from the creation of the Charges; and
- (c) the payments made by the Applicants pursuant to this Order and the granting of the Charges, do not and will not constitute preferences, fraudulent conveyances, transfers at undervalue, oppressive conduct, or other challengeable or voidable transactions under any applicable law.

Sale and Investment Solicitation Process

- 44. The sale and investment solicitation process ("**SISP**") procedures attached as **Schedule C** to this Order (the "**SISP Procedures**") are hereby approved, and the Monitor is hereby authorized to commence the SISP in accordance with the terms and conditions of the SISP Procedures. The Monitor and the Applicants are hereby authorized and directed to perfrom their respective obligations and to do all things reasonably necessary to perform their obligations under the SISP Procedures.
- 45. The Monitor and its affiliates, partners, directors, employees, agents and controlling persons shall have no liability with respect to any and all losses, claims, damages or liabilities, of any nature or kind, to any person in connection with or as a result of the conducting of the SISP, including any steps taken by the Monitor prior to the approval of the SISP and described in the First Report of the Monitor which are hereby approved *nunc pro tunc*, except to the extent such losses, claims, damages or liabilities result from the gross negligence or wilful misconduct of the Monitor in performing its obligations under the SISP Procedures, as determined by this Court.
- Pursuant to clause 7(3)(c) of the Canada Personal Information Protection and Electronic **46**. Documents Act, the Applicants, the Monitor and their respective advisors are hereby authorized and permitted, subject to the execution of a non-disclosure agreement, to disclose and transfer to prospective SISP participants (each, a "SISP Participant") and their advisors personal information of identifiable individuals but only to the extent desirable or required to negotiate or attempt to complete a transaction pursuant to the SISP (a "Transaction"). Each SISP Participant to whom such personal information is disclosed shall maintain and protect the privacy of such information and limit the use of such information to its evaluation for the purpose of effecting a Transaction, and if it does not complete a Transaction, shall return all such information to the Applicants or the Monitor, or in the alternative destroy all such information and provide confirmation of its destruction if requested by the Applicants or the Monitor. Any Successful Party shall maintain and protect the privacy of such information and, upon closing of the Transaction(s) contemplated in the Successful Bid(s), shall be entitled to use the personal information provided to it that is related to the Business and/or Property acquired pursuant to the SISP in a manner that is in all material respects identical to the prior use of such information by the Applicants, and shall return all other personal information to the Applicants or the Monitor, or ensure that all other personal information is destroyed and provide confirmation of its destruction if requested by the Applicants or the Monitor.

Service and Notice

- 47. Subject to further Order of this Court, all applications in these CCAA proceedings are to be brought on not less than five (5) days' notice to all Persons on the service list. Each application shall specify a date (the "Initial Hearing Date") and time (the "Initial Hearing Time") for the hearing.
- **48**. **44**. Any Person wishing to object to the relief sought on an application in these CCAA proceedings must serve a detailed written contestation stating the objection to the application and the grounds for such objection (a "**Contestation**") in writing to the moving party, the Applicants and the Monitor, with a copy to all Persons on the service list, no later than 5 p.m. Halifax time on the date that is three (3) days prior to the Initial Hearing Date (the "**Objection Deadline**").
- 49. 45. If no Contestation is served by the Objection Deadline, the Judge having carriage of the application (the "**Presiding Judge**") may determine: (i) whether a hearing is necessary; (ii) whether such hearing will be in person, virtually, by telephone or by written submissions only; and (iii) the parties from whom submissions are required (collectively, the "**Hearing Details**"). In the absence of any such determination, a hearing will be held in the ordinary course.
- 50. 46.—If no Contestation is served by the Objection Deadline, the Applicants shall communicate with the Presiding Judge regarding whether a determination has been made by the Presiding Judge concerning the Hearing Details. The Applicants shall thereafter advise the service list of the Hearing Details and the Applicants shall report upon its dissemination of the Hearing Details to the Court in a timely manner.
- 51. 47. If a Contestation is served by the Objection Deadline, the interested parties shall appear before the Presiding Judge on the Initial Hearing Date at the Initial Hearing Time, or such earlier or later time as may be directed by the Court, to, as the Court may direct: (i) proceed with the hearing on the Initial Hearing Date and at the Initial Hearing Time; or (ii) establish a schedule for the delivery of materials and the hearing of the contested application and such other matters, including interim relief, as the Court may direct.
- 52. 48. Except as otherwise specified herein, the Applicants and the Monitor are at liberty to serve any notice, proof of claim form, proxy, circular or other document in connection with these CCAA proceedings by forwarding copies by prepaid ordinary mail, courier, personal delivery or electronic transmission to Persons or other appropriate parties at their respective given addresses as last shown on the records of the Applicants and that any such service shall be deemed to be received on the date of delivery if by personal delivery or electronic transmission, on the following business day if delivered by courier, or three business days after mailing if by ordinary mail.
- 53. 49. The Applicants and the Monitor may give notice of this Order, any other materials and orders in these CCAA proceedings, and any notices, and provide correspondence, by forwarding PDF copies originals or true copies by prepaid ordinary mail, courier, personal delivery, or electronic transmission, including by email, to the Applicants' creditors or other interested parties at their respective addresses as last shown on the records of the Applicants and any such notice by courier, personal delivery, or electronic transmission shall be deemed to be received on the next business day following the date of forwarding thereof, or if sent by ordinary mail, on the third business day after mailing.

- 54. 50. The Applicants and the Monitor, and any party who has filed a demand of notice may serve any court materials in these proceedings by e-mailing a PDF or other electronic copy of such materials to counsel's e-mail addresses as recorded on the service list from time to time, and the Monitor may post a copy of any or all such materials on its website at http://cfcanada.fticonsulting.com/imv.
- 55. 51. Unless otherwise provided herein, under the CCAA, or ordered by this Court, no document, order or other material need be served on any Person in respect of these CCAA proceedings, unless such Person has filed a demand of notice, or appears on the service list prepared by counsel for the Monitor, save and except when an order is sought against a Person not previously involved in these CCAA proceedings.

Comeback Hearing

52. Notwithstanding paragraphs 43 to 47 of this Order, the Comeback Hearing shall be held at 2 p.m. (Halifax time) on May 5, 2023.

General

- 56. 53. The Applicants or the Monitor may, from time to time, apply to this Court for directions concerning the exercise of their respective powers, duties and rights hereunder or in respect of the proper execution of this Order.
- 57. 54. Nothing in this Order shall prevent the Monitor from acting as an interim receiver, a receiver, a receiver and manager, construction lien trustee, or a trustee in bankruptcy of the Applicants, the Business or the Property.
- 58. 55. This Order and all other orders in these proceedings shall have full force and effect in all provinces and territories in Canada.
- 59. 56.-IMV Inc. is authorized to apply as it may consider necessary or desirable, with or without notice, to any other court or administrative body, whether in Canada, the United States of America, or elsewhere, for orders which aid and complement this Order and any subsequent orders of this Court and, without limitation to the foregoing, any orders under Chapter 15 of the United States Bankruptcy Code, including an order for recognition of these CCAA proceedings as "Foreign Main Proceedings" in the United States of America pursuant to Chapter 15 of the United States Bankruptcy Code, and for which IMV Inc. shall be the foreign representative of the Applicants (the "Foreign Representative"). All courts and administrative bodies of all such jurisdictions are hereby respectively requested to make such orders and to provide such assistance to the Applicants and the Foreign Representative as may be deemed necessary or appropriate for that purpose.
- 60. 57. The aid and recognition of any Court, tribunal, regulatory or administrative body in Canada, the United States of America or elsewhere, to give effect to this Order and to assist the Applicants, the Monitor and their respective agents in carrying out the terms of this Order. All Courts, tribunals, regulatory and administrative bodies are hereby respectfully requested to make such orders and to provide such assistance to the Applicants, and the Monitor as may be necessary or desirable to give effect to this Order, to grant representative status to the Monitor or the authorized representative of the Applicants in any foreign proceeding, to assist the Applicants, and the Monitor, and to act in aid of and to be complementary to this Court, in carrying out the terms of this Order.

- 61. 58. For the purposes of any applications authorized by paragraphs 5659 and 5760, Applicants' "*centre of main interest*" is located in the province of Nova Scotia, Canada.
- 62. 59. Each of the Applicants and the Monitor may apply to any court, tribunal, or regulatory or administrative body, wherever located, for the recognition of this Order and for assistance in carrying out the terms of this Order, and the Monitor may act as a representative in respect of the within proceedings for the purpose of having these proceedings recognized in a jurisdiction outside Canada.
- 63. <u>The schedules to the Key Employee Bonus Agreement (Schedule B to this Order) shall</u> be and remain sealed and kept confidential until further Order of this Court.
- 64. 60. Any interested party, including the Applicant and the Monitor, may apply to this Court to vary or amend this Order on such notice required under the *Civil Procedure Rules* or as this Court may order.

,

Issued 2023

Prothonotary

PARTY	ADDRESS	METHOD OF SERVICE
I MV Inc. and Immunovaccine Technologies Inc.	c/o Alain N. Tardif and François Alexandre Toupin McCarthy Tétrault LLP 1000 De La Gauchetière Street West Suite MZ400 Montréal, QC H3B 0A2	atardif@mccarthy.ca fatoupin@mccarthy.ca sscott@stewartmckelvey.com
	Sara L. Scott Stewart McKelvey 600-1741 Lower Water Street Halifax, NS-B3J 0J2	
Monitor, FTI Canada Consulting Inc.	c/o Jeffrey Rosenberg Senior Managing Director TD South Tower 79 Wellington Street West Toronto Dominion Centre, Suite 2010 P.O. Box 104 Toronto, ON M5K 1G8	jeffrey.rosenberg@fticonsulting.com

Schedule "A" – Service List

T

Schedule "B" – KERP

Schedule "C" – SISP Procedures

Document comparison by Workshare Compare on jeudi, mai 04, 2023 10:28:04 Input:

•	
Document 1 ID	iManage://mccarthytetrault-mobility-ca.imanage.work/MTD OCS/47626273/4
Description	#47626273v4 <mccarthytetrault-mobility-ca.imanage.work> - DRAFT - Initial Order</mccarthytetrault-mobility-ca.imanage.work>
Document 2 ID	iManage://mccarthytetrault-mobility-ca.imanage.work/MTD OCS/47684994/4
Description	#47684994v4 <mccarthytetrault-mobility-ca.imanage.work> - DRAFT - Amended and Restated Initial Order</mccarthytetrault-mobility-ca.imanage.work>
Rendering set	Standard

Legend:	
Insertion	
Deletion	
Moved from	
Moved to	
Style change	
Format change	
Moved deletion	
Inserted cell	
Deleted cell	
Moved cell	
Split/Merged cell	
Padding cell	

Statistics:	
	Count
Insertions	75
Deletions	62
Moved from	1
Moved to	1
Style changes	0
Format changes	0
Total changes	139